SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

x ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended December 31, 2002

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____.

Commission File Number: 1-14100

IMPAC MORTGAGE HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Maryland

(State or other jurisdiction of incorporation or organization)

33-0675505 (I.R.S. Employer Identification No.)

1401 Dove Street, Newport Beach, California 92660 (Address of principal executive offices)

(949) 475-3600

(Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Act:

 Title of each class
 Name of each exchange on which registered

Common Stock, \$0.01 par value Preferred Share Purchase Rights

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes x No o

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of the Form 10-K or any amendment to this Form 10-K. o

Indicate by check mark whether the registrant is an accelerated filer (as defined in Exchange Act Rule 12b-2)

Yes x No o

As of June 28, 2002 the aggregate market value of the voting stock held by non-affiliates of the registrant was approximately \$531.3 million, based on the closing sales price of common stock on the American Stock Exchange on that date. For purposes of the calculation only, in addition to affiliated companies, all directors and executive officers of the registrant have been deemed affiliates. There were 49,165,068 shares of Common Stock outstanding as of March 26, 2003.

DOCUMENTS INCORPORATED BY REFERENCE

Portions of the Registrant's Definitive Proxy Statement issued in connection with the 2003 Annual Meeting of Stockholders of the Registrant are incorporated by reference into Part III.

Amorican Stock Exchange

American Stock Exchange American Stock Exchange

IMPAC MORTGAGE HOLDINGS, INC. 2002 FORM 10-K ANNUAL REPORT TABLE OF CONTENTS

PART I

ITEM 1.	BUSINESS	1
ITEM 2.	PROPERTIES	28
ITEM 3.	LEGAL PROCEEDINGS	28
ITEM 4.	SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS	29
	PART II	
ITEM 5.	MARKET FOR REGISTRANT'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS	29
ITEM 6.	SELECTED CONSOLIDATED FINANCIAL DATA	30
ITEM 7.	MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	33
ITEM 7A.	QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK	48
ITEM 8.	FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA	55
ITEM 9.	CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE	55
	PART III	
ITEM 10.	DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT	55
ITEM 11.	EXECUTIVE COMPENSATION	55
ITEM 12.	SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	55
ITEM 13.	CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS	55
ITEM 14	CONTROLS AND PROCEDURES	55
	PART IV	
ITEM 15	EXHIBITS, FINANCIAL STATEMENT SCHEDULES AND REPORTS ON FORM 8-K	55
<u>SIGNATUI</u>	RES	59
CERTIFIC	ATIONS	60

ITEM 1. BUSINESS

PART I

Forward-Looking Statements

This annual report on Form 10-K contains certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Forward-looking statements, some of which are based on various assumptions and events that are beyond our control, may be identified by reference to a future period or periods or by the use of forward-looking terminology, such as "may," "will," "believe," "expect," "plan," "anticipate," "continue," or similar terms or variations on those terms or the negative of those terms. Actual results could differ materially from those set forth in forward-looking statements, including, among other things, failure to achieve projected earning levels, the timely and successful implementation of strategic initiatives, the ability to generate sufficient liquidity, the ability to access the capital markets, interest rate fluctuations on our assets that differ from those on our liabilities, increase in prepayment rates on our mortgage assets, changes in assumptions regarding estimated loan losses or interest rates, the availability of financing and, if available, the terms of any financing, changes in general market and economic conditions and other factors described in this annual report. For a discussion of the risks and uncertainties that could cause actual results to differ from those contained in the forward-looking statements see "Risk Factors" and Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations" in this annual report. We do not undertake, and specifically disclaim any obligation, to publicly release the results of any revisions that may be made to any forward-looking statements to reflect the occurrence of anticipated or unanticipated events or circumstances after the date of such statements.

Available Information

Our Internet website address is www.impaccompanies.com. We make our annual report on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and proxy statement for our annual shareholders' meetings, as well as any amendments to those reports, available free of charge through our website as soon as reasonably practicable after we electronically file such material with, or furnish it to, the Securities and Exchange Commission, or "SEC." You can learn more about us by reviewing our SEC filings on our website by clicking on "Impac Mortgage Holdings, Inc." located on our home page and proceeding to "SEC filings." The SEC also maintains a website at www.sec.gov that contains reports, proxy statements and other information regarding SEC registrants, including the Company.

General

Unless the context otherwise requires, the terms "Company," "we," "us," and "our" refer to Impac Mortgage Holdings, Inc., a Maryland corporation incorporated in August 1995, and its subsidiaries, IMH Assets Corp., or "IMH Assets," Impac Warehouse Lending Group, Inc., or "IWLG," Impac Multifamily Capital Corporation, or "IMCC," and its affiliate Impac Funding Corporation, or "IFC," together with its wholly-owned subsidiaries Impac Secured Assets Corp., or "ISAC," and Novelle Financial Services, Inc., or "Novelle." References to Impac Mortgage Holdings, Inc., or "IMH," are made to differentiate IMH, the publicly traded Company, as a separate entity from IMH Assets, IWLG, IMCC and IFC.

We are a mortgage real estate investment trust, or "REIT." Together with our subsidiaries and affiliate we are a nationwide acquirer and originator of nonconforming Alt-A mortgages, or "Alt-A mortgages." Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but have loan characteristics that make them non-conforming under those guidelines. Some of the principal differences between mortgages purchased by Fannie Mae and Freddie Mac and Alt-A mortgages are as follows:

- credit and income histories of the mortgagor;
- documentation required for approval of the mortgagor; and
- applicable loan-to-value ratios.

For instance, mortgages may have higher loan-to-value, or "LTV," ratios than allowable under Fannie Mae or Freddie Mac guidelines. However, these mortgages that we acquire and originate may include private mortgage insurance. Furthermore, mortgages may not have certain documentation or verifications that are required by Fannie Mae and Freddie Mac. Therefore, in making our credit decisions, we are more reliant upon the borrower's credit score and the adequacy of the underlying collateral. We believe that Alt-A mortgages provide an attractive net earnings profile by producing higher yields without commensurately higher credit losses than other types of mortgages. Since 1999 we have acquired and originated primarily Alt-A mortgages. We also provide warehouse and repurchase financing to originators of mortgages. Our goal is to generate consistent reliable income for distribution to our stockholders primarily from earnings generated by our mortgage loan investment portfolio.

We operate three core businesses:

- long-term investment operations;
- mortgage operations; and
- warehouse lending operations.

The long-term investment operations invest primarily in adjustable and fixed rate Alt-A mortgages that are acquired and originated by our mortgage operations and small-balance multi-family mortgages originated by IMCC. This business primarily generates net interest income from its mortgage loan investment portfolio and, to a lesser extent, its investment securities portfolio. Our investment in Alt-A mortgages and small-balance multi-family mortgages, or "multi-family mortgages," is financed with cash flow from the mortgage loan investment portfolio, collateralized mortgage obligations, or "CMO," financing, short-term borrowings under reverse repurchase agreements and proceeds from the sale of capital stock.

The mortgage operations acquire, originate, sell and securitize primarily adjustable and fixed rate Alt-A mortgages. Our mortgage operations generate income by securitizing and selling loans to permanent investors, including our long-term investment operations. This business also earns revenue from fees associated with mortgage servicing rights, master servicing agreements and interest income earned on mortgages held-for-sale. Our mortgage operations use warehouse facilities provided by the warehouse lending operations to finance the acquisition and origination of mortgages.

The warehouse lending operations provide short-term financing to mortgage loan originators, including the mortgage operations, by funding mortgages from their closing date until they are sold to pre-approved investors. This business earns fees from warehouse transactions as well as net interest income from the difference between its cost of borrowings and the interest earned on warehouse advances.

Long-Term Investment Operations

The long-term investment operations are conducted by IMH, IMH Assets, a wholly-owned specialty purpose entity through which the long-term investment operations conducts its CMO borrowings, and IMCC, an originator of multi-family mortgages. The long-term investment operations invest primarily in Alt-A mortgages, multi-family mortgages and, to a lesser extent, mortgage-backed securities secured by or representing interests in mortgages. Income is primarily earned from net interest income received on mortgages and mortgage-backed securities acquired and held for long-term investment. Mortgages and mortgage-backed securities are financed with borrowings provided from CMOs, short-term borrowings under reverse repurchase agreements, borrowings secured by mortgage-backed securities and capital.

The mortgage operations supports the investment objectives of the long-term investment operations by supplying mortgages at prices that are comparable to those available through investment bankers and other third parties. We believe that purchasing mortgages acquired and originated by our mortgage operations affords us a competitive advantage because of our historical understanding of the underlying credit of these mortgages and the extensive information on the performance and prepayment patterns of these types of mortgages.

Mortgage Loan Investment Portfolio

We believe that Alt-A mortgages provide an attractive net earnings profile by producing higher yields without commensurately higher credit risks than other types of mortgages. We primarily acquire non-conforming "A" or "A-" credit quality mortgages, collectively, Alt-A mortgages. As defined by us, A credit quality mortgages generally have a credit score of 640 or better and A- credit quality mortgages generally have a credit score of between 600 and 639. As a comparison, Fannie Mae and Freddie Mac generally purchase conforming mortgages with credit scores greater than 620. We believe that by improving the overall credit quality of our mortgage investment portfolio we can more consistently generate a higher level of future cash flow and earnings.

Alt-A mortgages that we acquire for long-term investment are primarily adjustable rate mortgages, or "ARMs," hybrid adjustable rate mortgages, or "hybrid ARMs," and, to a lesser extent, fixed rate mortgages, or "FRMs." ARMs are generally fully amortizing mortgages with maturity periods ranging from 15 to 30 years. The interest rate on ARMs are typically tied to an index, such as the six-month London Interbank Offered Rate, or "LIBOR," plus a spread and adjust periodically, subject to lifetime interest rate caps and periodic interest rate and payment caps. The initial interest rates on ARMs are typically lower than the average comparable fixed rate mortgage but may be higher than average comparable FRMs over the life of the mortgage. Hybrid ARMs are generally fully amortizing mortgages with maturity periods ranging from 15 to 30 years with initial fixed interest rate periods generally ranging from two to five years, which subsequently adjust to ARMs. FRMs are generally fully amortizing mortgages with maturity periods ranging from 15 to 30 years.

The majority of mortgages acquired by the long-term investment operations have prepayment penalty features that have prepayment penalty periods ranging from two to five years. During 2002 the long-term investment operations acquired from the mortgage operations \$3.9 billion in principal balance of primarily adjustable and fixed rate Alt-A mortgages for long-term investment. The acquisition and retention of Alt-A and origination of multi-family mortgages increased CMO collateral to \$5.2 billion at year-end 2002.

The following table presents selected information about mortgages held as CMO collateral for the periods indicated:

		As of December 31,	
	2002	2001	2000
Percent Alt-A mortgages	99	95	90
Percent ARMs	85	87	65
Percent hybrid ARMs	35	68	65
Weighted average months until hybrid ARMs interest rate adjustment	14	15	16
Weighted average coupon	6.57	7.92	9.33
Weighted average margin	3.01	3.42	4.15
Weighted average original LTV	82	83	85
Weighted average original credit score	683	678	675
Percent with active prepayment penalty	76	54	36
Weighted average months until prepayment penalty expiration	24	22	27
Prior 12-month prepayment rate	25	34	25
Lifetime prepayment rate	33	34	31
Percent of mortgages in California	63	63	53
Percent purchase transactions	62	70	74
Percent owner occupied	93	94	96
Percent first lien	99	99	95

The following table presents mortgage acquisitions by loan characteristic for the periods indicated (in thousands):

		For the year ended December 31,									
		2002 (1	L)	2001				2000			
	Pri	ncipal Balance	%	Princ	cipal Balance	%	Prin	cipal Balance	%		
By Loan Type:											
Fixed rate	\$	599,566	15	\$	17,028	1	\$	28,140	6		
Adjustable rate		3,291,380	85		1,469,056	99		414,801	92		
Second trust deed		311	0		259	0		7,771	2		
Total volume	\$	3,891,257	100	\$	1,486,343	100	\$	450,712	100		
By Credit Quality:											
Alt-A	\$	3,875,903	100	\$	1,475,269	99	\$	419,219	93		
B/C		15,354	0		11,074	1		31,493	7		
Total volume	\$	3,891,257	100	\$	1,486,343	100	\$	450,712	100		
By Purpose:											
Purchase	\$	2,336,451	60	\$	997,350	67	\$	369,430	82		
Refinance		1,554,806	40		488,993	33		81,282	18		
Total volume	\$	3,891,257	100	\$	1,486,343	100	\$	450,712	100		
By Prepayment Penalty:											
With prepayment penalty	\$	3,074,741	79	\$	876,798	59	\$	230,967	51		
Without prepayment penalty		816,516	21		609,545	41		219,745	49		
Total volume	\$	3,891,257	100	\$	1,486,343	100	\$	450,712	100		
				_			_				

(1) Excludes \$25.8 million of multi-family mortgages originated by IMCC.

In addition to acquiring mortgages from our mortgage operations, the long-term investment operations originate multi-family mortgages through IMCC, a wholly-owned subsidiary of IMH. IMCC was formed to primarily originate small-balance multi-family ARMs with high credit quality and conservative LTV ratios with balances ranging from \$250,000 to \$1.5 million. Multi-family mortgages generally have interest rate floors, which is generally the initial start rate, and prepayment penalty periods ranging from three to five years. Multi-family mortgages provide greater asset diversification on our balance sheet as multi-family mortgages typically have higher interest rate spreads and longer lives than residential mortgages. IMCC originated \$25.8 million of multi-family mortgages in 2002.

For additional information regarding the mortgage loan investment portfolio refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations Results of Operations — Impac Mortgage Holdings, Inc.," "Note C — Mortgages Held-for-Investment" and "Note D — CMO Collateral" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Investment Securities Portfolio

Prior to 1998 and in connection with the issuance of mortgage-backed securities by the mortgage operations in the form of real estate mortgage investment conduits, or "REMICs," the long-term investment operations acquired senior or subordinated securities as regular interests on a short- or long-term basis and subordinated mortgage-backed securities issued by third parties. These securities or investments may subject us to credit, interest rate and prepayment risks. In general, subordinated classes of a particular series of securities bear all losses prior to the related senior classes. Losses in excess of expected losses at the time such securities are purchased would adversely affect our yield on these securities and could result in failure to recoup our initial investment.

Since 1998 our investment strategy has been to only acquire or invest in mortgage-backed securities that are secured by Alt-A mortgages underwritten and acquired and originated by our mortgage operations due to our understanding of the underlying credit of these mortgages and their historical performance. Since 1999 we have acquired no mortgage-backed securities from the mortgage operations and pursuant to our current investment policy we no longer acquire or invest in interest-only, principal-only or subordinate mortgage-backed securities. Since 2000 our

investment securities portfolio, which primarily includes subordinated mortgage-backed securities, as a percentage of total assets has declined significantly and represented less than 0.05% of total assets at year-end 2002. For additional information regarding the investment securities portfolio refer to "Note B — Investment Securities Available-for-Sale" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Financing

We primarily finance our long-term mortgage loan investment portfolio and investment securities portfolio as follows:

- issuance of CMOs;
- short-term borrowings under reverse repurchase agreements;
- borrowings secured by mortgage-backed securities; and
- proceeds from the sale of capital stock.

As we accumulate mortgages we may issue CMOs secured by such mortgages as a means of financing. The decision to issue CMOs is based on our current and future investment needs, market conditions and other factors. Each issue of CMOs is fully payable from the principal and interest payments on the underlying mortgages securing such debt, any cash or other collateral pledged as a condition of receiving the desired rating on the debt and any investment income on mortgage collateral. We earn a net interest spread on interest income on mortgages held as CMO collateral less interest and other expenses associated with CMO financing. Net interest spreads may be directly impacted by levels of early prepayment of underlying mortgages and, to the extent each CMO class has variable rates of interest, may be affected by changes in short-term interest rates. Our CMOs typically are structured as adjustable rate securities that are indexed to one-month LIBOR and fixed rate securities with interest payable monthly. Certain CMOs are guaranteed to certificate holders by a mortgage insurer, which give the CMOs the highest rating established by a nationally recognized rating agency. For additional information regarding CMOs refer to "Note H — CMO Borrowings" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

When we issue CMOs for financing purposes, we seek an investment grade rating for our CMOs by a nationally recognized rating agency. To secure such a rating, it is often necessary to pledge collateral in excess of the principal amount of the CMOs to be issued, or to obtain other forms of credit enhancements such as additional mortgage insurance. The need for additional collateral or other credit enhancements depends upon factors such as the type of collateral provided, the interest rates paid, the geographic concentration of the mortgaged property securing the collateral, and other criteria established by the rating agencies. The pledge of additional collateral reduces our capacity to raise additional funds through short-term secured borrowings or additional CMOs, and diminishes the potential expansion of our mortgage loan investment portfolio. As a result, our objective is to pledge additional collateral for CMOs only in the amount required to obtain an investment grade rating for our CMOs by a nationally recognized rating agency. Our total loss exposure is limited to the equity invested in the CMOs at any point in time.

Prior to the issuance of CMOs we use reverse repurchase agreements as short-term financing at interest rates that are consistent with our investment objectives. A reverse repurchase agreement acts as a financing vehicle under which we effectively pledge our mortgages as collateral to secure a short-term loan. Generally, the other party to the agreement makes the loan in an amount equal to a percentage of the market value of the pledged collateral. At the maturity of the reverse repurchase agreement, we are required to repay the loan and correspondingly we receive our collateral. Our borrowing agreements require us to pledge cash, additional mortgages or additional investment securities backed by mortgages in the event the market value of existing collateral declines. We may be required to sell assets to reduce our borrowings to the extent that cash reserves are insufficient to cover such deficiencies in collateral. For additional information regarding reverse repurchase agreements refer to "Note G — Reverse Repurchase Agreements" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

We finance a portion of our investment securities portfolio with principal-only notes. The principal-only notes represent senior or subordinated interests in trust funds primarily consisting of a pool of mortgages. The principal-only notes represent non-recourse obligations. For additional information regarding borrowings secured by investment

securities refer to "Note I — Borrowings Secured by Investment Securities Available-for-Sale" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Interest Rate Hedging

We follow an interest rate hedge program intended to limit our exposure to changes in interest rates primarily associated with cash flows on our adjustable rate CMO borrowings. Our primary objective is to hedge our exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of our adjustable rate CMO borrowings. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate hedge program is formulated with the intent to offset the potential adverse effects of changing interest rates on cash flows on adjustable rate CMO borrowings:

- interest rate adjustment limitations on mortgages held as CMO collateral due to periodic and lifetime interest rate cap features; and
- mismatched interest rate adjustment periods between mortgages held as CMO collateral and CMO borrowings.

To mitigate our exposure to the effect of changing interest rates on cash flows on our adjustable rate CMO borrowings, we acquire derivative instruments in the form of interest rate cap agreements, or "caps," interest rate floor agreements, or "floors," and interest rate swap agreements, or "swaps." For additional information regarding interest rate hedging refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk" and "Note P — Derivative Instruments and Hedging Activities" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Mortgage Operations

The mortgage operations are conducted by IFC, Impac Lending Group, or "ILG," a division of IFC, and Novelle and ISAC, wholly-owned subsidiaries of IFC. The mortgage operations acquire, originate, sell and securitize primarily adjustable and fixed rate Alt-A mortgages and, to a lesser extent, sub-prime mortgages, or "B/C mortgages," from correspondents, mortgage brokers and retail customers. IFC acquires primarily adjustable and fixed rate Alt-A mortgages from its network of third party correspondents on a flow (loan-by-loan) basis or on a bulk basis from other mortgage companies, which may include customers of IFC. Correspondents originate and close mortgages under IFC's mortgage programs. Correspondents include savings and loan associations, commercial banks and mortgage bankers and brokers. IFC acts as an intermediary between the originators of mortgages that do not meet the guidelines for purchase by Fannie Mae and Freddie Mac and permanent investors in mortgage-backed securities secured by or representing an ownership interest in mortgages. ILG markets, underwrites, processes and funds mortgages for both wholesale and retail customers. ILG works directly with mortgage brokers to originate, underwrite and fund their mortgages. Many of ILG's wholesale customers cannot conduct business with the mortgage operations as correspondents because they do not meet the higher net worth requirements. Through its retail channel, ILG markets mortgages directly to the public. ILG offers the same mortgage programs to its mortgage brokers and retail customers that are offered by IFC. Novelle is a wholly-owned subsidiary of IFC that originates B/C mortgages through a network of wholesale brokers and subsequently sells its loans to third party investors for cash gains. Novelle began operations during September of 2001 in San Diego, California, where it is currently headquartered, with capital provided by IFC. ISAC is a wholly-owned specialty purpose entity through which IFC issues itsREMICs.

Marketing Strategy

We believe that we can compete effectively with other Alt-A mortgage conduits through our efficient loan purchasing process, flexible purchase commitment options and competitive pricing and by designing Alt-A mortgages that suit the needs of our correspondents and mortgage brokers and their borrowers, which is intended to provide sufficient credit quality to our investors. Our principle strategy is to expand our market position as a low-cost nationwide acquirer and originator of Alt-A mortgages, while continuing to emphasize an efficient centralized operating structure. To help accomplish this, we have developed a second-generation web-based automated underwriting system called Impac Direct Access System for Lending, or "iDASLg2." iDASLg2 substantially increases efficiencies for our

customers and our mortgage operations by significantly decreasing the processing time for a mortgage while improving employee productivity and maintaining superior customer service.

iDASLg2 is an interactive Internet-based system that allows our customers to automatically underwrite mortgages, enabling our customers to pre-qualify borrowers for various mortgage programs and receive automated approval decisions. iDASLg2 is intended to increase efficiencies not only for our customers but also for the mortgage operations by significantly decreasing the processing time for a mortgage. iDASLg2 improves employee production and maintains superior customer service, which together leads to higher closing ratios, improved profit margins and increased profitability at all levels of our business operations. Most importantly, iDASLg2 allows us to move closer to our correspondents and mortgage brokers with minimal future capital investment while maintaining centralization, a key factor in the success of our operating strategy. During 2002 average monthly volume of mortgages submitted through IDASL and iDASLg2 increased by 70% to \$1.4 billion per month as compared to \$794.9 million per month in 2001. All of our correspondents submit loans via iDASLg2 and all wholesale loans delivered by mortgage brokers are directly underwritten through iDASLg2. Mortgages purchased on a bulk basis are not underwritten through iDASLg2.

We also focus on expansion opportunities to attract correspondent originators and mortgage brokers to our nationwide network in order to increase mortgage acquisitions and originations in a controlled manner. This allows us to shift the high fixed costs of interfacing with the homeowner to our correspondents and mortgage brokers. This marketing strategy is designed to accomplish three objectives as follow:

- attract a geographically diverse group of both large and small correspondent originators and mortgage brokers;
- establish relationships with correspondents and mortgage brokers that facilitate their ability to offer a variety of loan products designed by the mortgage operations; and
- purchase mortgages and securitize and sell them in the secondary market or to the long-term investment operations.

As of February 28, 2003 we maintained relationships with 222 correspondents and approximately 4,860 mortgage bankers and brokers. In order to accomplish our production objectives, we design and offer mortgage products that we believe are attractive to potential Alt-A borrowers and to end-investors in Alt-A mortgages and mortgage-backed securities. We have historically emphasized and continue to emphasize flexibility in our mortgage product mix as part of our strategy to attract and establish long-term relationships with our correspondents and mortgage brokers. We also maintain relationships with numerous end-investors so that we may develop mortgage products that may be of interest to them as market conditions change, which in turn may be offered through our correspondent network. In response to the needs of our correspondents, and as part of our strategy to facilitate the sale of our mortgages through the mortgage operations, our marketing strategy offers efficient response time in the purchase process, direct and frequent contact with our correspondents and mortgage brokers through a trained sales force and flexible commitment programs. Finally, due to the price sensitivity of most homebuyers, we are competitive in pricing our products in order to attract sufficient numbers of mortgages.

Mortgage Acquisitions and Originations

The majority of mortgages acquired and originated by the mortgage operations are adjustable and fixed rate Alt-A mortgages. A portion of Alt-A mortgages that are acquired and originated by the mortgage operations exceed the maximum principal balance for a conforming loan purchased by Fannie Mae or Freddie Mac, which is currently \$322,700. Mortgages that exceed such maximum principal balance are referred to as "jumbo loans." Alt-A mortgages generally consist of jumbo loans or other mortgages that are acquired and originated in accordance with underwriting or product guidelines that differ from those applied by Fannie Mae and Freddie Mac. Alt-A mortgages may involve greater risk as a result of different underwriting and product guidelines. A portion of mortgages acquired and originated through the mortgage operations are B/C mortgages, which may entail greater credit risks than Alt-A mortgages. We generally do not acquire or originate mortgages made to borrowers with lower credit ratings than borrowers of higher credit quality, Alt-A mortgages and are normally subject to higher rates of loss and delinquency than other Alt-A mortgages acquired and originated by the mortgage operations. As a result, B/C mortgages normally bear a higher rate of interest and are typically subject to higher fees than Alt-A mortgages. In general, greater emphasis is placed upon the value of the mortgaged property and, consequently, the quality of

appraisals, and less upon the credit history of the borrower in underwriting B/C mortgages than in underwriting Alt-A mortgages. In addition, B/C mortgages are generally subject to lower LTV ratios than Alt-A mortgages. Mortgages acquired by the mortgage operations are generally secured by first liens and, to a lesser extent, second liens on single-family residential properties with either adjustable or fixed interest rates. FRMs have a constant interest rate over the life of the loan, which is generally 15 or 30 years. The interest rate on ARMs are typically tied to an index, such as six-month LIBOR plus a spread and adjust periodically, subject to lifetime interest rate caps and periodic interest rate and payment caps. The initial interest rates on ARMs are typically lower than the average comparable FRM but may be higher than average comparable FRMs over the life of the loan. We acquire and originate mortgages with the following most common loan characteristics, although we may purchase mortgages with other interest rate and maturity characteristics:

- FRMs that have original terms to maturity ranging from 15 to 30 years with two- to five-year prepayment penalty periods;
- ARMs that adjust based on six-month LIBOR with terms to maturity ranging from 15 to 30 years with one- to five-year prepayment penalty periods; and
- two- and three-year FRMs with terms to maturity ranging from 15 to 30 years that subsequently adjust to six-month LIBOR ARMs with one- to fiveyear prepayment penalty periods.

The following table presents mortgage acquisitions and originations by loan characteristic for the periods indicated (in thousands):

			I	or t	he year ended, Deco	ember 31,			
		2002			200	1		2000	
		rincipal Balance	%		Principal Balance	%		Principal Balance	%
By Loan Type:									
Adjustable rate	\$	3,703,657	62	\$	1,541,329	49	\$	487,346	23
Fixed rate		2,159,696	36		1,570,225	50		1,539,741	74
Second trust deed		82,145	2		43,074	1		51,737	3
Total volume	\$	5,945,498	100	\$	3,154,628	100	\$	2,078,824	100
By Production Channel:							_		
Correspondent acquisitions:									
Flow	\$	3,701,537	62	\$	2,213,736	70	\$	1,731,351	84
Bulk		750,004	13		169,282	5		71,283	3
Total correspondent acquisitions		4,451,541	75		2,383,018	75		1,802,634	87
Wholesale and retail originations		1,089,008	18		683,060	22		276,190	13
Novelle		404,949	7		88,550	3			0
Total volume	\$	5,945,498	100	\$	3,154,628	100	\$	2,078,824	100
By Credit Quality:									
	\$	5,515,573	93	\$	3,046,532	97	\$	1,999,246	96
B/C (1)	Ψ	429,925	7		108,096	3	Ψ	79,578	4
Total volume	\$	5,945,498	100	\$	3,154,628	100	\$	2,078,824	100
By Purpose:									
Purchase	\$	3,288,566	55	\$	1,938,715	61	\$	1,675,893	81
Refinance		2,656,932	45		1,215,913	39		402,931	19
Total volume	\$	5,945,498	100	\$	3,154,628	100	\$	2,078,824	100
By Prepayment Penalty:									
	\$	4,677,078	79	\$	2,058,746	65	\$	1,104,154	53
Without prepayment penalty		1,268,420	21		1,095,882	35	_	974,670	47
Total volume	\$	5,945,498	100	\$	3,154,628	100	\$	2,078,824	100

^{(1) 2002} and 2001 includes \$404.9 million and \$88.6 million, respectively, of B/C mortgages originated by Novelle that are subsequently sold to third party investors for cash gains.

Our mortgage acquisition and origination activities focus on those regions of the country where higher volumes of Alt-A mortgages are originated including California, Florida, Colorado, Georgia, New Jersey, Illinois, Texas, New York, Washington and Virginia. The highest concentration of Alt-A mortgages acquired and originated relate to properties located in California and Florida because of generally higher property values and mortgage balances. During the years ended December 31, 2002 and 2001 mortgages secured by California and Florida properties accounted for approximately 56% and 10%, respectively, of mortgage acquisitions and originations during both years. Of the \$5.9 billion in principal balance of mortgages acquired and originated in 2002, \$2.2 billion, or 37%, were acquired from our top ten correspondents and wholesale brokers. Express Lending Group accounted for \$811.4 million, or 14%, of mortgages acquired and originated by the mortgage operations in 2002. No other correspondents or brokers accounted for more than 10% of the total mortgages acquired and originated by the mortgage operations in 2002.

Underwriting

We have developed comprehensive purchase guidelines for the acquisition and origination of mortgages. Each mortgage underwritten assesses the borrower's credit score, ability and willingness to repay the mortgage obligation and the adequacy of the mortgaged property as collateral for the mortgage. Subject to certain exceptions and the type of mortgage product, each purchased mortgage must conform to the loan parameters and eligibility requirements specified in our seller/servicer guide with respect to, among other things, loan amount, type of property, compliance, LTV ratio, mortgage insurance, credit history, debt service-to-income ratio, appraisal and loan documentation. We also perform a full legal documentation review prior to the purchase of all mortgages.

All mortgages acquired or originated under our loan programs are underwritten either by our employees or by contracted mortgage insurance companies or delegated sellers. Under all of our underwriting methods, loan documentation requirements for verifying the borrower's income and assets vary according to LTV ratios and other factors. Generally, as the standards for required documentation are lowered, the borrowers' down payment requirements are increased and the required LTV ratios are decreased. The borrower is also required to have a stronger credit history, larger cash reserves and an appraisal of the property that may be validated by an enhanced desk or field review, depending on the loan program. Lending decisions are based on a risk analysis assessment after the review of the entire mortgage file. Each mortgage is individually underwritten with emphasis placed on the overall quality of the mortgage.

Seller Eligibility Requirements

Mortgages acquired by the mortgage operations are originated by various sellers, including mortgage bankers, savings and loan associations and commercial banks. Sellers are required to meet certain regulatory, financial, insurance and performance requirements established by us before they are eligible to participate in our mortgage purchase programs. Sellers must also submit to periodic reviews to ensure continued compliance with these requirements. Our current criteria for seller participation generally includes a minimum tangible net worth requirement of \$500,000, approval as a Fannie Mae or Freddie Mac seller/servicer in good standing, a Housing and Urban Development, or "HUD," approved mortgagee in good standing or a financial institution that is insured by the Federal Deposit Insurance Corporation, or "FDIC," or comparable federal or state agency, and that the seller is examined by a federal or state authority.

In addition, sellers are required to have comprehensive mortgage origination quality control procedures. In connection with its qualification, each seller enters into an agreement that generally provides for recourse by us against the seller in the event of a breach of representations or warranties made by the seller with respect to mortgages sold to us, which includes but is not limited to any fraud or misrepresentation during the mortgage loan origination process or upon early payment default on mortgages.

Securitization and Sale

After acquiring mortgages from correspondents on a flow or bulk basis, and originating mortgages through ILG's and Novelle's wholesale and retail channels the mortgage operations sell or securitize mortgages to permanent investors. The mortgage operations sell substantially all of its ARM acquisitions and originations and approximately 50% of its FRM acquisitions and originations to the long-term investment operations at prices comparable to prices available to third party investors at the date of sale. During 2002 the mortgage operations sold \$3.9 billion in principal balance of mortgages to the long-term investment operations. Generally, the mortgage operations sell all of its

mortgage acquisitions and originations to third party investors as servicing released, which means that we do not retain primary mortgage servicing rights.

When a sufficient volume of FRMs with similar characteristics has been accumulated, generally \$100 million to \$350 million, the mortgage operations will sell bulk packages, referred to as whole loan sales, to third party investors in mortgages or will securitize them through the issuance of mortgage-backed securities in the form of REMICs. During 2002 the mortgage operations sold \$1.1 billion in principal balance of mortgages as whole loan sales and \$600.0 million in principal balance of mortgages as REMICs.

The period of time between when we commit to purchase mortgages and the time we sell or securitize mortgages generally ranges from 10 to 90 days, depending on certain factors including the length of the purchase commitment period, volume by product type and the securitization process. REMIC transactions are generally accounted for as sales of the mortgages and can eliminate or minimize any long-term residual investment in such mortgages. REMIC securities consist of one or more classes of "regular interests" and a single class of "residual interest." The regular interests are tailored to the needs of investors and may be issued in multiple classes with varying maturities, average lives and interest rates.

REMICs created by us are structured so that one or more of the classes of securities are rated investment grade by at least one nationally recognized rating agency. The ratings for our REMICs are based upon the perceived credit risk by the applicable rating agency of the underlying mortgages, the structure of the securities and the associated level of credit enhancement. Credit enhancement is designed to provide protection to the security holders in the event of borrower defaults and other losses including those associated with fraud or reductions in the principal balances or interest rates on mortgages as required by law or a bankruptcy court.

Hedging Activities

The mortgage operations hedges interest rate risk and price volatility on its mortgage loan investment portfolio during the time it commits to acquire or originate mortgages at a pre-determined rate and the time it sells or securitizes mortgages. To mitigate interest rate and price volatility risks, IFC may enter into forward commitments and derivative transactions. The nature and quantity of derivative transactions are determined based on various factors, including market conditions and the expected volume of mortgage acquisitions and originations.

For additional information regarding interest rate hedging refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk" and "Note K — Derivative Instruments and Hedging Activities" in the accompanying notes to the consolidated financial statements for Impac Funding Corporation.

Master Servicing

We retain master servicing rights on substantially all of our Alt-A mortgage acquisitions and originations. Our function as master servicer includes collecting loan payments from other loan servicers and remitting loan payments, less master servicing fees receivable and other fees, to a trustee or other purchaser for each series of mortgage-backed securities or loans master serviced. In addition, as master servicer, we monitor compliance with our servicing guidelines and are required to perform, or to contract with a third party to perform, all obligations not adequately performed by any loan servicer. We may also be required to advance funds to cover interest payments not received from borrower's depending on the status of their mortgages.

With respect to IFC's function as a master servicer for loans owned by the long-term investment operations, IFC and IMH have entered into agreements with terms substantially similar to those of independent third parties. Master servicing fees are generally 0.03% per annum on the declining principal balances of the loans serviced. At year-end 2002 we master serviced 48,520 mortgages with a principal balance of \$8.7 billion.

The following table presents the amount of delinquent mortgages in our master servicing portfolio for the periods indicated (dollars in thousands):

	 As of December 31,										
	2002		200	1	2000						
	Principal % of Master Balance of Servicing Mortgages Portfolio		Principal Balance of Mortgages	% of Master Servicing Portfolio	Principal Balance of Mortgages	% of Master Servicing Portfolio					
Loans delinquent for:											
60-89 days	\$ 100,878	1.16%\$	72,460	1.30%\$	65,809	1.63%					
90 days and over	71,466	0.82	72,544	1.30	25,952	0.64					
Total 60 days and over	172,344	1.98	145,004	2.60	91,761	2.27					
Foreclosures pending	212,309	2.44	132,571	2.38	57,133	1.41					
Bankruptcies pending	26,402	0.30	22,054	0.40	22,556	0.56					
	 	·									
Total	\$ 411,055	4.72%\$	299,629	5.38%\$	171,450	4.24%					

Servicing

We subcontract all of our servicing obligations to independent third parties pursuant to sub-servicing agreements. We believe that the selection of thirdparty sub-servicers or the sale of servicing rights is more effective than establishing a servicing department within our mortgage operations. However, part of our responsibility is to continually monitor the performance of the sub-servicers or servicers through performance reviews and regular site visits. Depending on subservicer reviews, we may in the future rely on our internal collection group to take an ever more active role to assist the sub-servicer in the servicing of mortgages.

Servicing includes collecting and remitting loan payments, making required advances, accounting for principal and interest, holding escrow or impound funds for payment of taxes and insurance, if applicable, making required inspections of the mortgaged property, contacting delinquent borrowers, and supervising foreclosures and property dispositions in the event of un-remedied defaults in accordance with our guidelines. Servicing fees range from 0.25% per annum for FRMs to 0.50% per annum for B/C mortgages and 0.375% for ARMs on the declining principal balances of loans serviced. Additionally, we earn interest, or "float," on principal and interest payments we receive from our borrowers until those payments are remitted to the investors in those mortgages. We generally acquire all of our mortgages on a servicing released basis. To the extent the mortgage operations finances the acquisition of mortgages with warehouse facilities provided by IWLG, the mortgage operations pledges mortgages and the related servicing rights to IWLG as collateral. As a result, IWLG has an absolute right to control the servicing of such mortgages, including the right to collect payments on the underlying mortgages, and to foreclose upon the underlying real property in the case of default. Typically, IWLG delegates its right to service the mortgage securing the warehouse line to IFC.

The following table presents information regarding our mortgage servicing portfolio, including mortgages held-for-sale and mortgages held-for-investment, for the periods shown (dollars in millions, except average loan size):

	 As of December 31,					
	 2002	2001	2000			
Beginning servicing portfolio	\$ 1,754.4 \$	\$ 2,428.9	\$ 2,393.4			
Add: Loan acquisitions and originations	5,945.5	3,154.7	2,078.8			
Less: Servicing transferred	(4,620.3)	(2,866.1)	(1,266.0)			
Principal paydowns (1)	 (426.2)	(963.1)	(777.3)			
Ending servicing portfolio	\$ 2,653.4 \$	\$ 1,754.4	\$ 2,428.9			
Number of loans serviced	15,987	14,570	20,644			
Average loan size	\$ 166,000 \$	\$ 120,000	\$ 118,000			
Weighted average coupon	7.26%	8.80%	9.88%			

(1) Includes normal principal runoff and principal prepayments.

Warehouse Lending Operations

The warehouse lending operations is conducted by IWLG. The warehouse lending operations provide reverse repurchase financing to affiliated companies and to approved mortgage banks, some of which are correspondents of the mortgage operations, to finance mortgages during the time from the closing of the mortgages to their sale or other settlement with pre-approved investors. The warehouse lending operations relies mainly on the sale or liquidation of the mortgages as a source of repayment. Any claim of the warehouse lending operations as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay. Borrowings under the warehouse facilities are presented on our balance sheets as finance receivables. Terms of non-affiliated warehouse lines, including the commitment amount, are determined based upon the financial strength, historical performance and other qualifications of the borrower.

Regulation

We establish underwriting guidelines that include provisions for inspections and appraisals, require credit reports on prospective borrowers and determine maximum loan amounts. Our mortgage acquisition and origination activities are subject to, among other laws, the Equal Credit Opportunity Act, Federal Truthin-Lending Act, Real Estate Settlement Procedures Act and Home Mortgage Disclosure Act and the regulations promulgated thereunder. These laws and regulations prohibit discrimination and require the disclosure of certain basic information to mortgagors concerning credit terms and settlement costs, limit payment for settlement services to the reasonable value of the services rendered and require the maintenance and disclosure of information regarding the disposition of mortgage applications based on race, gender, geographical distribution and income level. Our mortgage acquisition and origination activities are also subject to state and local laws and regulations, including predatory lending laws, and may also be subject to applicable state usury statutes. IFC, our mortgage operations, is an approved Fannie Mae and Freddie Mac seller and/or servicer. IFC is subject to the rules and regulations of Fannie Mae and Freddie Mac audited financial statements, and each regulatory entity has its own financial requirements for sellers/servicers. IFC's affairs are also subject to examination by Fannie Mae and Freddie Mac with applicable regulations, policies and procedures.

Competition

In acquiring and originating Alt-A mortgages and issuing securities backed by such loans, we compete with established mortgage conduit programs, investment banking firms, savings and loan associations, banks, thrift and loan associations, finance companies, mortgage bankers, insurance companies, other lenders and other entities purchasing mortgage assets. Consolidation in the mortgage banking industry may also reduce the number of current correspondents and independent mortgage brokers available to the mortgage operations, reducing our potential customer base and resulting in the mortgage operations acquiring and originating a larger percentage of mortgages from a smaller number of customers. Changes of this nature could negatively impact our businesses. Mortgagebacked securities issued by the mortgage operations and the long-term investment operations face competition from other investment opportunities available to prospective investors. We face competition in our mortgage operations and warehouse lending operations from other financial institutions, including but not limited to banks and investment banks. Our competitors include IndvMac Bancorp, Inc., Greenpoint Financial Corporation, Residential Funding Corporation, Aurora Loan Services, Inc. and Credit Suisse First Boston Corporation. Competition can take place on various levels, including convenience in obtaining a mortgage, service, marketing, origination channels and pricing. We depend primarily on correspondents and independent mortgage brokers for the acquisition and origination of mortgages. These independent mortgage brokers deal with multiple lenders for each prospective borrower. We compete with these lenders for the independent brokers' business on the basis of price, service, loan fees, costs and other factors. Our competitors also seek to establish relationships with such brokers, who are not obligated by contract or otherwise to do business with us. Many of the institutions with which we compete in our mortgage operations and warehouse lending operations have significantly greater financial resources than we have. However, we can compete effectively with other Alt-A mortgage conduits through our efficient loan purchasing process, flexible purchase commitment options and competitive pricing and by designing Alt-A mortgages that suit the needs of our correspondents and their borrowers, which is intended to provide sufficient credit quality to our investors.

Employees

At year-end 2002 we had 432 full- and part-time employees and temporary and contract employees. The mortgage operations employed 393 full- and part-time employees and temporary and contract employees, of which 85 were employed by Novelle. IWLG and IMCC employed 30 and 9 full- and part-time employees and temporary and contract employees, respectively. Management believes that relations with its employees are good. The Company is not a party to any collective bargaining agreement.

Revisions in Policies and Strategies

Our board of directors has approved the Company's investment and operating policies and strategies. Our core operations involve the acquisition and origination of mortgages and their subsequent securitization and sale. We also act as a warehouse lender providing financing facilities to mortgage originators. These operations, their associated policies and strategies, are further described herein. Our board of directors has delegated our asset/liability management to the Asset/Liability Committee, or "ALCO," which reports to the board of directors at least quarterly. See a further discussion of ALCO in Item 7. "Management's Discussion of Financial Condition and Results of Operations" and Item 7A. "Quantitative and Qualitative Disclosures About Market Risk." Any of our policies, strategies and activities may be modified or waived by our board of directors without stockholder consent. Developments in the market, which affect the policies and strategies mentioned herein or which change our assessment of the market, may cause our board of directors to revise our policies and financing strategies.

We have elected to qualify as a REIT for tax purposes. We have adopted certain compliance guidelines, which include restrictions on the acquisition, holding and sale of assets. Prior to the acquisition of any asset, we determine whether the asset meets REIT requirements. Substantially all of the assets that we have acquired and will acquire for investment are expected to qualify as REIT assets. This requirement limits our investment strategies.

The long-term investment operations primarily invest in Alt-A mortgages and, to a lesser extent, B/C mortgages that are acquired from our mortgage operations. Our mortgage operations generally do not acquire or originate mortgages with principal balances above \$750,000 for Alt-A credit quality mortgages and \$500,000 for B/C mortgages. The long-term investment operations does not have a limit on the proportion of its assets that may be invested in each type of mortgage. Our goal is that mortgage-backed securities do not consist more than 10% of our total assets.

We closely monitor our acquisition and investment in mortgage assets and the sources of our income, including income from our hedging strategies, to ensure at all times that we maintain our qualifications as a REIT. We have developed certain accounting systems and testing procedures to facilitate our ongoing compliance with the REIT provisions of the Internal Revenue Code. No changes in our investment policies and operating strategies, including credit criteria for mortgage asset investments, may be made without the approval of our board of directors.

We may at times and on terms that our board of directors deems appropriate:

- Issue senior securities In December 1998 we issued \$30.0 million of Series B 10.5% Convertible Preferred Stock. In February 2000 all shares of Series B Convertible Preferred Stock were exchanged for Series C 10.5% Cumulative Convertible Preferred Stock and the conversion rate was adjusted to \$4.72 per share from \$4.95 per share convertible into 5.29661 shares of common stock from 5.050505 shares of common stock. In August 2001 the shares of Series C Preferred Stock were converted to common stock;
- Borrow money We finance our operations in large part through the issuance of CMOs, short-term borrowings under reverse repurchase agreements and borrowings secured by mortgage-backed securities. In addition, in March 1999, certain of our stockholders exchanged 1,359,507 shares of our common stock held by them, at an average price of \$5.70 per share, for our 11% senior subordinated debentures due February 15, 2004. The debentures bore interest at 11% per annum and mature on February 15, 2004. Commencing on February 15, 2001, the debentures became redeemable at our option, and, in June 2001, we redeemed all of the debentures at a price equal to the face amount of the debentures plus accrued and unpaid interest;
- Make loans to other persons The warehouse lending operations provide reverse repurchase financing to affiliated companies and to approved mortgage banks, some of which are correspondents of the mortgage

operations, to finance mortgages during the time from the closing of the mortgages to their sale or other settlement with pre-approved investors;

- Invest in securities of other issuers for the purpose of exercising control;
- Engage in the purchase and sale of investments In connection with the issuance of mortgage-backed securities by our mortgage operations in the form of REMICs, our long-term investment operations may retain senior or subordinated securities on a short- or long-term basis;
- Offer securities in exchange for property In March 1999 certain of our stockholders exchanged 1,359,507 shares of their common stock at an average price of \$5.70 per share for 11% senior subordinated debentures; and,
- Repurchase or otherwise reacquire our shares or other securities in the future In October 1998 we adopted a repurchase plan to repurchase up to \$5.0 million of our common stock in the open market. In 1999 the board of directors approved common stock repurchases up to an additional \$5.0 million, or a total of \$10.0 million. During 1999 we repurchased 2.0 million shares of our common stock for \$9.9 million. During 2000 we adopted a repurchase plan to repurchase up to \$3.0 million of our common stock in the open market. During 2001 and 2000 we repurchased 1,015,950 shares for \$2.3 million. During 2002 we did not repurchase any shares of common stock.

We may also underwrite the securities of other issuers, although we have not done so in the past and have no present intention to do so. Historically, we have and intend to continue to distribute annual reports to our stockholders, including financial statements certified by independent auditors, describing our current business and strategy.

Risk Factors

The following risk factors reference our mortgage loan investment portfolio, which consists of CMO collateral, mortgages held-for-investment and finance receivables. For information on the characteristics of the mortgage loan investment portfolio refer to "Note C—Mortgages Held-for-Investment", "Note D—CMO Collateral" and "Note E — Finance Receivables" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

A prolonged economic downturn or recession would likely result in a reduction of our mortgage origination activity which would adversely affect our financial results.

The United States economy has recently undergone and may in the future, undergo, a period of slowdown, which some observers view as a recession. An economic downturn or a recession may have a significant adverse impact on our operations and our financial condition. For example, a reduction in new mortgages will adversely affect our ability to expand our mortgage portfolio, our principal means of increasing our earnings. In addition, a decline in new mortgage activity will likely result in reduced activity for our warehouse lending operations and our long-term investment operations. In the case of our mortgage operations, a decline in mortgage activity may result in fewer loans that meet its criteria for purchase and securitization, thus resulting in a reduction in interest income and fees and gain on sale of loans. We may also experience larger than previously reported losses on our investment portfolio due to a higher level of defaults or foreclosures on our mortgages.

Terrorist attacks or military action may adversely affect our financial results.

The effects that terrorist attacks in the United States or other incidents and related military action, or possible military action by U.S. forces in Iraq or other regions, may have on the mortgage operations' ability to acquire or originate mortgages, the performance of the mortgages held by the long-term investment operations or on the values of mortgaged properties cannot be determined at this time. As a result of terrorist activity or military action, there may be a reduction in new mortgages, which will adversely affect our ability to expand our mortgage portfolio. In addition, a decline in new mortgage activity will likely result in reduced activity for our warehouse lending operations and our long-term investment operations. Mortgages held by our long-term investment operations may experience higher rates of delinquency, default and prepayment. These potential consequences of terrorist attacks or military action will have

an adverse affect on our financial results. Federal agencies and non-government lenders have, and may continue to, defer, reduce or forgive payments and delay foreclosure proceedings in respect of loans to borrowers affected in some way be recent and possible future events. In addition, activation of a substantial number of U.S. military reservists or members of the National Guard may significantly increase the proposition of mortgages whose mortgage rates are reduced by application of the Soldiers' and Sailors' Civil Relief Act of 1940 or similar state laws, and neither the master servicers nor the servicers will be required to advance for any interest shortfall caused by any such reduction. Interest payable to senior and subordinate certificate holders will be reduced on a pro rata basis by any reductions in the amount of interest collectible as a result of application of the Soldiers' and Sailors' Civil Relief Act of 1940 are senior of the Soldiers' and Sailors' Civil Relief Act of application of the Soldiers' and Sailors are reduced by any such reduction. Interest payable to senior and subordinate certificate holders will be reduced on a pro rata basis by any reductions in the amount of interest collectible as a result of application of the Soldiers' and Sailors' Civil Relief Act of 1940 or similar state laws.

If we are unable to generate sufficient liquidity we will be unable to conduct our operations as planned.

If we cannot generate sufficient liquidity, we will be unable to continue to grow our operations, grow our asset base, maintain our current hedging policy and pay dividends. We have traditionally derived our liquidity from four primary sources:

- financing facilities provided to us by others to acquire or originate mortgage assets;
- whole loan sales and securitizations of acquired or originated mortgages;
- our issuance of equity and debt securities; and
- earnings from operations.

We cannot assure you that any of these alternatives will be available to us, or if available, that we will be able to negotiate favorable terms. Our ability to meet our long-term liquidity requirements is subject to the renewal of our credit and repurchase facilities and/or obtaining other sources of financing, including additional debt or equity from time to time. Any decision by our lenders and/or investors to make additional funds available to us in the future will depend upon a number of factors, such as our compliance with the terms of our existing credit arrangements, our financial performance, industry and market trends in our various businesses, the lenders' and/or investors' own resources and policies concerning loans and investments, and the relative attractiveness of alternative investment or lending opportunities. If we cannot raise cash by selling debt or equity securities, we may be forced to sell our assets at unfavorable prices or discontinue various business activities. Our inability to access the capital markets could have a negative impact on our earnings growth and also our ability to pay dividends.

Any significant margin calls under our financing facilities would adversely affect our liquidity and may adversely affect our financial results.

Prior to the fourth quarter of 1998, we generally had no difficulty in obtaining favorable financing facilities or in selling acquired mortgages. However, during the fourth quarter of 1998, the mortgage industry experienced substantial turmoil as a result of a lack of liquidity in the secondary markets. At that time, investors expressed unwillingness to purchase interests in securitizations due, in part, to:

- the lack of financing to acquire these securitization interests;
- the widening of returns expected by institutional investors on securitization interests over the prevailing Treasury rate; and
- market uncertainty.

As a result, many mortgage originators, including our Company, were unable to access the securitization market on favorable terms. This resulted in some companies declaring bankruptcy. Originators, like our Company, were required to sell loans on a whole loan basis and liquidate holdings of mortgage-backed securities to repay short-term borrowings. However, the large amount of mortgages available for sale on a whole loan basis affected the pricing offered for these mortgages, which in turn reduced the value of the collateral underlying the financing facilities. Therefore, many providers of financing facilities initiated margin calls. Margin calls resulted when our lenders evaluated the market

value of the collateral securing our financing facilities and required us to provide them with additional equity or collateral to secure our borrowings.

Our financing facilities were short-term borrowings and due to the turmoil in the mortgage industry during the latter part of 1998 many traditional providers of financing facilities were unwilling to provide facilities on favorable terms, or at all. Our current financing facilities continue to be short-term borrowings and we expect this to continue. If we cannot renew or replace maturing borrowings, we may have to sell, on a whole loan basis, the loans securing these facilities which, depending upon market conditions, may result in substantial losses.

We incurred losses for fiscal years 1997, 1998 and 2000 and may incur losses in the future.

During the year ended December 31, 2000 we experienced a net loss of \$54.2 million. The net loss incurred during 2000 included accounting charges of \$68.9 million. The accounting charges were the result of write-downs of non-performing investment securities secured by mortgages and additional increases in the provision for loan losses to provide for the deterioration of the performance of collateral supporting specific investment securities. During the year ended December 31, 1998 we experienced a net loss of \$5.9 million primarily as the mortgage industry experienced substantial turmoil as a result of a lack of liquidity in the secondary markets which caused us to sell mortgages at losses to meet margin calls on our warehouse facilities. During the year ended December 31, 1997 we experienced a net loss of \$16.0 million. The net loss incurred during 1997 included an accounting charge of \$44.4 million that was the result of expenses related to the termination and buyout of our management agreement with Imperial Credit Advisors, Inc. We cannot be certain that revenues will remain at current levels or improve or that we will be profitable in the future, which could prevent us from effectuating our business strategy.

If we are unable to complete securitizations, we would face a liquidity shortage which would adversely affect our operating results.

We rely significantly upon securitizations to generate cash proceeds to repay borrowings and to create credit availability. Any reduction in our ability to complete securitizations may require us to utilize other sources of financing, which, if available at all, may be on unfavorable terms. In addition, delays in closing securitizations of our mortgages increase our risk by exposing our Company to credit and interest rate risks for this extended period of time. Furthermore, gains on sales from our securitizations represent a significant portion of our earnings. Several factors could affect our ability to complete securitizations of our mortgages, including:

- conditions in the securities and secondary markets;
- credit quality of the mortgages acquired or originated through our mortgage operations;
- volume of our mortgage loan acquisitions and originations;
- our ability to obtain credit enhancements; and
- lack of investors purchasing higher risk components of the securities.

If we are unable to profitably securitize a significant number of our mortgages in a particular financial reporting period, then we could experience lower income or a loss for that period. As a result of turmoil in the securitization market during the latter part of 1998, many mortgage lenders, including our Company, were required to sell mortgages on a whole loan basis under adverse market conditions in order to generate liquidity. Many of these sales were made at prices lower than our carrying value of the mortgages and we experienced substantial losses. We cannot assure you that we will be able to continue to profitably securitize or sell our loans on a whole loan basis, or at all.

The market for first loss risk securities, which are securities that take the first loss when mortgages are not paid by the borrowers, is generally limited. In connection with our REMIC securitizations, we endeavor to sell all securities subjecting us to a first loss risk. If we cannot sell these securities, we may be required to hold them for an extended period, subjecting us to a first loss risk.

Our borrowings and use of substantial leverage may cause losses.

Our use of collateralized mortgage obligations may expose our operations to credit losses.

To grow our investment portfolio, we borrow a substantial portion of the market value of substantially all of our investments in mortgages in the form of CMOs. Historically, we have borrowed approximately 98% of the market value of such investments. There are no limitations on the amount we may borrow, other than the aggregate value of the underlying mortgages. We currently use CMOs as financing vehicles to increase our leverage, since mortgages held for CMO collateral are retained for investment rather than sold in a secondary market transaction.

Retaining mortgages as collateral for CMOs exposes our operations to greater credit losses than does the use of other securitization techniques that are treated as sales because as the equity holder in the security, we are allocated losses from the liquidation of defaulted loans first prior to any other security holder. Although our liability under a collateralized mortgage obligation is limited to the collateral used to create the collateralized mortgage obligation, we generally are required to make a cash equity investment to fund collateral in excess of the amount of the securities issued in order to obtain the appropriate credit ratings for the securities being sold, and therefore obtain the lowest interest rate available, on the CMOs. If we experience greater credit losses than expected on the pool of loans subject to the CMO, the value of our equity investment will decrease and we would have to increase the allowance for loan losses on our financial statements.

The cost of our borrowings may exceed the return on our assets.

The cost of borrowings under our financing facilities corresponds to a referenced interest rate plus or minus a margin. The margin varies depending on factors such as the nature and liquidity of the underlying collateral and the availability of financing in the market. We will experience net interest losses if the returns on our assets financed with borrowed funds fail to cover the cost of our borrowings, and we did not implement any applicable financial hedges.

If we default under our financing facilities, we may be forced to liquidate the collateral at prices less than the amount borrowed.

If we default under our financing facilities, our lenders could force us to liquidate the collateral. If the value of the collateral is less than the amount borrowed, we could be required to pay the difference in cash. Furthermore, if we default under one facility, it would generally cause a default under our other facilities. If we were to declare bankruptcy, some of our reverse repurchase agreements may obtain special treatment and our creditors would then be allowed to liquidate the collateral without any delay. On the other hand, if a lender with whom we have a reverse repurchase agreement declares bankruptcy, we might experience difficulty repurchasing our collateral, or enforcing our claim for damages, and it is possible that our claim could be repudiated and we could be treated as an unsecured creditor. If this occurs, our claims would be subject to significant delay and we may receive substantially less than our actual damages or nothing at all.

If we are forced to liquidate we may have few unpledged assets for distribution to unsecured creditors.

We have pledged a substantial portion of our assets to secure the repayment of CMOs issued in securitizations, our financing facilities and our other borrowings. We will also pledge substantially all of our current and future mortgages to secure borrowings pending their securitization or sale. The cash flows we receive from our investments that have not yet been distributed, pledged or used to acquire mortgages or other investments may be the only unpledged assets available to our unsecured creditors and you if our Company was liquidated.

Interest rate fluctuations may adversely affect our operating results.

Our operations, as a mortgage loan acquirer and originator or a warehouse lender, may be adversely affected by rising and falling interest rates. Higher interest rates may discourage potential borrowers from refinancing mortgages, borrowing to purchase homes or seeking second mortgages. This may decrease the amount of mortgages available to be acquired or originated by our mortgage operations and decrease the demand for warehouse financing provided by our warehouse lending operations. If short-term interest rates exceed long-term interest rates, there is a higher risk of increased loan prepayments, as borrowers may seek to refinance their fixed and adjustable rate mortgages at lower long-

term fixed interest rates. Increased loan prepayments could lead to a reduction in the number of loans in our investment portfolio and reduce our net interest income.

We are subject to the risk of rising mortgage interest rates between the time we commit to purchase mortgages at a fixed price through the issuance of individual, bulk or other rate-locks and the time we sell or securitize those mortgages. An increase in interest rates will generally result in a decrease in the market value of mortgages that we have committed to purchase at a fixed price, but have not been sold or securitized or have not been properly hedged. As a result, we may record a smaller gain, or even a loss, upon the sale or securitization of those mortgages.

We may experience losses if our liabilities re-price at different rates than our assets.

Our principal source of revenue is net interest income or net interest spread from our investment portfolio, which is the difference between the interest we earn on our interest earning assets and the interest we pay on our interest bearing liabilities. The rates we pay on our borrowings are independent of the rates we earn on our assets and may be subject to more frequent periodic rate adjustments. Therefore, we could experience a decrease in net interest income or a net interest loss because the interest rates on our borrowings could increase faster than the interest rates on our assets. If our net interest spread becomes negative, we will be paying more interest on our borrowings than we will be earning on our assets and we will be exposed to a risk of loss.

Additionally, the rates paid on our borrowings and the rates received on our assets may be based upon different indices. If the index used to determine the rate on our borrowings, typically one-month LIBOR, increases faster than the indices used to determine the rates on our assets, such as six-month LIBOR or the prime rate, we will experience a declining net interest spread, which will have a negative effect on our profitability, and may result in losses.

An increase in our adjustable interest rate borrowings may decrease the net interest margin on our adjustable rate mortgages.

Our long-term investment portfolio includes mortgages that are hybrid ARMs. These are mortgages with fixed interest rates for an initial period of time, after which they begin bearing interest based upon short-term interest rate indices and adjust periodically. We generally fund mortgages with adjustable interest rate borrowings having interest rates that are indexed to short-term interest rates and adjust periodically at various intervals. To the extent that there is an increase in the interest rate index used to determine our adjustable interest rate borrowings and that increase is not offset by a corresponding increase in the rates at which interest accrues on our assets or by various interest rate hedges that we have in place at any given time, our net interest margin will decrease or become negative. We may suffer a net interest loss on our adjustable rate mortgages that have interest rate caps if the interest rates on our related borrowings increase.

Adjustable rate mortgages typically have interest rate caps, which limit interest rates charged to the borrower during any given period. Our borrowings are not subject to similar restrictions. As a result, in a period of rapidly increasing interest rates, the interest rates we pay on our borrowings could increase without limitation, while the interest rates we earn on our adjustable rate mortgage assets would be capped. If this occurs, our net interest spread could be significantly reduced or we could suffer a net interest loss.

Increased levels of prepayments of our adjustable rate mortgages may accelerate our expenses and decrease our net income.

Mortgage prepayments generally increase on our adjustable rate mortgages when fixed mortgage interest rates fall below the then-current interest rates on outstanding adjustable rate mortgages. Prepayments on mortgages are also affected by the terms and credit grades of the mortgages, conditions in the housing and financial markets and general economic conditions. If we acquire mortgages at a premium and they are subsequently repaid, we must expense the unamortized premium at the time of the prepayment. We could possibly lose the opportunity to earn interest at a higher rate over the expected life of the mortgage. Also, if prepayments on our adjustable rate mortgages increase when interest rates are declining, our net interest income may decrease if we cannot reinvest the prepayments in mortgage assets bearing comparable rates. Prepayments on fixed rate mortgages will also decrease our net interest income when interest rates are declining.

We generally acquire mortgages on a servicing released basis, meaning we acquire both the mortgages and the rights to service them. This strategy requires us to pay a higher purchase price or premium for the mortgages. If the mortgages that we acquire at a premium prepay faster than originally projected, generally accepted accounting principles require us to write down the remaining capitalized premium amounts at a faster speed than was originally projected, which would decrease our current net interest income.

The value of our portfolio of mortgage-backed securities may be adversely affected by unforeseen events.

Our prior investments in residual interest and subordinated debt investments exposed us to greater risks as compared to those associated with senior mortgage-backed securities.

Prior to 1998, we invested in mortgage-backed securities known as interest-only, principal-only, residual interest or other subordinated securities. Investments in residual interest and subordinated securities are much riskier than investments in senior mortgage-backed securities because these subordinated securities bear all credit losses prior to the related senior securities. The risk associated with holding residual interest and subordinated securities is greater than that associated with holding the underlying mortgages directly due to the concentration of losses attributed to the subordinated securities.

If the projected value of our portfolio of residual interest and subordinated debt instruments is incorrect we would have to write down the value of these securities.

We estimate future cash flows from these securities and value them utilizing assumptions based in part on projected discount rates, mortgage loan prepayments and credit losses. If our actual experience adversely differs from our assumptions, we would be required to reduce the value of these securities. The market for our asset-backed securities is extremely limited and we cannot assure you that we could sell these securities at their reported value, or at any value or that we could recoup our initial investment.

In addition, we may not obtain our anticipated yield or we may incur losses if the credit support available within certain mortgage-backed securities is inadequate due to unanticipated levels of losses, or due to difficulties experienced by the credit support provider. Delays or difficulties encountered in servicing mortgages securing mortgage-backed securities may cause greater losses than anticipated and, therefore, we may have a higher rate of mortgage insurance claims than were originally anticipated. This may cause a rating agency to downgrade certain classes of our mortgage-backed securities and cause a reduction of the value of the security.

We undertake additional risks by acquiring and investing in mortgages.

We may be subject to losses on mortgages for which we do not obtain credit enhancements.

We do not obtain credit enhancements such as mortgage pool or special hazard insurance for all of our mortgages and investments. Generally, we require mortgage insurance on any mortgage with a loan-to-value ratio greater than 80%. During the time we hold mortgages for investment, we are subject to risks of borrower defaults and bankruptcies and special hazard losses that are not covered by standard hazard insurance. If a borrower defaults on a mortgage that we hold, we bear the risk of loss of principal to the extent there is any deficiency between the value of the related mortgaged property and the amount owing on the mortgage loan and any insurance proceeds available to us through the mortgage insurer. In addition, since defaulted mortgages, which under our financing arrangements are mortgages that are generally 60 to 90 days delinquent in payments, may be considered negligible collateral under our borrowing arrangements, we could bear the risk of being required to own these loans without the use of borrowed funds until they are ultimately liquidated or possibly sold at a loss.

Alt-A mortgages expose us to greater credit risks.

We are an acquirer and originator of Alt-A mortgages. These are residential mortgages that do not qualify for purchase by government sponsored agencies such as Fannie Mae and Freddie Mac. Our operations may be negatively affected due to our investments in Alt-A mortgages. Credit risks associated with Alt-A mortgages are greater than those associated with conforming mortgages. The interest rates we charge on Alt-A loans are often higher than those charged for conforming loans in order to compensate for the higher risk and lower liquidity. Lower levels of liquidity may cause us to hold loans or other mortgage-related assets supported by these loans that we otherwise would not hold. By doing

this, we assume the potential risk of increased delinquency rates and/or credit losses as well as interest rate risk. Additionally, the combination of different underwriting criteria and higher rates of interest leads to greater risk, including higher prepayment rates and higher delinquency rates and/or credit losses.

Lending to Alt-A borrowers may expose us to a higher risk of delinquencies, foreclosures and losses.

As a lender of Alt-A mortgages, our market includes borrowers who may be unable to obtain mortgage financing from conventional mortgage sources. Mortgages made to such Alt-A borrowers generally entail a higher risk of delinquency and higher losses than mortgages made to borrowers who utilize conventional mortgage sources. Delinquency, foreclosures and losses generally increase during economic slowdowns or recessions. The actual risk of delinquencies, foreclosures and losses on mortgages made to Alt-A borrowers could be higher under adverse economic conditions than those currently experienced in the mortgage lending industry in general.

Further, any material decline in real estate values increases the loan-to-value ratios of mortgages previously made by us, thereby weakening collateral coverage and increasing the possibility of a loss in the event of a borrower default. Any sustained period of increased delinquencies, foreclosures or losses after the mortgages are sold could adversely affect the pricing of our future loan sales and our ability to sell or securitize our mortgages in the future. In the past, certain of these factors have caused revenues and net income of many participants in the mortgage industry, including us, to fluctuate from quarter to quarter.

Our use of second mortgages exposes us to greater credit risks.

Our security interest in the property securing second mortgages is subordinated to the interest of the first mortgage holder and the second mortgages have a higher combined loan-to-value ratio than does the first mortgage. If the value of the property is equal to or less than the amount needed to repay the borrower's obligation to the first mortgage holder upon foreclosure, our second mortgage loan will not be repaid.

The geographic concentration of our mortgages increases our exposure to risks in those areas.

We do not set limitations on the percentage of our mortgage asset portfolio composed of properties located in any one area (whether by state, zip code or other geographic measure). Concentration in any one area increases our exposure to the economic and natural hazard risks associated with that area. Historically, a majority of our mortgage acquisitions and originations by the mortgage operations and mortgages held for investment by our long term investment operations were secured by properties in California and, to a lesser extent, Florida. For instance, certain parts of California have experienced an economic downturn in past years and California and Florida have suffered the effects of certain natural hazards. Declines in those residential real estate markets may reduce the values of the properties collateralizing the mortgages, increase foreclosures and losses and have material adverse effect on our results of operations or financial condition.

Furthermore, if borrowers are not insured for natural disasters, which are typically not covered by standard hazard insurance policies, then they may not be able to repair the property or may stop paying their mortgages if the property is damaged. This would cause increased foreclosures and decrease our ability to recover losses on properties affected by such disasters. This would have a material adverse effect on our results of operations or financial condition.

Representations and warranties made by us in our loan sales and securitizations may subject us to liability.

In connection with our securitizations, we transfer mortgages acquired and originated by us into a trust in exchange for cash and, in the case of a CMO, residual certificates issued by the trust. The trustee will have recourse to us with respect to the breach of the standard representations and warranties made by us at the time such mortgages are transferred. While we generally have recourse to our customers for any such breaches, there can be no assurance of our customers' abilities to honor their respective obligations. Also, we engage in bulk whole loan sales pursuant to agreements that generally provide for recourse by the purchaser against us in the event of a breach of one of our representations or warranties, any fraud or misrepresentation during the mortgage origination process, or upon early default on such mortgage. We generally limit the potential remedies of such purchasers to the potential remedies we receive from the people from whom we acquired or originated the mortgages. However, in some cases, the remedies available to a purchaser of mortgages from us may be broader than those available to us against the sellers of the mortgages and should a purchaser enforce its remedies against us, we may not always be able to enforce whatever

remedies we have against our customers. Furthermore, if we discover, prior to the sale or transfer of a loan, that there is any fraud or misrepresentation with respect to the mortgage and the originator fails to repurchase the mortgage, then we may not be able to sell the mortgage or we may have to sell the mortgage only at a discount.

In the ordinary course of our business, we are subject to claims made against us by borrowers and trustees in our securitizations arising from, among other things, losses that are claimed to have been incurred as a result of alleged breaches of fiduciary obligations, misrepresentations, errors and omissions of our employees, officers and agents (including our appraisers), incomplete documentation and our failure to comply with various laws and regulations applicable to our business. Any claims asserted against us may result in legal expenses or liabilities that could have a material adverse effect on our results of operations or financial condition.

We face conflicts of interests based on the ownership of the voting stock of Impac Funding Corporation by certain officers and directors of Impac Mortgage Holdings, Inc.

We are subject to conflicts of interest arising from our relationship with Impac Mortgage Holdings, Inc., our long-term investment operations, Impac Funding Corporation, our mortgage operations, and their officers and directors. Our long-term investment operations acquires Alt-A mortgages from our mortgage operations. Impac Mortgage Holdings, Inc. owns all of the preferred stock, and 99% of the economic interest in, Impac Funding Corporation. Joseph R. Tomkinson, our Chairman and Chief Executive Officer, William S. Ashmore, our Chief Operating Officer, President and a director, and Richard J. Johnson, our Executive Vice President and Chief Financial Officer, are holders of all of the outstanding voting stock of, and 1% of the economic interest in, Impac Funding Corporation. They have the right to elect all directors of Impac Funding Corporation and the ability to control the outcome of all matters for which the consent of the holders of the common stock of Impac Funding Corporation is required. Messer's Tomkinson, Ashmore and Johnson are also the sole directors of Impac Funding Corporation. Decisions made by these officers at one company may be at conflict with and have an adverse affect on the operations of the other. We are currently negotiating with management to purchase the common shares of Impac Funding Corporation, however, we cannot assure you that terms will be agreed upon or that a transaction will be completed.

A substantial interruption in our use of iDASLg2 may adversely affect our level of mortgage loan acquisitions and originations.

We utilize the Internet in our business principally for the implementation of our automated mortgage origination program, iDASLg2, which stands for the second generation of Impac Direct Access System for Lending. iDASLg2 allows our customers to pre-qualify borrowers for various mortgage programs based on criteria requested from the borrower and renders an automated underwriting decision by issuing an approval of the mortgage loan or a referral for further review or additional information. Substantially, all of our correspondents submit mortgages through iDASLg2 and all wholesale mortgages delivered by mortgage brokers are directly underwritten through the use of iDASLg2. iDASLg2 may be interrupted if the Internet experiences periods of poor performance, if our computer systems or the systems of our third-party service providers contain defects, or if customers are reluctant to use or have inadequate connectivity to the Internet. Increased government regulation of the Internet in providing our services is impaired, our ability to originate or acquire mortgages on an automated basis could be delayed or reduced. Furthermore, we rely on a third party hosting company in connection with the use of iDASLg2. If the third party hosting company fails for any reason, and adequate back-up is not implemented in a timely manner, it may delay and reduce those mortgage acquisitions and originations done through iDASLg2. Any substantial delay and reduction in our mortgage acquisitions and originations will reduce our net earnings for the applicable period.

We are subject to risks of operational failure that are beyond our control.

Substantially all of our operations are located in Newport Beach, California and San Diego, California. Our systems and operations are vulnerable to damage and interruption from fire, flood, telecommunications failure, break-ins, earthquake and similar events. Our operations may also be interrupted by power disruptions, including rolling black-outs implemented in California due to power shortages. We do not maintain alternative power sources. Furthermore, our security mechanisms may be inadequate to prevent security breaches to our computer systems, including from computer viruses, electronic break-ins and similar disruptions. Such security breaches or operational failures could expose us to liability, impair our operations, result in losses, and harm our reputation.

Competition for mortgages is intense and may adversely affect our operations.

We compete in acquiring and originating Alt-A mortgages and issuing mortgage-backed securities with other mortgage conduit programs, investment banking firms, savings and loan associations, banks, thrift and loan associations, finance companies, mortgage bankers, insurance companies, other lenders, and other entities purchasing mortgage assets.

We also face intense competition from Internet-based lending companies where entry barriers are relatively low. Some of our competitors are much larger than we are, have better name recognition than we do, and have far greater financial and other resources. Government-sponsored entities, in particular Fannie Mae and Freddie Mac, are also expanding their participation in the Alt-A mortgage industry. These government-sponsored entities have a size and cost-of-funds advantage over us that allows them to price mortgages at lower rates than we are able to offer. This phenomenon may seriously destabilize the Alt-A mortgage industry. In addition, if as a result of their less-conservative, risk-adjusted pricing these government-sponsored entities experience significantly higher-than-expected losses, it would likely adversely affect overall investor perception of the Alt-A mortgage industry because the losses would be made public due to the reporting obligations of these entities.

The intense competition in the Alt-A mortgage industry has also led to rapid technological developments, evolving industry standards and frequent releases of new products and enhancements. As mortgage products are offered more widely through alternative distribution channels, such as the Internet, we may be required to make significant changes to our current retail and wholesale structure and information systems to compete effectively. Our inability to continue enhancing our current Internet capabilities, or to adapt to other technological changes in the industry, could have a material adverse effect on our business, financial condition, liquidity and results of operations.

The need to maintain mortgage loan volume in this competitive environment creates a risk of price competition in the Alt-A mortgage industry. Competition in the industry can take many forms, including interest rates and costs of a loan, less stringent underwriting standards, convenience in obtaining a loan, customer service, amount and term of a loan and marketing and distribution channels. Price competition would lower the interest rates that we are able to charge borrowers, which would lower our interest income. Price-cutting or discounting reduces profits and will depress earnings if sustained for any length of time. If our competition uses less stringent underwriting standards we will be pressured to do so as well, resulting in greater loan risk without being able to price for that greater risk. Our competitors may lower their underwriting standards to increase their market share. If we do not relax underwriting standards in the face of competition, we may lose market share. Increased competition may also reduce the volume of our loan originations and acquisitions. Any increase in these pricing and credit pressures could have a material adverse effect on our business, financial condition, liquidity and results of operations.

We are exposed to potential credit losses in providing warehouse financing.

As a warehouse lender, we lend money to mortgage bankers on a secured basis and we are subject to the risks associated with lending to mortgage bankers, including the risks of fraud, borrower default and bankruptcy, any of which could result in credit losses for us. Our claims as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay.

We may not pay dividends to stockholders.

REIT provisions of the Internal Revenue Code generally require that we annually distribute to our stockholders at least 90% of all of our taxable income. These provisions restrict our ability to retain earnings and thereby renew capital for our business activities. We may decide at a future date to terminate our REIT status, which would cause us to be taxed at the corporate levels and cease paying regular dividends.

In addition, for any year that we do not generate taxable income, we are not required to declare and pay dividends to maintain our REIT status. For instance, due to losses incurred in 2000, we did not declare any dividends from September 2000 until September 2001.

To date, a portion of our taxable income and cash flow has been attributable to our receipt of dividend distributions from IFC. IFC is not a REIT and is not, therefore, subject to the above-described REIT distribution requirements. Because IFC is seeking to retain earnings to fund the future growth of our mortgage operations business, its board of

directors may decide that IFC should cease making dividend distributions in the future. This would materially reduce the amount of our taxable income and in turn, would reduce the amount we would be required to distribute as dividends.

If we fail to maintain our REIT status, we may be subject to taxation as a regular corporation.

We believe that we have operated and intend to continue to operate in a manner that enables us to meet the requirements for qualification as a REIT for federal income tax purposes. We have not requested, and do not plan to request, a ruling from the Internal Revenue Service that we qualify as a REIT.

Moreover, no assurance can be given that legislation, new regulations, administrative interpretations or court decisions will not significantly change the tax laws with respect to qualification as a REIT or the federal income tax consequences of such qualification. Our continued qualification as a REIT will depend on our satisfaction of certain asset, income, organizational and stockholder ownership requirements on a continuing basis.

If we fail to qualify as a REIT, we would not be allowed a deduction for distributions to stockholders in computing our taxable income and would be subject to federal income tax at regular corporate rates. We also may be subject to the federal alternative minimum tax. Unless we are entitled to relief under specific statutory provisions, we could not elect to be taxed as a REIT for four taxable years following the year during which we were disqualified. Therefore, if we lose our REIT status, the funds available for distribution to you would be reduced substantially for each of the years involved. Failure to qualify as a REIT could adversely affect the value of our common stock.

Delayed mortgage loan sales or securitization closings could have a material adverse affect on our operations.

A delay in closing a particular mortgage sale or securitization would increase our exposure to interest rate fluctuations by lengthening the period during which our variable rate borrowings under our warehouse facilities are outstanding. Any disruption in our ability to complete securitizations may require us to utilize other sources of financing, which, if available at all, may be on unfavorable terms. If we were unable to sell a sufficient number of mortgages at a premium during a particular reporting period, our revenues for that period would decline, which could have a material adverse affect on our operations.

Our share prices have been and may continue to be volatile.

Historically, the market price of our common stock has been volatile. During 2002 our stock reached a high of \$13.48 per share on June 28, 2002 and a low of \$7.80 on February 1, 2002. The market price of our common stock is likely to continue to be highly volatile and could be significantly affected by factors including:

- the amount of dividends paid;
- availability of liquidity in the securitization market;
- loan sale pricing;
- calls by warehouse lenders or changes in warehouse lending rates;
- unanticipated fluctuations in our operating results;
- prepayments on mortgages;
- valuations of securitization related assets;
- cost of funds; and
- general market conditions.

In addition, significant price and volume fluctuations in the stock market have particularly affected the market prices for the common stock of mortgage REIT companies such as ours. These broad market fluctuations have



adversely affected and may continue to adversely affect the market price of our common stock. If our results of operations fail to meet the expectations of securities analysts or investors in a future quarter, the market price of our common stock could also be materially adversely affected and we may experience difficulty in raising capital.

If actual prepayments or defaults with respect to mortgages serviced occurs more quickly than originally assumed, the value of our mortgage servicing rights would be subject to downward adjustment.

When we purchase mortgages that include the associated servicing rights, the allocated cost of the servicing rights is reflected on our financial statements as mortgage servicing rights. To determine the fair value of these servicing rights, we use assumptions to estimate future net servicing income including projected discount rates, mortgage loan prepayments and credit losses. If actual prepayments or defaults with respect to loans serviced occur more quickly than we originally assumed, we would have to reduce the carrying value of our mortgage servicing rights. We do not know if our assumptions will prove correct.

Our operating results may be adversely affected by the results of our hedging activities.

To offset the risks associated with our mortgage operations, we enter into transactions designed to hedge our interest rate risks. To offset the risks associated with our long-term investment operations, we attempt to match the interest rate sensitivities of our adjustable rate mortgage assets held for investment with the associated financing liabilities. Our management determines the nature and quantity of the hedging transactions based on various factors, including market conditions and the expected volume of mortgage loan purchases. We do not limit management's use of certain instruments in such hedging transactions. While we believe that we properly hedge our interest rate risk, we may not, and in some cases will not, be permitted to use hedge accounting as established by FASB under the provisions of SFAS 133 to account for our hedging activities. The effect of our hedging strategy may result in some volatility in our quarterly earnings as interest rates go up or down. It is possible that there will be periods during which we will incur losses after the accounting for our hedging activities. In addition, if the counter parties to our hedging transactions are unable to perform according to the terms of the contracts, we may incur losses. While we believe we properly hedge our interest rate risk, we cannot assure you that our hedging transactions will offset the risk of adverse changes in net interest margins.

A reduction in the demand for residential mortgages and our Alt-A loan products may adversely affect our operations.

The availability of sufficient mortgages meeting our criteria is dependent in part upon the size and level of activity in the residential real estate lending market and, in particular, the demand for Alt-A mortgages, which is affected by:

- interest rates;
- national economic conditions;
- residential property values; and
- regulatory and tax developments.

If our mortgage purchases decrease, we will have:

- decreased economies of scale;
- higher origination costs per loan;
- reduced fee income;
- smaller gains on the sale of non-conforming mortgages; and
- an insufficient volume of loans to generate securitizations which thereby causes us to accumulate mortgages over a longer period.

Our delinquency ratios and our performance may be adversely affected by the performance of parties who sub-service our mortgages.

We contract with third-party sub-servicers for the sub-servicing of all the mortgages in which we retain servicing rights, including those in our securitizations. Our operations are subject to risks associated with inadequate or untimely servicing. Poor performance by a sub-servicer may result in greater than expected delinquencies and losses on our mortgages. A substantial increase in our delinquency or foreclosure rate could adversely affect our ability to access the capital and secondary markets for our financing needs. Also, with respect to mortgages subject to a securitization, greater delinquencies would adversely impact the value of any interest-only, equity interest, principal-only and subordinated securities we hold in connection with that securitization.

In a securitization, relevant agreements permit us to be terminated as servicer or master servicer under specific conditions described in these agreements, such as the failure of a sub-servicer to perform certain functions within specific time periods. If, as a result of a sub-servicer's failure to perform adequately, we were terminated as servicer of a securitization, the value of any servicing rights held by us would be adversely affected.

Potential characterization of distributions or gain on sale as unrelated business taxable income to tax-exempt investors.

If (1) all or a portion of our assets are subject to the rules relating to taxable mortgage pools, (2) we are a "pension-held REIT," (3) a tax-exempt stockholder has incurred debt to purchase or hold our common stock, or (4) the residual REMIC interests we buy generate "excess inclusion income," then a portion of the distributions to and, in the case of a stockholder described in (3), gains realized on the sale of common stock by such tax-exempt stockholder may be subject to Federal income tax as unrelated business taxable income under the Internal Revenue Code.

Classification as a taxable mortgage pool could subject us or certain of our shareholders to increased taxation.

If we have borrowings with two or more maturities and, (1) those borrowings are secured by mortgages or mortgage-backed securities and, (2) the payments made on the borrowings are related to the payments received on the underlying assets, then the borrowings and the pool of mortgages or mortgage-backed securities to which such borrowings relate may be classified as a taxable mortgage pool under the Internal Revenue Code. If any part of our Company were to be treated as a taxable mortgage pool, then our REIT status would not be impaired, but a portion of the taxable income we recognize may, under regulations to be issued by the Treasury Department, be characterized as "excess inclusion" income and allocated among our stockholders to the extent of and generally in proportion to the distributions we make to each stockholder. Any excess inclusion income would:

- not be allowed to be offset by a stockholder's net operating losses;
- be subject to a tax as unrelated business income if a stockholder were a tax-exempt stockholder;
- be subject to the application of federal income tax withholding at the maximum rate (without reduction for any otherwise applicable income tax treaty) with respect to amounts allocable to foreign stockholders; and
- be taxable (at the highest corporate tax rate) to us, rather than to our stockholders, to the extent the excess inclusion income relates to stock held by disqualified organizations (generally, tax-exempt companies not subject to tax on unrelated business income, including governmental organizations).

Based on advice of our tax counsel, we take the position that our existing financing arrangements do not create a taxable mortgage pool.

We may be subject to fines or other penalties based upon the conduct of our independent brokers or correspondents.

The mortgage brokers and correspondents from which we obtain loans have parallel and separate legal obligations to which they are subject. While these laws may not explicitly hold the originating lenders responsible for the legal violations of mortgage brokers, increasingly federal and state agencies have sought to impose such liability. Previously, for example, the United States Federal Trade Commission, or "FTC," entered into a settlement agreement with a mortgage lender where the FTC characterized a broker that had placed all of its loan production with a single lender as the "agent" of the lender; the FTC imposed a fine on the lender in part because, as "principal," the lender was legally

responsible for the mortgage broker's unfair and deceptive acts and practices. The United States Justice Department in the past has sought to hold a sub-prime mortgage lender responsible for the pricing practices of its mortgage brokers, alleging that the mortgage lender was directly responsible for the total fees and charges paid by the borrower under the Fair Housing Act even if the lender neither dictated what the mortgage broker could charge nor kept the money for its own account. Accordingly, we may be subject to fines or other penalties based upon the conduct of our independent mortgage brokers or correspondents.

After July, 1, 2003 we may no longer be able to rely on the Alternative Mortgage Transactions Parity Act to preempt certain state law restrictions on prepayment penalties, which may cause us to be unable to compete effectively with financial institutions that are exempt from such restrictions on ARMs.

The value of a mortgage depends, in part, upon the expected period of time that the mortgage will be outstanding. If a borrower pays off a mortgage in advance of this expected period, the holder of the mortgage does not realize the full value expected to be received from the mortgage. A prepayment penalty payable by a borrower who repays a mortgage earlier than expected helps discourage such a prepayment or helps offset the reduction in value resulting from the early payoff. Prepayment penalties are an important feature on the mortgages we acquire or originate.

Certain state laws restrict or prohibit prepayment penalties on mortgages. We have historically relied on the federal Alternative Mortgage Transactions Parity Act, or the "Parity Act," and related regulations issued by the Office of Thrift Supervision, or "OTS," to preempt state limitations on prepayment penalties on ARMs. The Parity Act was enacted to extend to financial institutions other than federally chartered depository institutions the federal preemption which federally chartered depository institutions enjoy. However, on September 25, 2002, the OTS issued final regulations that reduce the scope of the Parity Act preemption. The OTS subsequently delayed the effective date of the final regulations until July 1, 2003. The National Home Equity Mortgage Association has filed a lawsuit against the OTS challenging the OTS's authority to issue the regulations. Unless the court determines prior to July 1, 2003 that the regulations were not validly issued on that date, we will no longer be able to rely on the Parity Act to preempt state restrictions on prepayment penalties. The elimination of this federal preemption could have a material adverse affect on our ability to compete effectively with financial institutions that will continue to enjoy federal preemption of state restrictions on prepayment penalties on ARMs.

Our operations may be adversely affected if we are subject to the Investment Company Act.

We intend to conduct our business at all times so as not to become regulated as an investment company under the Investment Company Act. The Investment Company Act exempts entities that are primarily engaged in the business of purchasing or otherwise acquiring mortgages and other liens on and interests in real estate.

In order to qualify for this exemption we must maintain at least 55% of our assets directly in mortgages, qualifying pass-through certificates and certain other qualifying interests in real estate. Our ownership of certain mortgage assets may be limited by the provisions of the Investment Company Act. If the Securities and Exchange Commission adopts a contrary interpretation with respect to these securities or otherwise believes we do not satisfy the above exception, we could be required to restructure our activities or sell certain of our assets. To insure that we continue to qualify for the exemption we may be required at times to adopt less efficient methods of financing certain of our mortgage assets and we may be precluded from acquiring certain types of higher-yielding mortgage assets. The net effect of these factors will be to lower at times our net interest income. If we fail to qualify for exemption from registration as an investment company, our ability to use leverage would be substantially reduced, and we would not be able to conduct our business as described. Our business will be materially and adversely affected if we fail to qualify for this exemption.

If we conduct future offerings the market price of our securities may be adversely affected.

We may elect to increase our capital resources by making additional public or private offerings of securities in the future. We do not know:

- the actual or perceived effect of these offerings;
- the timing of these offerings;
- the dilution of the book value or earnings per share of our securities then outstanding; and

the effect on the market price of our securities then outstanding.

Sales of additional common stock may adversely affect its market price.

•

To sustain our growth strategy we intend to raise capital through the sale of equity. The sale or the proposed sale of substantial amounts of our common stock in the public market could materially adversely affect the market price of our common stock or other outstanding securities. In December 2001 we filed a shelf registration statement with the SEC, which allows us to sell up to \$300.0 million of securities, including common stock, preferred stock, debt securities and warrants. As of February 28, 2003 we have sold 16,976,082 shares of common stock from our shelf registration statement and we may sell additional securities from the shelf registration statement in the future. We have also registered an aggregate of 2,120,069 shares of common stock in connection with our 2001 Stock Option, Deferred Stock and Restricted Stock Plan, and pursuant to the terms of the plan, approved an increase of 1,500,000 authorized shares, which we plan to register. As of February 28, 2003, our 1995 Stock Option, Deferred Stock Plan had 67,390 shares reserved and available for issuance and that were registered. The sale of a large amount of shares or the perception that such sales may occur, could adversely affect the market price for our common stock or other outstanding securities.

New regulatory laws affecting the mortgage industry may increase our costs and decrease our mortgage origination and acquisition.

The regulatory environments in which we operate have an impact on the activities in which we may engage, how the activities may be carried out, and the profitability of those activities. Therefore, changes to laws, regulations or regulatory policies can affect whether and to what extent we are able to operate profitably. For example, proposed state and federal legislation targeted at predatory lending could have the unintended consequence of raising the cost or otherwise reducing the availability of mortgage credit for those potential borrowers with less than prime- quality credit histories, thereby resulting in a reduction of otherwise legitimate Alt-A lending opportunities. We cannot provide any assurance that these proposed laws, rules and regulations, or other similar laws, rules or regulations, will not be adopted in the future. Adoption of these laws and regulations could have a material adverse impact on our business by substantially increasing the costs of compliance with a variety of inconsistent federal, state and local rules, or by restricting our ability to charge rates and fees adequate to compensate us for the risk associated with certain loans.

We are a defendant in purported class actions and may not prevail in these matters.

Class action lawsuits and regulatory actions alleging improper marketing practices, abusive loan terms and fees, disclosure violations, improper yield spread premiums and other matters are risks faced by all mortgage originators, particularly those in the Alt-A market. We are a defendant in four purported class actions, (including an action that was dismissed but there has been a notice of an appeal) pending in four different states. All allege generally that the loan originator improperly charged fees in violation of various state lending or consumer protection laws in connection with mortgages that we acquired. Although the suits are not identical, they generally seek unspecified compensatory damages, punitive damages, pre- and post-judgment interest, costs and expenses and rescission of the mortgages, as well as a return of any improperly collected fees. These actions are in the early stages of litigation and, accordingly, it is difficult to predict the outcome of these matters. We believe we have meritorious defenses to the actions and intend to defend against them vigorously; however, an adverse judgment in any of these matters could have a material adverse effect on us.

We may be subject to possible adverse consequences as a result of limits on ownership of our shares.

Our charter limits ownership of our capital stock by any single stockholder to 9.5% of our outstanding shares unless waived by the board of directors. Our board of directors may increase the 9.5% ownership limit. In addition, to the extent consistent with the REIT provisions of the Internal Revenue Code, our board of directors may, pursuant to our articles of incorporation, waive the 9.5% ownership limit for a stockholder or purchaser of our stock. In order to waive the 9.5% ownership limit our board of directors must require the stockholder requesting the waiver to provide certain representations to the Company to ensure compliance with the REIT provisions of the Internal Revenue Code. Our charter also prohibits anyone from buying shares if the purchase would result in us losing our REIT status. This could happen if a share transaction results in fewer than 100 persons owning all of our shares or in five or fewer persons, applying certain broad attribution rules of the Internal Revenue Code, owning more than 50% (by value) of our shares.

If you or anyone else acquires shares in excess of the ownership limit or in violation of the ownership requirements of the Internal Revenue Code for REITs, we:

- will consider the transfer to be null and void;
- will not reflect the transaction on our books;
- may institute legal action to enjoin the transaction;
- will not pay dividends or other distributions with respect to those shares;
- will not recognize any voting rights for those shares;
- may redeem the shares; and
- will consider the shares held in trust for the benefit of a charitable beneficiary as designated by us.

The trustee shall sell the shares held in trust and the owner of the excess shares will be entitled to the lesser of:

- (a) the price paid by the owner;
- (b) if the owner did not purchase for the excess shares, the closing price for the shares on the national securities exchange on which the Company is listed on the day of the event causing the shares to be held in trust; or
- (c) the price received by the trustee from the sale of the shares.

Limitations on acquisition and change in control ownership limit.

The 9.5% ownership limit discussed above may have the effect of precluding acquisition of control of our Company by a third party without consent of our board of directors.

ITEM 2. PROPERTIES

Our primary executive and administrative offices are located in Newport Beach, California where we have a 10-year lease expiring May 2008 to use approximately 74,000 square feet of office space. The mortgage operations has other offices located in various states with lease terms ranging from month to month or one to two years with square footage ranging from approximately 93 square feet to approximately 1,185 square feet. NFS has offices in San Diego with lease terms expiring in March 2004 for approximately 21,000 square feet of office space. We believe that these facilities will adequately provide for the near-term future growth needs of the long-term investment operations, mortgage operations and warehouse lending operations.

ITEM 3. LEGAL PROCEEDINGS

On September 1, 2000, a complaint captioned Michael P. and Shellie Gilmor v. Preferred Credit Corporation and Impac Funding Corporation, et al. was filed in the Circuit Court for Clay County, Missouri, Case No. CV100-4263-CC, as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loans Act and Merchandising Practices Act. In July 2001 the Missouri complaint was amended to include IMH and other Impac-related entities. A plaintiffs class was certified on January 2, 2003. Plaintiffs have also filed a motion to certify a defendant class, which motion will be held in abeyance pending resolution by the Missouri Supreme Court of certain issues relating to the applicable statute of limitations. On August 2, 2001, a complaint captioned Frazier, et al. v. Preferred Credit, et al., was filed in the Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004762-01. This case was subsequently removed to federal court, Case No. 01-2714 (W.D. Tenn). This was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration of the ruling. Plaintiffs' motion was denied in its entirety by order dated March 7, 2003. This case is currently dismissed as to all defendants. On August 8, 2001, a complaint captioned Mattie L. Street v. PSB Lending Corp., et al., was filed in the

Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004888-01. This case was subsequently removed to federal court, Case No. 01-2751 (W.D. Tenn.). The Street action was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration. Plaintiffs' motion was denied in its entirety by order dated March 6, 2003. This case is currently dismissed as to all defendants. On October 2, 2001, a complaint captioned Deborah Searcy, Shirley Walker, et al. vs. Impac Funding Corporation, Impac Mortgage Holdings, Inc. et. al. was filed in the Wayne County Circuit Court, State of Michigan, as a purported class action lawsuit alleging that the defendants violated Michigan's Secondary Mortgage Loan Act, Credit Reform Act and Consumer Protection Act. In February 2003, the Court granted a motion to accept for filing an amended complaint adding certain defendants, including an Impac-related entities. Service of the amended complaint has not yet occurred. On October 10, 2001, a complaint captioned Hayes v Impac Funding Corporation, et al was filed in the Circuit Court of Vanderburgh County, Indiana as Case No. 82C01-0110-CP580. This was stated as a purported class action lawsuit alleging a violation of the Indiana Uniform Consumer Credit Code when the loans were originated. This matter was dismissed and the plaintiffs have noticed an appeal to the Indiana Court of Appeals. This appeal remains pending. On November 30, 2001, a complaint captioned Garry Lee Skinner and Judy Cooper Skinner, et al. v. Preferred Credit, et al. was filed in the Superior Court of Durham County, North Carolina as Case No. 1CV-05596. This is stated as a purported class action alleging a violation of the North Carolina Interest Statutes and Unfair and Deceptive Trade Practices Act when the secondary mortgage loans were originated by the defendants. A motion to dismiss the

All of the purported class action lawsuits are similar in nature in that they allege that the mortgage loan originators violated the respective state's statutes by charging excessive fees and costs when making second mortgage loans on residential real estate. The complaints allege that IFC was a purchaser, and is a holder, along with other Impac-related entities, of second mortgage loans originated by other lenders. The plaintiffs in the lawsuits are seeking damages that include disgorgement, restitution, rescission, actual damages, statutory damages, exemplary damages, pre-judgement interest and punitive damages. No material amount of damages are specified in the complaints.

The Company believes that it has meritorious defenses to the above claims and intends to defend these claims vigorously. Nevertheless, litigation is uncertain and the Company may not prevail in the lawsuits and can express no opinion as to its ultimate outcome.

The Company is a party to other litigation and claims, which are normal in the course of its operations. While the results of such other litigation and claims cannot be predicted with certainty, the Company believes the final outcome of such other matters will not have a material adverse effect on the Company.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

No matters were submitted to the security holders to be voted on during the fourth quarter of 2002.

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS

Our common stock is listed on the American Stock Exchange, or "AMEX," under the symbol IMH. The following table summarizes the high, low and closing sales prices for our common stock as reported by the AMEX for the periods indicated:

		2002			2001	
	 High	Low	Close	High	Low	Close
First Quarter	\$ 9.55 \$	7.80 \$	9.42	\$ 4.49	\$ 2.85 \$	4.19
Second Quarter	13.48	8.85	13.48	7.25	3.89	7.15
Third Quarter	13.10	8.16	11.15	8.15	5.76	7.44
Fourth Quarter	12.28	9.08	11.50	9.35	6.85	8.50

On March 26, 2003, the last reported sale price of our common stock on the AMEX was \$12.99 per share. As of March 26, 2003, there were 485 holders of record, including holders who are nominees for an undetermined number of beneficial owners, of our common stock.

Share Repurchase Program. During 2000 our board of directors authorized us to repurchase up to \$3.0 million of our common stock, \$0.01 par value, in open market purchases from time to time at the discretion of management; the timing and extent of the repurchases were dependent on market conditions. During 2001 and 2000 we repurchased 1,015,950 shares of our common stock for \$2.3 million. All repurchased shares were canceled. During 2002 we did not repurchase any shares of our common stock.

Stockholder Rights Plan. Refer to "Note R — Stockholder's Equity" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Securities Authorized for Issuance under Equity Compensation Plans. Refer to Item 12. "Security Ownership of Certain Beneficial Owners and Management."

Dividend Distributions. To maintain our qualification as a REIT, we intend to make annual distributions to stockholders at an amount that maintains our REIT status in accordance with the Internal Revenue Code, which may not necessarily equal net earnings as calculated in accordance with generally accepted accounting principles, or "GAAP." Our dividend policy is subject to revision at the discretion of the board of directors. All distributions in excess of those required to maintain our REIT status will be made at the discretion of the board of directors and will depend on our taxable income, financial condition and other factors as the board of directors deems relevant. The board of directors has not established a minimum distribution level. Distributions to stockholders will generally be taxable as ordinary income, although a portion of such distributions may be designated by us as capital gain or may constitute a tax-free return of capital. We annually furnish to each of our stockholders a statement setting forth distributions paid during the preceding year and their characterization as ordinary income, capital gain or return of capital. Of the total dividends paid during 2002 and 2001, none represented a tax-free return of capital. The following table presents our dividends paid or declared for the periods indicated:

Quarter ended December 31, 2001 Quarter ended March 31, 2002 Quarter ended June 30, 2002 Quarter ended September 30, 2002	Stockholder Record Date		Per Share Dividend Amount
	October 1, 2001 and November 1	l,	
Quarter ended September 30, 2001	2001 (1)	\$	0.25
Quarter ended December 31, 2001	January 2, 2002 (2)	\$	0.44
Quarter ended March 31, 2002	April 3, 2002	\$	0.40
Quarter ended June 30, 2002	July 3, 2002	\$	0.43
Quarter ended September 30, 2002	October 2, 2002	\$	0.45
Quarter ended December 31, 2002	January 2, 2003	\$	0.48

We did not declare a dividend for the quarters ended March 31, 2001 and June 30, 2001.

ITEM 6. SELECTED CONSOLIDATED FINANCIAL DATA

The following selected consolidated statements of operations data for each of the years in the five-year period ended December 31, 2002 and the consolidated balance sheet data for the five-year period ended December 31, 2002 were derived from the consolidated financial statements of Impac Mortgage Holdings, Inc. and Impac Funding Corporation. Such selected financial data should be read in conjunction with the consolidated financial statements and the notes to the consolidated financial statements starting on page F-1 and with Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations."

⁽¹⁾ Impac Mortgage Holdings, Inc. declared a dividend of \$0.25 per share for the third quarter of 2001 payable in two installments: the first installment of \$0.13 was paid on October 15, 2001 to stockholders of record on October 1, 2001, and the second installment of \$0.12 per share was paid on November 15, 2001 to stockholders of record on November 1, 2001.

⁽²⁾ Consisted of a regular cash dividend of \$0.37 per share and a special cash dividend of \$0.07 per share.

IMPAC MORTGAGE HOLDINGS, INC. (dollar amounts in thousands, except per share data)

				For the	year e	ended December 31	,		
		2002		2001		2000	1999		1998
Statement of Operations Data:									
Net interest income:									
Interest income	\$	226,416	\$	156,615	\$	147,079 \$	5 119,458	\$	163,658
Interest expense		144,807		112,012		124,096	89,795		121,695
Net interest income		81,609		44,603		22,983	29,663		41,963
Provision for loan losses		19,848		16,813		18,839	5,547		4,361
Net interest income after loan loss provision		61,761		27,790		4,144	24,116		37,602
Non-interest income:									
Equity in net earnings (loss) of IFC		17,073		10,912		(1,762)	4,292		(13,876)
Equity in net loss of ICH						_			(998)
Loss on sale of mortgages		_							(3,111)
Gain on sale of securities				312			93		427
Other income		4,509		6,155		4,275	2,517		4,019
Total non-interest income (loss)		21,582		17,379		2,513	6,902		(13,539)
Non-interest expense:									
General and administrative and other expenses		7,233		6,261		5,500	4,505		5,081
Write-down on securities available-for-sale		1,039		2,217		53,576	2,037		14,132
(Gain) loss on disposition of real estate owned		154		(1,931)		1,814	2,159		1,707
Mark-to-market loss – SFAS 133				3,821					
Loss on equity investment of ICH		_							9,076
Total non-interest expense		8,426		10,368		60,890	8,701		29,996
Earnings (loss) before extraordinary item and cumulative effect of									
change in accounting principle		74,917		34,801		(54,233)	22,317		(5,933)
Extraordinary item		—		(1,006)		—	—		—
Cumulative effect of change in accounting Principle		—		(617)			—		
Net earnings (loss)	\$	74,917	\$	33,178	\$	(54,233) \$	5 22,317	\$	(5,933)
Net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting principle:									
Basic	\$	1.87	\$	1.41	\$	(2.70) \$	6 0.83	\$	(0.25)
Diluted	\$	1.84	\$	1.25	\$	(2.70) \$	6 0.76	\$	(0.25)
Net earnings (loss) per share:	_							-	
Basic	\$	1.87	\$	1.34	\$	(2.70) \$	5 0.83	\$	(0.25)
Diluted	\$	1.84	\$	1.19	\$	(2.70) \$	6 0.76	\$	(0.25)
Dividends declared per share	\$	1.76	\$	0.69	\$	0.36 \$	5 0.48	\$	1.46
	Ψ	1.70	Ψ	0.05	Ψ	0.00 4	0.40	ψ	1.40

	 As of December 31,							
	 2002	2001	2000	1999	1998			
Balance Sheet Data:								
Investment securities available-for-sale	\$ 26,065 \$	32,989 \$	36,921 \$	93,206 \$	93,486			
Mortgages held-for-investment and CMO collateral	5,207,216	2,249,246	1,389,716	1,313,112	1,181,847			
Finance receivables	1,140,248	466,649	405,438	197,119	311,571			
Investment in Impac Funding Corporation	20,787	19,126	15,762	17,372	13,246			
Due from affiliates	14,500	14,500	14,500	14,500	17,904			

6,551,773	2,854,734	1,898,838	1,675,430	1,665,504
5,041,751	2,151,400	1,291,284	850,817	1,072,316
1,168,029	469,491	398,653	539,687	323,625
6,248,285	2,651,369	1,720,398	1,436,586	1,413,898
303,488	203,365	178,440	238,844	251,606
	5,041,751 1,168,029 6,248,285	5,041,7512,151,4001,168,029469,4916,248,2852,651,369	5,041,7512,151,4001,291,2841,168,029469,491398,6536,248,2852,651,3691,720,398	5,041,7512,151,4001,291,284850,8171,168,029469,491398,653539,6876,248,2852,651,3691,720,3981,436,586

IMPAC FUNDING CORPORATION (dollar amounts in thousands, except Operating Data)

			For the	year ended Decem	ber 31,	
	20	02	2001	2000	1999	1998
Statement of Operations Data:						
Net interest income:						
Total interest income	\$	30,393 \$	24,175	\$ 28,6	49 \$ 21,225	\$ 48,510
Total interest expense		22,125	20,865	30,0	56 20,953	40,743
Net interest income (expense)	_	8,268	3,310	(1,4	07) 272	7,767
Non-interest income:						
Gain (loss) on sale of mortgages		71,064	46,949	19,7	27 27,098	(11,663)
Loan servicing income (expense)		(1,863)	2,140	6,2	86 5,221	7,071
Gain (loss) on sale of investment securities		1,708	3,710		51 —	(706)
Mark-to-market loss on investment securities		—	—	_		(805)
Other income		432	1,295	1,0	54 979	420
Total non-interest income (loss)		71,341	54,094	27,1	18 33,298	(5,683)
Non-interest expense:						
General and administrative and other operating		42 749	20.011	19,6	24 14.065	14 205
Expense		43,748 4,059	28,911 4,519	19,6	,	14,385 6,361
Amortization of mortgage servicing rights Provision for repurchases		1 () () () () () () () () () (79 5,331 71 385	367
Impairment of mortgage servicing rights		2,806 855	3,498 825	S	- 1,078	3,722
Write-down of securities available-for-sale		000	025	1,5		5,722
Mark-to-market (gain) loss - SFAS 133		(1,700)	346	-		_
Total non-interest expense		49,768	38,099	26,7	21 26,011	24,835
Earnings (loss) before income taxes		29,841	19,305	(1,0	10) 7,559	(22,751)
Income taxes (benefit)		12,595	8,300	7	70 3,227	(8,738)
Earnings (loss) before cumulative effect of change in accounting principle		17,246	11,005	(1,7	80) 4,332	(14,013)
Cumulative effect of change in accounting						
Principle		<u> </u>	17			
Net earnings (loss)	\$	17,246 \$	11,022	\$ (1,7	80) \$ 4,332	\$ (14,013)

	 As of December 31,					
	 2002	2001	2000	1999	1998	
Balance Sheet Data:						
Mortgages held-for-sale	\$ 495,877	\$ 174,172	\$ 275,570	\$ 68,084	\$ 252,568	
Mortgage servicing rights	8,274	8,468	10,938	15,621	14,062	
Total assets	575,586	239,802	317,163	116,246	313,872	
Borrowings from IWLG	491,383	174,136	266,994	66,125	192,900	
Other borrowings	—	—	_	181	67,058	
Due to affiliates	14,500	14,500	14,500	14,500	24,382	
Total liabilities	554,589	220,482	301,242	98,698	301,009	
Total shareholders' equity	20,997	19,320	15,921	17,548	12,863	
Operating Data (in millions):						
Mortgage acquisitions and originations	\$ 5,945	\$ 3,204	\$ 2,113	\$ 1,672	\$ 2,249	
Master servicing portfolio at period-end	8,694	5,569	4,043	2,879	3,714	
Servicing portfolio at period-end	2,653	1,754	2,429	2,393	3,714	

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Management's discussion and analysis of financial condition and results of operations contain certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Refer to Item 1. — "Forward-Looking Statements" for a complete description of forward-looking statements.

General

All of our businesses actively work together to deliver comprehensive mortgage investment and lending services to our correspondents, mortgage brokers, retail customers and capital market investors through a wide array of mortgage loan programs, web-based technology and centralized operations so that we can provide high levels of customer service at low costs per loan. We elected to be taxed as a REIT for federal income tax purposes, which generally allows us to pass through income to stockholders without payment of federal income tax at the corporate level. Our goal is to generate consistent reliable income for distribution to our stockholders primarily from the earnings of our core businesses. Refer to Item 1. "Business" for additional information on our businesses and operating segments.

Critical Accounting Policies

We define critical accounting policies as those that are important to the portrayal of our financial condition and results of operations and require estimates and assumptions based on our judgment of changing market conditions and the performance of our assets and liabilities at any given time. In determining which accounting policies meet this definition, we considered our policies with respect to the valuation of our assets and liabilities and estimates and assumptions used in determining those valuations. We believe the most critical accounting issues that require the most complex and difficult judgments and that are particularly susceptible to significant change to our financial condition and results of operations including the following:

- allowance for loan losses; and
- investment securities.

Allowance for loan losses. In evaluating the adequacy of the allowance for loan losses management takes several items into consideration. For instance, a detailed analysis of historical loan performance data is accumulated and reviewed. It is analyzed for loss performance and prepayment performance by product type, origination year and securitization deal. The results of that analysis are then applied to the current mortgage loan investment portfolio and an estimate is created. In accordance with Statement of Financial Accounting Standards No. 5, management believes that pooling of mortgages with similar characteristics is an appropriate methodology in which to evaluate the allowance for loan losses. In addition, management acknowledges that there are mortgages in the mortgage loan investment portfolio that have not been underwritten within the mortgage operations guidelines. As such, management does not have historical loss performance data in which to capture and make a specific allowance for those mortgages. Therefore, management must include an additional loan loss reserve for those mortgages not underwritten within our guidelines.

Management also recognizes that there are qualitative factors that must be taken into consideration when evaluating and measuring potential impairment in the mortgage loan investment portfolio. These items include, but are not limited to, economic indicators that may affect the borrower's ability to pay, political factors and industry statistics. From an industry standpoint, there is a wide range of reserve levels that are maintained by our competitors. For additional information regarding provision for loan losses refer to "Financial Condition" below.

Investment securities. Investment securities consist primarily of subordinated mortgage-backed securities that are classified as available-for-sale and are therefore recorded at fair market value. Any changes in the fair market value of these investment securities are reported as a component of accumulated other comprehensive income in stockholders' equity. To determine the fair value of investment securities available-for-sale, management must estimate future rates of mortgage prepayments, prepayment penalties to be received, delinquency rates, constant default rates and default loss severity and their impact on estimated cash flows. Estimates are based on historical loss data for comparable

mortgages. Management estimates mortgage prepayments by evaluating historical prepayment performance of comparable mortgages and trends in the industry. Management determines the estimated fair value of the residuals by discounting the expected cash flows using a discount rate, which it believes is commensurate with the risks involved. If management's estimates and assumptions used in determining future cash flows on mortgage-backed securities vary significantly from actual cash flows, we may be required to record impairment on the mortgage-backed securities, which would require us to write-down the carrying amount of the mortgage-backed securities through earnings. For additional information regarding fair valuation of investment securities available-for-sale refer to "Note B — Investment Securities Available-for-Sale" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Financial Highlights for 2002

- Earnings per share increased 55% to \$1.84 compared to \$1.19 for 2001;
- Estimated taxable income per share increased 53% to \$2.10 compared to taxable income of \$1.37 for 2001;
- Total assets increased 128% to \$6.6 billion at year-end 2002 from \$2.9 billion for the prior year-end;
- Return on average assets and equity was 1.70% and 28.70%, respectively, as compared to 1.48% and 17.40%, respectively, for 2001;
- Total return to stockholders was 56% based on common stock price appreciation and dividends declared for 2002; and
- Declared total cash dividends of \$1.76 per share compared to \$0.69 per share for 2001.

Results of Operations — Impac Mortgage Holdings, Inc.

Net earnings (loss) was \$74.9 million, or \$1.84 per diluted share, for 2002 compared to \$33.2 million, or \$1.19 per diluted share, for 2001 and \$(54.2) million, or \$(2.70) per diluted share, for 2000. The 2002 year-over-year increase in net earnings of \$41.7 million was primarily due to the following:

- \$37.0 million increase in net interest income; and
- \$6.2 million increase in equity in net earnings of IFC.

The 2001 year-over-year increase in net earnings was primarily due to the following:

- \$51.4 million decrease in write-down of investment securities available-for-sale;
- \$21.6 million increase in net interest income; and
- \$12.7 million increase in equity in net earnings of IFC.

The 2002 and 2001 year-over-year variances are discussed in further detail below in "Net Interest Income," "Non-Interest Income" and "Non-Interest Expense."

Taxable Income

When we file our annual tax returns there are certain adjustments that we make to net earnings and taxable income due to differences in the nature and extent that revenues and expenses are recognized under the two methods. For instance, to calculate taxable income we can only deduct loan loss provisions to the extent that we incurred actual loan losses as compared to net earnings that allows for a complete deduction of loan loss provisions to derive net earnings. As a REIT, we are required to distribute a minimum of 90% of our annual taxable income to our stockholders. Because we pay dividends based on taxable income, dividends may be more or less than net earnings. After adjusting for our estimates of the differences between net earnings and taxable income, estimated taxable income was \$85.6 million, or

\$2.10 per diluted share, for 2002 compared to \$38.3 million, or \$1.37 per diluted share, for 2001 and \$(54.9) million, or \$(2.58) per diluted share, for 2000. We paid total regular and special cash dividends of \$1.72 per share in 2002, which combined with our tax loss carry-forward, met all of our dividend distribution requirements. As of December 31, 2002 estimated federal and state net operating tax loss carry-forwards were \$16.7 million, which expire in the year 2020 and 2010, respectively, that are available to offset future taxable income. Actual net operating tax loss carry-forwards, which may be used to offset future taxable income, will not be determined until we file our 2002 tax return.

The following table presents a reconciliation of net earnings to estimated taxable income for the periods indicated (dollars in thousands, except per share amounts):

 For the yea	r ended, December 31	ļ,
 2002	2001	2000
\$ 74,917 \$	33,178 \$	(54,233)
—	550	—
19,848	16,813	18,839
12,870	8,894	—
(4,938)	(10,211)	(17,778)
(17,073)	(10,912)	(1,762)
 	13	7
\$ 85,624 \$	38,325 \$	(54,927)
\$ 2.10 \$	1.37 \$	(2.58)
\$	2002 \$ 74,917 \$ 19,848 12,870 (4,938) (17,073) \$ 85,624 \$	\$ 74,917 \$ 33,178 \$ 550 19,848 16,813 12,870 8,894 (4,938) (10,211) (17,073) (10,912) 13 \$ 85,624 \$ 38,325 \$

(1) Excludes annual tax deductions of approximately \$11.0 million for amortization of the termination of our management agreement, which occurred in 1997, the deduction for dividends paid and the availability of a deduction attributable to net operating loss carryforwards.

Net Interest Income

We earn interest income primarily on mortgage assets which include CMO collateral, mortgages held-for-investment, finance receivables and investment securities available-for-sale, or collectively, "Mortgage Assets," and, to a lesser extent, interest income earned on cash and cash equivalents and due from affiliates. Interest expense is primarily interest paid on borrowings on Mortgage Assets, which include CMO borrowings, reverse repurchase agreements and borrowings on investment securities available-for-sale, and, to a lesser extent, interest expense paid on due to affiliates and senior subordinated debt. We also receive or make cash payments on derivative instruments as an adjustment to the yield on Mortgage Assets or borrowings on Mortgage Assets depending on whether certain specified contractual interest rate levels are reached.

The following table summarizes average balance, interest and weighted average yield on Mortgage Assets and borrowings on Mortgage Assets for the periods indicated (dollars in thousands):

	For the year ended December 31,										
			2002			2001			2000		
		Average Balance	Interest	Yield	Average Balance	Interest	Yield	Average Balance	Interest	Yield	
MORTGAGE ASSETS											
Subordinated securities collateralized by mortgages	\$	28,931	\$ 1,537	5.31%\$	34,199 \$	3,517	10.28%\$	57,651	\$ 7,249	12.57%	
Subordinated securities collateralized by other loans		_		_	_	_	_	2,871	273	9.51%	
Total investment securities		28,931	1,537	5.31%	34,199	3,517	10.28%	60,522	7,522	12.43%	
Loans receivable:											
CMO collateral		3,387,720	177,389	5.24%	1,519,702	108,328	7.13%	1,198,478	85,923	7.17%	
Mortgages held-for-investment		114,519	5,625	4.91%	137,130	8,188	5.97%	119,326	8,966	7.51%	
Finance receivables:											
Affiliated		404,989	18,044	4.46%	268,718	18,362	6.83%	284,070	28,044	9.87%	
Non-affiliated		341,543	19,383	5.68%	205,474	15,556	7.57%	134,741	14,280	10.60%	
Total finance receivables		746,532	37,427	5.01%	474,192	33,918	7.15%	418,811	42,324	10.11%	
Total loans receivable		4,248,771	220,441	5.19%	2,131,024	150,434	7.06%	1,736,615	137,213	7.90%	
Total Mortgage Assets	\$	4,277,702	\$ 221,978	5.19%\$	2,165,223 \$	5 153,951	7.11%\$	1,797,137	\$ 144,735	8.05%	
BORROWINGS											
CMO borrowings	\$	3,302,988	\$ 117,756	3.57%\$	1,444,033 \$	5 77,813	5.39%\$	1,100,151	80,287	7.30%	
Reverse repurchase agreement		814,248	23,584	2.90%	580,605	30,804	5.31%	513,987	39,216	7.63%	
Borrowings secured by Investment securities		10,037	1,852	18.45%	17,199	2,566	14.92%	26,350	3,217	12.21%	
Total borrowings on Mortgage Assets	\$	4,127,273	\$ 143,192	3.47%\$	2,041,837 \$	5 111,183	5.45%\$	1,640,488	\$ 122,720	7.48%	
Net Interest Spread (1)				1.72%			1.66%			0.57%	
Net Interest Margin (2)				1.84%			1.98%			1.23%	

(1) Calculated by subtracting the yield on total borrowings on Mortgage Assets from the yield on total Mortgage Assets.

(2) Calculated by subtracting interest on total borrowings on Mortgage Assets from interest on total Mortgage Assets and dividing the result by the average balance of total Mortgage Assets.

Net interest income was \$81.6 million for 2002 compared to \$44.6 million for 2001 and \$23.0 million for 2000. The 2002 year-over-year increase in net interest income was primarily due to an increase in average Mortgage Assets, which increased to \$4.3 billion for 2002 compared to \$2.2 billion for 2001 as the long-term investment operations acquired \$3.9 billion of primarily Alt-A mortgages from the mortgage operations. The increase in net interest income was offset, in part, by a decrease in net interest margins on Mortgage Assets which declined to 1.84% for 2002 compared to 1.98% for 2001. The decrease in net interest margins on Mortgage Assets which declined to a nortgage held as CMO collateral and CMO borrowings as follows:

- interest rate adjustment limitations on mortgages held as CMO collateral due to periodic and lifetime interest rate cap features as compared to no such interest rate adjustment limitations on CMO borrowings; and
- mismatched interest rate adjustment periods between mortgages held as CMO collateral and CMO borrowings.

Net interest margins on Mortgage Assets declined primarily as net interest margins on CMO collateral declined 25 basis points to 1.76% for 2002 as compared to 2.01% for 2001. This decline was due to a number of factors, most of which were affected by a decrease in short-term interest rates during 2002 to historical lows, as follows:

- interest rate adjustments on mortgages;
- early principal prepayment of higher coupon mortgages;
- net cash payments on derivative instruments; and

Interest rate adjustments on mortgages. Interest rates on seasoned LIBOR ARMs, which secure adjustable rate CMO borrowings and are subject to periodic interest rate caps, have reset lower over the last two years as a result of short-term interest rate reductions by the Federal Reserve Bank during 2002 and 2001 which have, in effect, "caught up" to interest rate reductions on adjustable rate CMO borrowings, which are not subject to periodic interest rate caps. The result is that the interest rate spread differential between LIBOR ARMs, which are primarily indexed to six-month LIBOR, and adjustable rate CMO borrowings, which are primarily indexed to one-month LIBOR, has narrowed or compressed. In addition, many seasoned hybrid ARMs with initial fixed interest rate periods subsequently converted to LIBOR ARMs, which are primarily indexed to six-month LIBOR, in 2002 which resulted in those mortgages resetting to lower adjustable interest rates.

Early principal prepayment of higher coupon mortgages. Record refinancing activity in 2002 and 2001 resulted in the early principal payment of mortgages with higher interest rates than interest rates on new mortgages that were acquired by the long-term investment operations. As a result of interest rate resets and the early principal prepayment of higher coupon mortgages, the weighted average coupon of mortgages held as CMO collateral was 6.57% at year-end 2002 compared to 7.92% at year-end 2001 and 9.33% at year-end 2000. However, we believe that the high concentration of mortgages held as CMO collateral with prepayment penalty features helps to mitigate future mortgage prepayments. For 2002 the constant prepayment rate, or "CPR," of mortgages held as CMO collateral was 25% CPR compared to 34% CPR for 2001 and 25% CPR for 2000, all of which we believe are below industry prepayment rates.

Net cash payments on derivative instruments. The notional amount of derivative instruments rose as we acquired derivative instruments as interest rate hedges for CMO borrowings, which were used to finance the acquisition of \$3.9 billion of mortgages by the long-term investment operations. The increase in the notional amount of derivative instruments combined with the continued decline of short-term interest rates during 2002 increased net cash payments made on derivative instruments. We acquire derivative instruments to offset possible increased CMO borrowing costs from rising interest rates. However, as short-term interest rates decline, net interest margin compression may occur as the interest rate spread differential between the interest rate index, i.e. one-month LIBOR, and the contractual interest rate level, or strike price, which one-month LIBOR needs to reach in order for us to receive cash payments, widens. As a result of these factors, net cash payments on derivative instruments were \$27.8 million for 2002 compared to \$5.2 million for 2001. Excluding the impact of net cash payments on derivative instruments, net interest margins on Mortgage Assets would have been 2.49% for 2002 as compared to 2.22% for 2001. The goal of our interest rate hedging policy is to provide stable net interest margins and cash flows to our investors in various interest rate environments. We believe that our interest rate hedging policy is sound and net interest margins will remain relatively stable when and if interest rates rise. For additional information regarding interest rate hedging activities refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk."

Higher leverage. Because of the historically favorable prepayment and loss rates of our high credit quality Alt-A mortgages, we have received favorable credit ratings on our CMOs from credit rating agencies, which has reduced our required initial capital investment in CMOs. As our required initial investment of interest-free capital in CMOs declines as a percentage of mortgages securing the CMOs, referred to as the "haircut," net interest margins decline as well. Leverage on CMO collateral increased to 48 to 1 at year-end 2002 compared to 29 to 1 at year-end 2001. However, since CMOs are classes of bonds that are sold to investors in mortgage-backed securities, CMOs are not subject to the same margin call risk that reverse repurchase agreements are subject to. With increased leverage, we have been able to grow our balance sheet by efficiently using available capital. For additional information regarding leverage refer to "Financial Condition" below.

The 2001 year-over-year increase in net interest income was primarily due to an increase in net interest margins on Mortgage Assets and, to a lesser extent, an increase in average Mortgage Assets. Net interest margins on total Mortgage Assets increased 75 basis points to 1.98% for 2001 compared to 1.23% for 2000 as average Mortgage Assets increased to \$2.2 billion at year-end 2001 compared to \$1.8 billion at year-end 2000. As interest rates declined rapidly during 2001 as the Federal Reserve Bank lowered short-term interest rates in hopes of providing stimulus to a lackluster economy, which was exacerbated by the events of September 11, 2001, our adjustable rate CMO borrowing costs, which are indexed to one-month LIBOR and not subject to periodic interest rate caps, declined at a faster pace than coupons on mortgages held as CMO collateral, which were primarily hybrid ARMs with fixed interest rates or were LIBOR ARMs that are subject to periodic interest rate caps. As a result, net interest margins on CMO collateral increased 154 basis points to 2.01% for 2001 compared to 0.47% for 2000. The benefit on net interest margins from

declining short-term interest rates was partially offset by net cash payments on derivative instruments of \$5.2 million for 2001 compared to net cash received of \$237,000 for 2000.

Provision for Loan Losses

Provision for loan losses was \$19.8 million for 2002 compared to \$16.8 million for 2001 and \$18.8 million for 2000. The provision for estimated loan losses is primarily based on a migration analysis based on historical loss statistics, including cumulative loss percentages and loss severity, of similar loans in our mortgage loan investment portfolio. The loss percentage is used to determine the estimated inherent losses in the mortgage loan investment portfolio. Provision for loan losses is also determined based on the following:

- management's judgment of the net loss potential of mortgages in our mortgage loan investment portfolio based on prior loan loss experience;
- changes in the nature and volume of the mortgage loan investment portfolio;
- value of the collateral;
- delinquency trends; and
- current economic conditions that may affect the borrowers' ability to pay.

The 2002 year-over-year increase in provision for loan losses was due to growth in the mortgage loan investment portfolio as the long-term investment operations acquired \$3.9 billion of Alt-A mortgages from the mortgage operations in 2002, which warranted a corresponding increase in allowance for loan losses. The 2001 year-over-year decrease in provision for loan losses was due to high LTV second trust deeds that were provided for and written-off in 2000, which reduced the level of exposure to loan loss provisions required for that portfolio. For additional information regarding provision for loan losses refer to "Financial Condition" below.

Non-Interest Income

Non-interest income primarily consists of equity in net earnings (loss) of IFC, net warehouse fees from warehouse advances made to affiliated and nonaffiliated customers of IWLG and other revenue. Equity in net earnings (loss) of IFC results as IMH records 99% of the earnings or losses from IFC due to IMH's ownership of 100% of IFC's preferred stock, which represents 99% of the economic interest in IFC. Non-interest income was \$21.6 million for 2002 compared to \$17.4 million for 2001 and \$2.5 million for 2000. The year-over-year increases in non-interest income were primarily due to increases of \$6.2 million and \$12.7 million, respectively, in equity in net earnings of IFC as IFC's net earnings rose. For additional information and detail regarding IFC's financial condition and results of operations refer to "Results of Operations — Impac Funding Corporation" below.

Non-Interest Expense

Non-interest expense was \$8.4 million for 2002 compared to \$10.4 million for 2001 and \$60.9 million for 2000. The 2002 year-over-year decrease of \$2.0 million in non-interest expense was primarily due to the following:

- \$3.8 million decrease in mark-to-market loss SFAS 133;
- \$1.2 million decrease in write-down of investment securities available-for-sale; and
- \$2.1 million decrease in gain on disposition of other real estate owned.

The \$3.8 million decrease in mark-to-market loss – SFAS 133 was the result of a further interpretation of SFAS 133 by the Derivative Implementation Group, or "DIG," of FASB, which issued DIG G20 and which we adopted on October 1, 2001, that allows for the entire change in the market value of our derivative instruments to be recognized as other comprehensive income in stockholder's equity. Prior to the adoption of DIG G20 on October 1, 2001, the entire change in market value of derivative instruments was reflected as mark-to-market loss – SFAS 133, which was reflected



in our 2001 results of operations. The \$1.2 million decrease in write-down of investment securities available-for-sale was due to the improved quality and performance of mortgage-backed securities that remain in our investment securities portfolio after write-downs taken during 2000 on investment securities secured by mortgages not underwritten and acquired by the mortgage operations. Since 1998 our investment strategy has been to only acquire or invest in securities underwritten and purchased by our mortgage operations due to their superior historical performance. The \$2.1 million decrease in gain on disposition of other real estate owned was the result of estimates of write-downs on foreclosed property to a percentage of appraised value or a real estate broker's price opinion which were more in line with actual results upon the sale of the foreclosed property during 2002 than during 2001.

The 2001 year-over-year decrease of \$50.5 million in non-interest expense was primarily due to a \$51.4 million decrease in write-down of investment securities available-for-sale as we took a charge of \$52.6 million to write-off substantially all remaining book value of certain investment securities in 2000. The investment securities written-off included investment securities secured by high LTV second trust deeds, investment securities secured by franchise loans receivable and certain sub-prime subordinated securities, all of which were acquired prior to 1998. For additional information regarding the investment securities portfolio refer to Item 1.— "Investment Securities Portfolio" and "Note B — Investment Securities Available-for-Sale" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Financial Condition

Total assets grew 128% to \$6.6 billion at year-end 2002 as the long-term investment operations acquired \$3.9 billion of primarily adjustable and fixed rate Alt-A mortgages from the mortgage operations and retained \$25.8 million of multi-family mortgages originated by IMCC. The acquisition and retention of Alt-A mortgages increased CMO collateral and mortgages held-for-investment to \$5.2 billion at year-end 2002 while strong mortgage demand from affiliates and non-affiliates resulted in finance receivables increasing to \$1.1 billion. Our investment securities portfolio declined to \$26.1 million, or 0.40% of total assets, at year-end 2002 compared to \$33.0 million, \$36.9 million and \$93.2 million at year-end 2001, 2000 and 1999, respectively. Our investment securities portfolio declined since year-end 1999 as we wrote-off a substantial portion of our investment securities in 2000 and we have acquired no investment securities since 1999.

The following table presents selected financial data for the periods indicated (dollars in thousands, except per share data):

	 As of and for the year ended, December 31,					
	 2002	2001	2000			
Book value per share	\$ 6.70 \$	6.35 \$	6.67			
Return on average assets	1.70%	1.48%	(2.93)%			
Return on average equity	28.70%	17.40%	(27.04)%			
Assets to equity ratio	21.59:1	14.04:1	10.64:1			
Debt to equity ratio	20.48:1	12.95:1	9.63:1			
Allowance for loan losses as a percentage of loans receivable	0.42%	0.43%	0.28%			
CPR on mortgages held as CMO collateral	25%	34%	25%			
Total non-performing assets.	\$ 130,614 \$	69,273 \$	43,664			
Total non-performing assets to total assets	1.99%	2.43%	2.30%			
Mortgages owned 60+ days delinquent	\$ 161,260 \$	82,956 \$	64,208			
60+ day delinquency of mortgages owned	3.22%	3.86%	4.89%			

We believe that in order for us to generate positive cash flows and earnings we must successfully manage the following primary operational and market risks:

- credit risk;
- prepayment risk;
- liquidity risk; and
- interest rate risk.

Credit Risk. We manage credit risk by acquiring for long-term investment high credit quality Alt-A mortgages with conservative LTV ratios that are acquired and originated by our mortgage operations, adequately reserving for loan losses and actively managing delinquencies and defaults. We believe that by improving the overall credit quality of our mortgage loan investment portfolio we can consistently generate stable future cash flow and earnings. During 2002 we acquired primarily adjustable and fixed rate Alt-A mortgages from the mortgage operations with an original weighted average credit score of 688 and an original weighted average LTV ratio of 82%. Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but that have loan characteristics that make them non-conforming under those guidelines. We primarily acquire non-conforming "A" or "A-" credit quality mortgages, collectively, Alt-A mortgages. As defined by us, A credit quality mortgages generally have a credit score of between 600 and 639. As a comparison, Fannie Mae and Freddie Mac generally purchase conforming mortgages with credit scores greater than 620. By year-end 2002, the original weighted average credit score of mortgages held as CMO collateral was 683 and the original weighted average LTV ratio was 82%. For additional information regarding the mortgage loan investment portfolio refer to Item 1. " — Mortgage Loan Investment Portfolio" and "Note C — Mortgages Held-for-Investment" and "Note D — CMO Collateral" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

In addition to acquiring mortgages from our mortgage operations, the long-term investment operations originated \$25.8 million of multi-family mortgages through IMCC. IMCC was formed to primarily originate small balance multi-family mortgages with high credit quality, conservative LTV ratios and adjustable rate mortgages with balances ranging from \$250,000 to \$1.5 million. Multi-family mortgages provide greater asset diversification on our balance sheet as multi-family mortgages typically have higher interest rate spreads and longer lives than residential mortgages. All multi-family mortgages originated during 2002 had interest rate floors with prepayment penalty periods ranging from three to five years with the borrower's weighted average credit score of 731 and the weighted average LTV ratio of 67%.

We believe that we have adequately reserved for loan losses by maintaining a ratio of allowance for loan losses to total loans receivable, which includes CMO collateral, finance receivables, and mortgages held-for-investment, of 42 basis points at year-end 2002 as allowance for loan losses increased to \$26.6 million. As a comparison, at year-end 2002 the annualized trailing 12-month and life-to-date loss rates of mortgages held as CMO collateral was 25 basis points and 11 basis points, respectively. These loss rates include mortgages held as CMO collateral acquired prior to 2001 with lower credit scores than subsequent Alt-A mortgages acquired under our current Alt-A guidelines which have higher credit scores.

The following table summarizes annualized trailing 12-month and life-to-date loss rates on mortgages held as CMO collateral for specific CMOs as follows (dollars in thousands, annualized trailing 12-month and life-to-date loss rates in basis points of beginning collateral balance):

Issue Date	Issue Name	 Original Collateral Balance	Current Collateral Balance	Annualized Trailing 12-Mo Loss Rate	Life-to-Date Loss Rate
Jan 1998	Imperial CMB Trust Series 1998-1	\$ 373,022	\$ 54,172	47	28
Mar 1998	Impac CMB Trust Series 1998-2	225,804	38,128	32	36
Jan 2000	Impac CMB Trust Series 2000-1	460,004	110,510	48	40
Nov 2000	Impac CMB Trust Series 2000-2	497,076	150,432	52	46
May 2001	Impac CMB Trust Series 2001-1	359,643	199,256	12	14
Aug 2001	Impac CMB Trust Series 2001-2	403,480	267,669	0	0
Oct 2001	Impac CMB Trust Series 2001-3	399,997	303,538	6	5
Dec 2001	Impac CMB Trust Series 2001-4	350,000	273,701	12	12

All CMOs issued after Impac CMB Trust Series 2001-4 are unseasoned pools and have not experienced material losses as of year-end 2002. Actual net loan losses on the mortgage loan investment portfolio declined to \$4.9 million for 2002 compared to \$10.2 million for 2001 and \$17.8 million for 2000.

We monitor our sub-servicers to make sure that they perform loss mitigation, foreclosure and collection functions according to our servicing guide. This includes an effective and aggressive collection effort in order to minimize mortgages from becoming non-performing assets. However, when resolving non-performing assets, sub-servicers are

required to take timely and aggressive action. The sub-servicer is required to determine payment collection under various circumstances, which will result in maximum financial benefit. This is accomplished by either working with the borrower to bring the mortgage current or by foreclosing and liquidating the property. We perform ongoing review of mortgages that display weaknesses and believe that we maintain adequate loss allowance on the mortgages. When a borrower fails to make required payments on a mortgage and does not cure the delinquency within 60 days, we generally record a notice of default and commence foreclosure proceedings. If the mortgage is not reinstated within the time permitted by law for reinstatement, the property may then be sold at a foreclosure sale. In foreclosure sales, we generally acquire title to the property. As of year-end 2002 our mortgage loan investment portfolio included 3.22% of mortgages that were 60 days or more delinquent compared to 3.86% as of year-end 2001 and 4.89% as of year-end 2000.

The following table summarizes mortgages in our long-term mortgage loan investment portfolio that were 60 or more days delinquent for the periods indicated (in thousands):

	_	As of December 31,					
	_	2002	2001	2000			
60-89 days delinquent	\$	41,762	\$ 21,820	\$ 25,213			
90 or more days delinquent		33,822	21,416	10,597			
Foreclosures		74,597	32,979	16,994			
Delinquent bankruptcies		11,079	6,741	11,404			
Total 60 or more days delinquent	- \$	161,260	\$ 82,956	\$ 64,208			
	_	-					

Non-performing assets consist of mortgages that are 90 days or more delinquent, or "non-accrual loans," including loans in foreclosure and delinquent bankruptcies. It is our policy to place a mortgage on non-accrual status when it becomes 90 days delinquent. Any previously accrued interest will be reversed from earnings. When real estate is acquired in settlement of loans, called other real estate owned, the mortgage is written-down to a percentage of the property's appraised value or broker's price opinion. As of year-end 2002 non-performing assets as a percentage of total assets was 1.99% compared to 2.43% as of yearend 2001 and 2.30% as of year-end 2000.

The following table summarizes mortgages in our long-term mortgage loan investment portfolio that were non-performing for the periods indicated (in thousands):

	 As of December 31,						
	 2002	2001	2000				
Non-accrual	\$ 119,498 \$	61,136 \$	38,995				
Other real estate owned	 11,116	8,137	4,669				
Total non-performing assets	\$ 130,614 \$	69,273 \$	43,664				

Prepayment Risk. 79% of Alt-A mortgages acquired from the mortgage operations in 2002 had prepayment penalty features ranging from two to five years and at year-end 2002 76% of mortgages held as CMO collateral had active prepayment penalties. For 2002 the CPR of mortgages held as CMO collateral was 25% compared to 34% CPR for 2001 and 25% CPR for 2000, all of which we believe are below industry prepayment rates.

Liquidity Risk. We employ a leveraging strategy to increase assets by financing our mortgage loan investment portfolio primarily with CMO borrowings, reverse repurchase agreements and capital and then using cash proceeds to acquire additional mortgage assets. We acquire adjustable and fixed rate mortgages from the mortgage operations and finance the acquisition of those mortgages, during this accumulation period, with reverse repurchase agreements. After accumulating a pool of adjustable and fixed rate mortgages, generally between \$200 million and \$600 million, we securitize the mortgages in the form of CMOs. Our strategy is to securitize our mortgages every 30 to 45 days in order to reduce the accumulation period that mortgages are outstanding on short-term warehouse or reverse repurchase facilities, which reduces our exposure to margin calls on these facilities. CMOs are classes of bonds that are sold to investors in mortgage-backed securities and as such are not subject to margin calls. In addition, CMOs generally require a smaller haircut or initial cash investment as a percentage of mortgages financed than does interim warehouse and reverse repurchase financing. For additional information regarding financing refer to Item 1. "— Financing."

Because of the historically favorable prepayment and loss rates of our high credit quality Alt-A mortgages, we have received favorable credit ratings on our CMOs from credit rating agencies, which has reduced our required initial capital investment. The ratio of total assets to total equity, or "leverage ratio," was 21.59 to 1 at year-end 2002 compared to 14.04 to 1 at year-end 2001. However, after excluding CMO collateral and CMO borrowings, our leverage ratio was 7.17 to 1 at year-end 2002 compared to 4.98 to 1 at year-end 2001. With increased leverage, we have been able to grow our balance sheet by efficiently using available capital. We continually monitor our leverage ratio and liquidity levels to insure that we are adequately protected against adverse changes in market conditions. For additional information regarding liquidity refer to "Liquidity and Capital Resources" below.

Interest Rate Risk. Refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk."

Results of Operations — Impac Funding Corporation

Net earnings (loss) was \$17.2 million for 2002 compared to \$11.0 million for 2001 and \$(1.8) million for 2000. The 2002 year-over-year increase of \$6.2 million was primarily due to the following:

- \$24.1 million increase in gain on sale of loans;
- \$5.0 million increase in net interest income;
- \$16.0 million increase in non-interest expense and income taxes; and
- \$6.9 million decrease in loan servicing income and other non-interest income.

The 2001 year-over-year increase of \$12.8 million was primarily due to the following:

- \$27.2 million increase in gain on sale of loans;
- \$4.7 million increase in net interest income; and
- \$18.9 million increase in non-interest expense and income taxes.

Net Interest Income

Net interest income (expense) was \$8.3 million for 2002 compared to \$3.3 million for 2001 and \$(1.4) million for 2000. The 2002 year-over-year increase in net interest income was primarily due to an increase in average mortgages held-for-sale, which increased to \$420.9 million for 2002 compared to \$277.3 million for 2001 as the mortgage operations acquired \$5.9 billion of primarily Alt-A mortgages in 2002. The increase in net interest income was offset, in part, by a decrease in net interest margins on mortgages held-for-sale, which declined to 1.12% for 2002 compared to 1.59% for 2001. Net interest margins declined during 2002 as the mortgage operations acquired a larger percentage of ARMs, which generally have lower coupons than those of FRMs, than were acquired during 2001. The mortgage operations acquired \$2.4 billion of LIBOR ARMs, or 41% of total acquisitions, during 2002 compared to \$410.0 million of LIBOR ARMs, or 13% of total acquisitions, during 2001.

The 2001 year-over-year increase in net interest income was primarily due to an increase in net interest margins on mortgages held-for-sale, which improved to 1.59% for 2001 compared to (0.02)% for 2000. Net interest margins improved for 2001 primarily due to the following:

- widening yield curve;
- interest rate on warehouse financing was reduced from prime to prime minus 0.50%; and
- short-term interest rates rapidly declined during 2001 which reduced borrowing costs at a faster pace than mortgage interest rates declined.



The improvement in net interest margins was, in part, offset by a decrease in average mortgages held-for-sale, which declined to \$277.3 million for 2001 compared to \$294.6 million for 2000 as the mortgage operations sold and securitized mortgages more frequently during 2001 than during 2000. Even though loan production increased 52% to \$3.2 billion for 2001 compared to \$2.1 billion for 2000, average mortgages held-for-sale declined as more frequent loan sales reduced interest rate risk and price volatility during the mortgage accumulation period.

Non-Interest Income

Non-interest income was \$71.3 million for 2002 compared to \$54.1 million for 2001 and \$27.1 million for 2000. The 2002 year-over-year increase in non-interest income of \$17.2 million was primarily due to the following:

- \$24.1 million increase in gain on sale of loans; and
- \$6.9 million decrease in loan servicing income and other non-interest income.

Gain on sale of loans increased to \$71.1 million for 2002 compared to gain on sale of loans of \$46.9 million for 2001. Gain on sale of loans includes the difference between the price we acquire and originate mortgages and the price we receive on loan sale or securitization and direct mortgage origination revenue and cost, i.e. loan and underwriting fees, commissions, appraisal review fees, document expense, etc. Gain on sale of loans was 127 basis points on total loan sales volume of \$5.6 billion for 2002 compared to 146 basis points on total loan sales volume of \$3.2 billion for 2001. All mortgages were sold by the mortgage operations on a servicing released basis during 2002 and 2001 which results in 100% cash gains upon sale or securitizations, thereby reducing our exposure to interest rate risk and price volatility during the accumulation period of mortgages. For 2002 the mortgage operations sold and transferred \$4.6 billion of mortgages compared to \$2.9 billion for 2001.

The following table summarizes the principal balance of mortgage sales by the mortgage operations for the periods indicated:

	 For the year ended, December 31,					
	 2002	2001	2000			
REMICs	\$ 599,952 \$	1,625,492 \$	1,300,802			
Whole loan sales to third party investors	1,114,246	96,294	62,614			
Sales to the long-term investment operations	3,891,257	1,486,343	450,712			
Total sales	\$ 5,605,455 \$	3,208,129 \$	1,814,128			

Loan servicing income and other non-interest income decreased to \$277,000 for 2002 compared to \$7.1 million for 2001 primarily due to a \$4.0 million decrease in loan servicing income. Loan servicing income primarily decreased as interest earned from float on principal and interest payments declined by \$2.8 million as short-term interest rates declined during 2002. In addition, we continued to sell mortgage servicing rights during 2002 in conjunction with the sale or securitization of mortgages. Gain on sale of mortgage servicing rights is a component of gain on sale of loans. We believe that using proceeds from the sale of mortgage servicing rights to acquire and originate mortgages for subsequent sale and securitization provides a higher return on investment and greater profitability than income received from mortgage servicing functions.

The 2001 year-over-year increase in non-interest income was primarily due to a \$27.2 million increase in gain on sale of loans, which increased to \$46.9 million for 2001 compared to \$19.7 million for 2000. Gain on sale of loans was 146 basis points on total loan sales volume of \$3.2 billion for 2001 compared to 109 basis points on total loan sales volume of \$1.8 billion for 2000. For 2001 the mortgage operations sold and transferred \$2.9 billion of mortgages compared to \$1.3 billion for 2000. Loan servicing income decreased to \$2.1 million for 2001 compared to \$6.3 million for 2000 as loan servicing fees declined as mortgage servicing rights were sold and interest earned from float declined as interest rates declined during 2001.

Non-Interest Expense

Non-interest expense and income taxes was \$62.4 million for 2002 compared to \$46.4 million for 2001 and \$27.5 million for 2000. The 2002 year-overyear increase in non-interest income of \$16.0 million was primarily due to the following:

- \$14.1 million increase in operating costs;
- \$4.3 million increase in income taxes; and
- \$2.0 million decrease in mark-to-market loss SFAS 133.

The \$14.1 million increase in operating costs, which include personnel expense, professional services, equipment expense, occupancy expense, data processing expense and general and administrative and other expense, was primarily due to loan production growth. Operating costs increased 48% to \$42.9 million for 2002 compared to \$28.9 million for 2001 while mortgage acquisition and origination growth increased 84% year-over-year. We believe that our efficient centralized operating structure and our web-based automated underwriting system, iDASLg2, allows us to maintain our position as a low cost nationwide acquirer and originator of Alt-A mortgages.

The following table summarizes fully-loaded mortgage production costs to acquire or originate a single mortgage by production channel for the periods indicated (in basis points of mortgage acquisitions or originations):

		Fe	For the year ended, December 31,						
Production Channel	Company	2002	2001	2000					
Correspondent acquisitions	Impac Funding Corporation	66.17	86.20	84.60					
Wholesale/retail originations	Impac Lending Group	98.84	120.01	159.85					
B/C originations	Novelle Financial Services, Inc.	231.18	278.61						
Total Weighted Average Cost		83.51	98.92	94.60					

Fully-loaded cost to acquire or originate a single mortgage is based on mortgage production costs, including sales commissions which are a component of gain on sale of loans in the financial statements, and corporate administrative costs, including human resources, accounting, management information systems, corporate administration and marketing. Mortgage production costs exclude other non-recurring, non-production related expenses, including amortization and impairment of mortgage servicing rights, mark-to-marked gain (loss) from SFAS 133 and write-down of investment securities. For a breakdown of mortgage acquisitions and originations by production channel, refer to Item 1. "— Mortgage Operations."

The 2001 year-over-year increase in non-interest expense of \$18.9 million was primarily due to the following:

- \$9.3 million increase in operating costs;
- \$7.5 million increase in income taxes;
- \$3.1 million increase in provision for repurchases; and
- \$1.5 million decrease in write-down of investment securities.

The \$9.3 million increase in operating costs was primarily due to loan production growth. Operating costs increased 47% to \$28.9 million for 2001 compared to \$19.6 million for 2000 while mortgage acquisition and origination growth increased 52% year-over-year. The \$3.1 million increase in provision for repurchases was primarily the result of the following:

- provision for mortgage loan repurchases on \$271.9 million of mortgages brokered by Impac Mortgage Acceptance Corporation, or "IMAC," a division of the mortgage operations; and
- fraud losses on mortgages sold to third party investors which we subsequently repurchased.
 - 44

Liquidity and Capital Resources

We recognize the need to have funds available for our operating businesses and our customer's demands for obtaining short-term warehouse financing until the settlement or sale of mortgages with us or with other investors. It is our policy to have adequate liquidity at all times to cover normal cyclical swings in funding availability and mortgage demand and to allow us to meet abnormal and unexpected funding requirements. We plan to meet liquidity through normal operations with the goal of avoiding unplanned sales of assets or emergency borrowing of funds. Toward this goal, ALCO is responsible for monitoring our liquidity position and funding needs.

ALCO is comprised of the senior executives of the mortgage operations and warehouse lending operations. ALCO meets on a weekly basis to review current and projected sources and uses of funds. ALCO monitors the composition of the Company's balance sheet for changes in the liquidity of our assets. Our liquidity consists of cash and cash equivalents, short-term and marketable investment securities rated AAA through BBB and maturing mortgages, or "liquid assets." Our policy is to maintain a liquidity threshold of 5% of liquid assets to warehouse borrowings, reverse repurchase agreements, dividends payable and other short-term liabilities. During 2002 we were in compliance with this policy, which ALCO reports to the board of directors at least quarterly.

We believe that current liquidity levels, available financing facilities and liquidity provided by operating activities will adequately provide for our projected funding needs and asset growth. However, any future margin calls and, depending upon the state of the mortgage industry, terms of any sale of mortgage assets may adversely affect our ability to maintain adequate liquidity levels or may subject us to future losses.

Our operating businesses primarily use available funds as follows:

- acquisition and origination of mortgages;
- provide short-term warehouse advances to affiliates and non-affiliates; and
- pay common stock dividends.

Acquisition and origination of mortgages. During 2002 the mortgage operations acquired \$5.9 billion of primarily Alt-A mortgages. Equity invested in mortgages is outstanding until we sell or securitize mortgages, which is one of the reasons we attempt to sell or securitize mortgages frequently. Initial capital invested in mortgages includes premiums paid when mortgages are acquired and originated and the haircut required upon financing, based on the type of collateral provided. The mortgage operations paid weighted average premiums of 1.66% on the principal balance of mortgages acquired in 2002.

The long-term investment operations acquired and retained \$3.9 billion of Alt-A mortgages from the mortgage operations for long-term investment. In addition, IMCC originated and retained \$25.8 million of multi-family mortgages for long-term investment. When we complete CMOs our required total capital investment ranges from approximately 3% to 5% of the principal balance of mortgages, depending on premiums paid upon acquisition of mortgages, costs paid for completion of CMOs, costs to acquire derivative instruments and initial capital investment in CMOs required to achieve desired credit ratings.

Provide short-term warehouse advances to affiliates and non-affiliates. We utilize uncommitted warehouse facilities with various lenders to provide short-term warehouse financing to affiliates and external customers of the warehouse lending operations.

The warehouse lending operations provides short-term financing to external customers from the closing of the mortgages to their sale or other settlement with investors. The warehouse lending operations finances between 95% and 98% of the fair market value of mortgages, which equates to a haircut requirement of between 3% and 5%, at prime rate plus a spread. At year-end 2002 the warehouse lending operations had \$665.0 million of approved warehouse lines available to non-affiliates, of which \$664.0 million was outstanding.

Affiliates had \$15.2 million in pledge accounts with the warehouse lending operations at year-end 2002, which allows them to finance 100% of the fair market value of their mortgages at prime minus 0.50%. The mortgage operations has uncommitted warehouse line agreements to obtain financing of up to \$600.0 million from the warehouse lending operations to provide interim mortgage financing during the period that the mortgage operations accumulates

mortgages until the mortgages are securitized or sold. At year-end 2002 the warehouse lending operations had \$491.4 million in outstanding warehouse advances to the mortgage operations.

Our ability to meet liquidity requirements and the financing need of our customers is subject to the renewal of our credit and repurchase facilities or obtaining other sources of financing, if required, including additional debt or equity from time to time. Any decision our lenders or investors make to provide available financing to us in the future will depend upon a number of factors, including:

- our compliance with the terms of our existing credit arrangements;
- our financial performance;
- industry and market trends in our various businesses;
- the general availability of and rates applicable to financing and investments;
- our lenders or investors resources and policies concerning loans and investments; and
- the relative attractiveness of alternative investment or lending opportunities.

Pay common stock dividends. We made common stock dividend payments of \$66.3 million in 2002. For additional information regarding dividend distribution requirements refer to Item 5. "Market for Registrant's Common Equity and Related Stockholder Matters" and "Note A — Summary of Business and Significant Accounting Policies" in the accompanying notes to the consolidated financial statements of Impac Mortgage Holdings, Inc.

Our operating businesses are primarily funded as follows:

- CMO borrowings and reverse repurchase agreements;
- cash flows from our mortgage loan investment portfolio;
- sale and securitization of mortgages; and
- cash proceeds from the issuance of securities.

CMO borrowings and reverse repurchase agreements. We use reverse repurchase agreements and CMO borrowings to fund substantially all of our warehouse advances to affiliates and non-affiliates, the acquisition of mortgages to be held for long-term investment and, prior to 1999, the acquisition of investment securities. Since 1999 we have acquired no investment securities.

As we accumulate mortgages for long-term investment, we finance the acquisition of mortgages primarily through borrowings on reverse repurchase agreements with third party lenders. Since 1995 we have primarily used an uncommitted repurchase facility with a major investment bank to finance substantially all warehouse advances to affiliates and non-affiliates and mortgages acquired for long-term investment, as needed. However, during 2002 we added \$650.0 million of new uncommitted warehouse facilities with other lenders to finance asset growth. The new warehouse facilities provide us with a higher aggregate credit limit to fund the acquisition and origination of mortgages at terms comparable to those we have received in the past and the flexibility of having financial relationships with a larger cross-section of financial institutions. At year-end 2002 the warehouse lending operations had \$1.2 billion outstanding on warehouse facilities with various lenders.

We expect to continue to use short-term warehouse facilities to fund the acquisition of mortgages. If we cannot renew or replace maturing borrowings, we may have to sell, on a whole loan basis, the mortgages securing these facilities, which, depending upon market conditions, may result in substantial losses. Additionally, if for any reason the market value of our mortgages securing warehouse facilities decline, our lenders may require us to provide them with additional equity or collateral to secure our borrowings, which may require us to sell mortgages at substantial losses.

In order to mitigate the liquidity risk associated with reverse repurchase agreements, we attempt to securitize our mortgages more frequently. Although securitizing mortgages more frequently adds operating and securitization costs, we believe the added cost is offset as more liquidity is provided with less interest rate and price volatility, as the accumulation and holding period of mortgages is shortened. When we have accumulated a sufficient amount of mortgages, we issue CMOs and convert short-term advances under reverse repurchase agreements to long-term CMO borrowings. The use of CMOs provides the following benefits:

- allows us to lock in our financing cost over the life of the mortgages securing the CMO borrowings; and
- eliminates margin calls on the borrowings that are converted from reverse repurchase agreements to CMO financing.

For 2002 we completed \$3.9 billion of CMOs, of which \$3.3 billion was adjustable rate CMOs and \$597.0 million was fixed rate CMOs, to provide longterm financing for the acquisition of \$3.9 billion of primarily Alt-A mortgages from the mortgage operations. Because of the credit profile, historical loss performance and prepayment characteristics of our Alt-A mortgages, we have been able to borrow a higher percentage against mortgages held as CMO collateral, which means that we have to provide less initial capital upon completion of CMOs. Equity in the CMOs is established at the time CMOs are issued at levels sufficient to achieve desired credit ratings on the securities from credit rating agencies. Total credit loss exposure is limited to the capital invested in the CMOs at any point in time. At year-end 2002 total equity invested in mortgages held as CMO collateral was \$107.9 million, which includes premiums paid when mortgages are acquired from the mortgage operations, costs incurred to complete CMO borrowings, fair value of derivative instruments and the required initial capital investment in CMOs. We also determine the amount of equity invested in CMOs based upon the anticipated return on equity as compared to estimated proceeds from additional debt issuance. By decreasing the amount of capital we are required to invest in our CMOs to maintain desired credit ratings, we have been able to effectively utilize available cash to acquire additional mortgage assets.

For additional information regarding reverse repurchase agreements and CMO borrowings refer to Item 1. "— Financing" and "Note G — Reverse Repurchase Agreements" and "Note H — CMO Borrowings" in the accompanying notes to the consolidated financial statements for Impac Mortgage Holdings, Inc.

Cash flows from our mortgage loan investment portfolio. For 2002 mortgages held as CMO collateral, which is the majority of the mortgage loan investment portfolio, generated excess principal and interest cash flows of \$112.6 million compared to \$44.0 million for 2001. We receive excess principal and interest cash flows on mortgages held as CMO collateral after distributions are made to investors in CMOs to the extent cash or other collateral required to maintain desired credit ratings on the CMOs is fulfilled. Excess principal and interest cash flows represent the difference between principal and interest payments on the mortgages less the following:

- interest paid to bondholders;
- pro-rata early principal prepayments paid to bondholders;
- servicing fees paid to mortgage servicers;
- premiums paid to mortgage insurers; and
- actual losses incurred on disposition of real estate acquired in settlement of mortgages.

Sale and securitization of mortgages. When the mortgage operations accumulates a sufficient amount of mortgages, generally between \$100.0 million and \$300.0 million, it sells or securitizes the mortgages. The mortgage operations sold \$3.9 billion of mortgages to the long-term investment operations during 2002, \$1.1 billion of mortgages to third party investors and \$600.0 million was securitized as REMICs. The mortgage operations sold mortgage servicing rights on substantially all mortgages in 2002. The sale of mortgage servicing rights generated 100% cash gains, which was used to acquire and originate additional mortgages. In order to mitigate interest rate and market risk, the mortgage operations attempts to sell and securitize mortgages frequently. Since we rely significantly upon sales and securitizations to generate cash proceeds to repay borrowings and to create credit availability, any disruption in our ability to complete sales and securitizations may require us to utilize other sources of financing, which, if available at

all, may be on unfavorable terms. In addition, delays in closing sales and securitizations of our mortgages increase our risk by exposing us to credit and interest rate risk for this extended period of time.

Cash proceeds from the issuance of securities. In December 2001, we filed with the SEC a shelf registration statement that allows us to sell up to \$300.0 million of securities, including common stock, preferred stock, debt securities and warrants. We raised \$116.0 million in net proceeds from the issuance of 13.3 million new shares of common stock in 2002. By issuing new shares periodically throughout the year we believe that we were able to utilize new capital more efficiently and profitably.

Cash Flows

Operating Activities - Net cash provided by operating activities was \$101.5 million for 2002 compared to \$55.5 million for 2001. Net earnings of \$74.9 million provided most of the cash flows from operating activities.

Investing Activities - Net cash used in investing activities was \$3.7 billion for 2002 compared to \$961.0 million for 2001. Net cash flows of \$3.0 billion, including principal payments, was used in investing activities to acquire and originate mortgages.

Financing Activities - Net cash provided by financing activities was \$3.6 billion for 2002 compared to \$939.5 million for 2001. Net cash flows of \$3.9 billion provided by CMO financing was primarily provided by financing activities.

Inflation

The consolidated financial statements and corresponding notes to the consolidated financial statements have been prepared in accordance with GAAP, which require the measurement of financial position and operating results in terms of historical dollars without considering the changes in the relative purchasing power of money over time due to inflation. The impact of inflation is reflected in the increased costs of our operations. Unlike industrial companies, nearly all of our assets and liabilities are monetary in nature. As a result, interest rates have a greater impact on our performance than do the effects of general levels of inflation. Inflation affects our operations primarily through its effect on interest rates, since interest rates normally increase during periods of high inflation and decrease during periods of low inflation. During periods of increasing interest rates, demand for mortgages and a borrower's ability to qualify for mortgage financing in a purchase transaction may be adversely affected. During periods of decreasing interest rates, borrowers may prepay their mortgages, which in turn may adversely affect our yield and subsequently the value of our portfolio of mortgage assets.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

General

We focus on effectively managing the various operational and market risks associated with our businesses. We believe that the most critical of those risks are:

- credit risk;
- prepayment risk;
- liquidity risk; and
- interest rate risk;

We manage credit risk by acquiring high-credit quality Alt-A mortgages from the mortgage operations with favorable credit profiles and by acquiring mortgages with conservative loan-to-value ratios with mortgage insurance enhancements, when required, which reduces our effective loan-to-value ratio. Our belief is that high-credit quality Alt-A mortgages will result in favorable foreclosure rates and conservative loan-to-value ratios will result in favorable loss rates. We also believe that we maintain an adequate allowance for loan losses to provide for future loan losses.

We manage mortgage prepayment risk by acquiring Alt-A mortgages from the mortgage operations with prepayment penalty features.

We manage liquidity risk by frequently securitizing or selling our mortgages. We securitize mortgages through the issuance of CMOs and REMICs every 30 to 45 days. By frequently securitizing our mortgages, we reduce the volume of mortgages that are financed with short-term reverse repurchase agreements at any given time. The issuance of CMOs convert short-term reverse repurchase borrowings, which are subject to margin calls if the value of our mortgages collateralizing reverse repurchase borrowings decline, to long-term CMO financing that are not subject to margin calls. By securitizing mortgages as REMICs or selling mortgages as whole loan sales, ownership of the mortgages transfers to the trust in the case of REMICs and to the buyer in the case of whole loan sales. For additional information regarding these risks refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations."

Although we manage credit, prepayment and liquidity risk in the normal course of business, we consider interest rate risk to be a significant market risk, which could potentially have the largest material effect on our financial condition and results of operations. Since a significant portion of our revenues and earnings are derived from net interest income, we strive to manage our interest-earning assets and interest-bearing liabilities to generate what we believe to be an appropriate contribution from net interest income. When interest rates fluctuate, profitability can be adversely affected by changes in the fair market value of our assets and liabilities and by the interest spread earned on interest-earning assets and interest-bearing liabilities. We derive income from the differential spread between interest earned on interest-earning assets and interest-bearing liabilities. Any change in interest rates affects income received and income paid from assets and liabilities in varying and typically in unequal amounts. Changing interest rates may compress our interest rate margins and adversely affect overall earnings.

Interest rate risk management is the responsibility of ALCO, which reports results of interest rate risk analysis to the board of directors on a quarterly basis. ALCO establishes policies that monitor and coordinate sources, uses and pricing of funds. ALCO also attempts to reduce the volatility in net interest income by managing the relationship of interest rate sensitive assets to interest rate sensitive liabilities. In addition, various modeling techniques are used to value interest sensitive mortgage-backed securities, including interest-only securities. The value of mortgage-backed securities is determined using a discounted cash flow model using prepayment rate, discount rate and credit loss assumptions. Our investment securities portfolio is available-for-sale, which requires us to perform market valuations of the securities in order to properly record the portfolio. We continually monitor interest rates of our investment securities portfolio as compared to prevalent interest rates in the market. We do not currently maintain a securities trading portfolio and are not exposed to market risk as it relates to trading activities.

Interest Rate Sensitive Assets and Liabilities

The matching of assets and liabilities may be analyzed by examining the extent to which such assets and liabilities are "interest rate sensitive" and by monitoring our interest rate sensitivity "gap." An asset or liability is said to be interest rate sensitive within a specific time period if it will mature or re-price within that time period. The interest rate sensitivity gap is defined as the difference between the amount of interest-earning assets maturing or re-pricing within a specific time period and the amount of interest-bearing liabilities maturing or re-pricing within that time period. A gap is considered positive when the amount of interest rate sensitive liabilities. A gap is considered negative when the amount of interest rate sensitive liabilities exceeds the amount of interest rate sensitive assets.

Therefore, during a period of declining interest rates, the net earnings of an institution with a positive gap, theoretically, may be adversely affected due to its interest-earning assets re-pricing to a greater extent than its interest-bearing liabilities. Conversely, during a period of rising interest rates, theoretically, the net earnings of an institution with a positive gap position may rise as it is able to invest in higher yielding interest-earning assets at a more rapid pace than the re-pricing of its interest-bearing liabilities. Gap analysis does not take into consideration the constraints on the re-pricing of interest rates of ARM assets in a given period resulting from periodic and lifetime cap features, the behavior of various indices applicable to assets and liabilities or the effects of derivative instruments on net interest income. For additional information regarding the affect of changing interest rates on interest rate sensitive assets and liabilities refer to "Changes in Interest Rates" below.

The amount of assets and liabilities shown which re-price or mature during a particular period were determined in accordance with the earlier of (1) term to re-pricing or (2) the contractual term of the asset or liability adjusted for scheduled and unscheduled principal prepayments. Unscheduled prepayment rates are assumed on substantially all of our mortgage loan investment and mortgage-backed securities investment portfolios and are based on historic loan prepayment experience and anticipated future prepayments. The table does not include assets and liabilities that are not interest rate sensitive, such as interest receivables and payables, prepaid expenses and accrued expenses.

The following table summarizes the amount of interest-earning assets and interest-bearing liabilities outstanding as of December 31, 2002 (dollar amounts in thousands), which are anticipated to re-price or mature in each of the future time periods indicated:

	 2003	2004	2005	2006	2007	Over 5 Years	Total
Interest sensitive assets:							
Cash equivalents	\$ 113,345 \$	— \$	— \$	— \$	— \$	— \$	113,345
Average interest rate	2.00%	— %	— %	— %	— %	— %	2.00 %
Investment securities (1)	2,967	3,345	4,234	5,044	3,571	6,904	26,065
Average interest rate	17.93%	11.49%	2.98%	1.92%	1.85%	1.83%	5.11%
Finance receivables	1,140,248	_	_	_	_	_	1,140,248
Average interest rate	3.90%	— %	— %	— %	— %	— %	3.90%
CMO collateral (2)	4,012,599	559,677	133,812	78,907	59,750	304,936	5,149,681
Average interest rate	5.94%	8.75%	7.78%	7.48%	8.45%	7.20%	6.45%
Loans held-for-investment (3)	6,010	3⁄4	3⁄4	3⁄4	3⁄4	51,526	57,536
Average interest rate	— %	— %	— %	— %	— %	7.12%	6.37%
Due from affiliates	14,500	3⁄4	_	_	3⁄4	3⁄4	14,500
Average interest rate	9.50%	— %	— %	— %	— %	— %	9.50 %
Total interest-sensitive assets	\$ 5,289,669 \$	563,022 \$	138,046 \$	83,951 \$	63,321 \$	363,366 \$	6,501,375
Average interest rate	5.42%	8.76%	7.63%	7.15%	8.08%	7.09%	5.93%
Interest sensitive liabilities:							
CMO borrowings	\$ 5,041,751 \$	— \$	— \$	— \$	— \$	— \$	5,041,751
Average interest rate	2.12%	— %	— %	— %	— %	— %	2.12%
Reverse repurchase agreements	1,168,028	_	3⁄4	_	_	_	1,168,028
Average interest rate	3.44%	— %	— %	— %	— %	— %	3.44%
Borrowings secured by Securities available-for-sale	_	7,134	_	_	_	_	7,134
Average interest rate	— %	24.01%	— %	— %	— %	— %	24.01%
5	 						
Total interest-sensitive liabilities	\$ 6,209,779 \$	7,134 \$	— \$	— \$	— \$	— \$	6,216,913
Average interest rate	2.35%	24.01%	— %	— %	— %	— %	2.37%
Interest rate gap (1)	\$ (920,110) \$	555,888 \$	138,046 \$	83,951 \$	63,321 \$	363,366 \$	284,462
Cumulative interest rate gap	\$ (920,110) \$	(364,222) \$	(226,176) \$	(142,225) \$	(78,904) \$	284,462	

(1) Interest rate gap represents the difference between interest-earning assets and interest-bearing liabilities.

As of December 31, 2002 the difference between interest-earning assets and interest-bearing liabilities resulted in a negative estimated interest rate sensitivity gap for 2003. Under this scenario, net interest income could be positively affected by a decrease in interest rates as more interest-bearing liabilities could re-price downwards during 2003 than interest-bearing assets. Conversely, an increase in interest rates could have a negative affect on net interest income during 2003 as more interest-bearing liabilities could re-price upwards than interest-bearing assets. The estimated cumulative negative gap of \$917.6 million during the first year using data as of December 31, 2002 compares to an estimated cumulative negative gap of \$1.0 billion during the first year using data as of December 31, 2002 compares to an estimated cumulative negative gap of \$1.0 billion during the first year using data as of December 31, 2002 compares to an estimated cumulative negative gap of \$1.0 billion during the first year using data as of December 31, 2001. Since gap estimates are based upon numerous assumptions, such as mortgage prepayment rates and the expected duration of interest-earning assets and interest-bearing liabilities, actual sensitivity to interest rate changes could vary significantly if actual experience differs from those assumptions used in making the calculations.

Changes in Interest Rates

ALCO follows an interest rate hedge program intended to limit our exposure to changes in interest rates primarily associated with cash flows on our adjustable rate CMO borrowings. Our primary objective is to hedge our exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of our adjustable rate CMO borrowings. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate hedge program is formulated with the intent to offset the potential adverse effects of changing interest rates on cash flows on adjustable rate CMO borrowings:

- interest rate adjustment limitations on mortgages held as CMO collateral due to periodic and lifetime interest rate cap features; and
- mismatched interest rate adjustment periods between mortgages held as CMO collateral and CMO borrowings.

We primarily acquire for long-term investment six-month LIBOR ARMs and hybrid ARMs. Six-month LIBOR ARMs are generally subject to periodic and lifetime interest rate caps. This means that the interest rate of each ARM is limited to upwards or downwards movements on its periodic interest rate adjustment date, generally six months, or over the life of the mortgage. Periodic caps limit the maximum interest rate change, which can occur on any interest rate change date to generally a maximum of 1% per semiannual adjustment. Also, each ARM has a maximum lifetime interest rate cap. Generally, borrowings are not subject to the same periodic or lifetime interest rate limitations. During a period of rapidly increasing or decreasing interest rates, financing costs would increase or decrease at a faster rate than the periodic interest rate adjustments on mortgages would allow, which could affect net interest income. In addition, if market rates were to exceed the maximum interest rates of our ARMs, borrowing costs would increase while interest rates on ARMs would remain constant.

We also acquire hybrid ARMs that have initial fixed interest rate periods generally ranging from two to three years, and to a lesser extent five years, which subsequently convert to six-month LIBOR ARMs. During a rapidly increasing or decreasing interest rate environment financing costs would increase or decrease more rapidly than would interest rates on mortgages, which would remain fixed until their next interest rate adjustment date. In order to provide some protection against any resulting basis risk shortfall on the related liabilities we purchase derivative instruments. Derivative instruments are based upon the principal balance that would result under assumed prepayment speeds.

Although the static gap methodology is widely accepted in identifying interest rate risk, it does not take into consideration changes that may occur in investment and financing strategies, changes in the forward yield curve, changes in hedging strategy, changes in mortgage prepayment speeds and changes in volumes of mortgage acquisitions. Therefore, in addition to measuring interest rate risk via a gap analysis, we measure the sensitivity of our net interest income to changes in interest rates affecting interest sensitive assets and liabilities using simulations.

As part of various interest rate simulations, we calculate the effect of potential changes in interest rates on our interest-earning assets and interest-bearing liabilities and their affect on overall earnings. The simulations assume instantaneous and parallel shifts in interest rates and to what degree those shifts affect net interest income. First, we estimate our net interest income for the next twelve months using balance sheet data as of December 31, 2002 and 12-month projections of the following:

- future interest rates using forward yield curves, which are market consensus estimates of future interest rates;
- acquisition of derivative instruments;
- mortgage prepayment rate assumptions; and
- mortgage acquisitions.

We refer to this 12-month projection of net interest income as the "base case." Once the base case has been established, we "shock" the base case with instantaneous and parallel shifts in interest rates in 100 basis point increments upward and downward. Calculations are made for each of the defined instantaneous and parallel shifts in interest rates over or under the forward yield curve used to determine the base case and including any associated changes in projected mortgage prepayment rates caused by changes in interest rates. The results of each 100 basis point change in interest rates are then compared against the base case to determine the estimated change to net interest

income. The simulations consider the affect of interest rate changes on interest sensitive assets and liabilities as well as derivative instruments. The simulations also consider the impact that instantaneous and parallel shift in interest rates has on prepayment rates and the resulting affect of accelerating or decelerating amortization rates of premium and securitization costs on net interest income.

The following tables (dollar amounts in millions) estimates the financial impact to net interest income from the various instantaneous and parallel shifts in interest rates based on our balance sheet structure as of December 31, 2002 and 2001. Since these estimates are based upon numerous assumptions, actual sensitivity to interest rate changes could vary if actual experience differs from the assumptions used.

	Changes in	base case net interest inc	ome as of December 31, 2002	2 (2)
	Without deriv instrumen		With deriva instrumen	
Instantaneous and Parallel Change in Interest Rates (1)	(\$)	(%)	(\$)	(%)
Up 200 basis points, or 2%	(46.1)	(26)	(7.9)	(6)
Up 100 basis points, or 1%	(24.3)	(14)	(4.9)	(4)
Down 100 basis points, or 1%	27.9	16	8.1	6
Down 200 basis points, or 2%	32.0	18	10.0	8

Changes in base case net interest income as of December 31, 2001 (2)

	Without deri instrumer		With derivative instruments			
Instantaneous and Parallel Change in Interest Rates (1)	(\$)	(%)	(\$)	(%)		
Up 200 basis points, or 2%	(30.8)	(31)	(5.6)	(7)		
Up 100 basis points, or 1%	(14.5)	(15)	(2.4)	(3)		
Down 100 basis points, or 1%	9.9	10	(3.5)	(5)		
Down 200 basis points, or 2%	21.6	22	(5.3)	(7)		

(1) Instantaneous and parallel interest rate changes over and under the projected forward yield curve.

(2) The dollar and percentage changes represent base case net interest income for the next twelve months versus the change in net interest income in the various instantaneous and parallel interest rate change simulations.

The use of derivative instruments to hedge changes in interest rates is an integral part of our strategy to limit interest rate risk. Therefore, net interest income may be significantly impacted by cash payments we are required to make or cash payments we receive on derivative instruments. The amount of cash payments or cash receipts on derivative instruments is determined by (1) the notional amount of the derivative instrument and (2) current interest rate levels in relation to the various strike prices of derivative instruments during a particular time period.

The following table summarizes the notional amount (in thousands), weighted average strike price and low and high strike prices of interest rate caps, interest rate floors and interest rate swaps that were acquired to hedge the variability of interest rates and corresponding cash flows on CMO borrowings for the interest due dates indicated:

	Interest Rate Caps (1) Interest Rate Floors							Interest Rate Swaps				
Interest Due Date		Notional Balance	Weighted Average Strike	Low Strike	Notional Balance	Weighted Average Strike	Low Strike	High Strike	Notional Balance	Weighted Average Fixed Rate	Low Fixed Rate	High Fixed Rate
Jan 2003	\$	3,013,300	4.16%	1.90%\$	1,624,596	3.63%	2.62%	4.68%\$	935,503	2.65%	2.05%	5.18%
Feb 2003		2,929,466	4.08	1.79	1,585,277	3.56	2.66	4.50	941,384	2.64	2.05	5.18
Mar 2003		2,825,653	4.10	1.84	1,547,650	3.60	1.84	5.23	936,524	2.64	2.05	5.18
Apr 2003		2,720,175	4.34	2.03	1,502,007	3.83	1.84	5.44	930,935	2.64	2.05	5.18
May 2003		2,596,248	4.33	2.03	1,457,946	3.83	1.84	5.44	924,618	2.64	2.05	5.18
June 2003		2,484,014	4.37	2.12	1,414,711	3.83	3.32	4.80	887,907	2.56	2.05	5.18
July 2003		2,376,644	4.60	2.39	1,330,292	3.97	1.84	5.66	861,991	2.51	2.05	5.18
Aug 2003		2,239,429	4.56	2.39	1,211,940	3.93	2.09	5.40	854,410	2.51	2.05	5.18
Sep 2003		2,112,970	4.61	2.54	1,102,832	3.94	2.34	5.03	846,117	2.51	2.05	5.18
Oct 2003		2,022,564	4.81	2.88	1,036,466	4.11	2.59	5.55	837,126	2.51	2.05	5.18
Nov 2003		1,771,388	4.89	2.88	911,333	4.16	4.07	5.18	827,455	2.51	2.05	5.18
Dec 2003		1,616,238	4.99	3.01	825,602	4.22	3.09	5.59	817,118	2.51	2.05	5.18
Jan 2004		1,422,535	5.23	3.42	743,881	4.37	4.47	5.27	806,135	2.51	2.05	5.18
Feb 2004		1,290,226	5.07	3.31	702,961	4.36	4.33	5.30	794,527	2.51	2.05	5.18
Mar 2004		1,051,052	5.11	3.43	623,460	4.36	3.59	5.67	782,316	2.51	2.05	5.18
Apr 2004		717,354	5.18	3.74	355,711	4.23	3.59	5.75	769,528	2.51	2.05	5.18
May 2004		590,377	5.10	3.74	310,753	4.23	4.54	5.43	756,186	2.51	2.05	5.18
June 2004		509,406	5.12	3.83	242,025	4.40	3.59	5.77	738,558	2.50	2.05	3.70
July 2004		288,158	5.32	4.05	151,673	4.66	3.59	5.82	724,248	2.50	2.05	3.70
Aug 2004		165,490	5.71	4.05	87,774	4.63	3.59	5.55	709,469	2.50	2.05	3.70
Sep 2004		118,649	6.23	4.68	79,769	4.59	3.59	4.91	536,647	2.46	2.05	3.70
Oct 2004		113,771	6.29	4.80	75,640	4.62	3.59	5.51	504,068	2.46	2.05	3.70
Nov 2004		102,987	6.36	4.80	65,592	4.60	3.59	5.51	300,860	2.50	2.05	3.70
Dec 2004		77,605	6.42	4.87	59,003	4.61	3.59	5.55	182,480	2.70	2.10	3.70
Average Notional Balance of Derivatives Weighted	\$	16,015		\$	15,261			\$	9,873			
Average Strike			4.62%			3.95%				2.60%		

(1) The high strike price for each period indicated is 10.25%.

The strike prices of interest rate caps and interest rate floors are based on one-month LIBOR as that is the interest rate index that is used to determine interest rates on adjustable rate CMO borrowings. As of December 31, 2002 one-month LIBOR was 1.38%. Since one-month LIBOR was below the majority of strike prices on interest rate caps during 2002, we did not receive significant cash from these derivative instruments in 2002. Conversely, we made cash payments on many interest rate floors strike prices. We made cash payments on many interest rate floors as we hold short positions on the majority of interest rate floors. We originally sold interest rate floors to offset our cost of acquiring long positions in interest rate caps. If one-month LIBOR rises and approaches the floor strike prices, we will pay less on the interest rate floors.

We primarily acquired interest rate caps and floors during the first half of 2002 and throughout 2001 to protect net interest income during a potentially rising interest rate environment. With market volatility and uncertainty and the associated risk involved which was occurring during 2001 and that was further exacerbated by the terrorist events of September 11, 2001, we felt it was prudent, at the time, to protect net interest margins from possible extreme interest rate increases by acquiring interest rate caps and floors. However, because interest rates continued to decline to all-time lows during 2002, we began to acquire interest rate swaps to essentially convert our adjustable rate CMO borrowings with fixed rate derivative instruments. For instance, we receive one-month LIBOR on interest rate swaps, which offsets interest expense on adjustable rate CMO borrowings, and we pay a fixed interest rate. We continued to acquire interest rate swaps to manage interest rate risk during the first quarter of 2003.

The following table presents the extent to which changes in interest rates and changes in the volume of interest rate sensitive assets and interest rate sensitive liabilities have affected interest income and interest expense during the

periods indicated. Information is provided on Mortgage Assets and borrowings on Mortgage Assets, only, with respect to the following:

- changes attributable to changes in volume (changes in volume multiplied by prior rate);
- changes attributable to changes in rate (changes in rate multiplied by prior volume);
- changes in interest due to both rate and volume; and
- the net change.

	Year Ended December 31, 2002 over 2001							
		Volume	Rate	Rate/ Volume	Net Change			
			(in thousand	ls)				
Increase/(decrease) in:								
Subordinated securities collateralized by mortgages	\$	(542) \$	(1,700) \$	262 \$	(1,980)			
Loans receivable:								
CMO collateral		133,157	(28,753)	(35,343)	69,061			
Mortgages held-for-investment		(1,350)	(1,452)	239	(2,563)			
Finance receivables:								
Affiliated		9,312	(6,389)	(3,241)	(318)			
Non-affiliated		10,301	(3,895)	(2,579)	3,827			
			<u></u>	·				
Total finance receivables		19,613	(10,284)	(5,820)	3,509			
Total loans receivable		151,420	(40,489)	(40,924)	70,007			
Change in interest income on Mortgage Assets		150,878	(42,189)	(40,662)	68,027			
					<u> </u>			
CMO borrowings		100,171	(26,331)	(33,897)	39,943			
Reverse repurchase agreements		12,396	(13,987)	(5,629)	(7,220)			
Borrowings secured by investment securities		(1,069)	608	(253)	(714)			
		·	<u> </u>		<u> </u>			
Change in interest expense on borrowings on Mortgage Assets		111,498	(39,710)	(39,779)	32,009			
	<u></u>		(0. 170) *	(000) +	20.010			
Change in net interest income	\$	39,380 \$	(2,479) \$	(883) \$	36,018			

	Y	Zear Ended December 31;	, 2001 over 2000	
	 Volume	Rate	Rate/ Volume	Net Change
		(in thousand	ls)	
Increase/(decrease) in:				
Subordinated securities collateralized by mortgages	\$ (2,949) \$	(1,320) \$	537 \$	(3,732)
Subordinated securities collateralized by other loans	(273)	(273)	273	(273)
Total investment securities	 (3,222)	(1,593)	810	(4,005)
Loans receivable:	 			
CMO collateral	23,030	(493)	(132)	22,405
Mortgages held-for-investment	1,338	(1,841)	(275)	(778)
Finance receivables:				
Affiliated	(1,516)	(8,633)	467	(9,682)
Non-affiliated	7,496	(4,079)	(2,141)	1,276
Total finance receivables	 5,980	(12,712)	(1,674)	(8,406)
Total loans receivable	 30,348	(15,046)	(2,081)	13,221
Change in interest income on Mortgage Assets	 27,126	(16,639)	(1,271)	9,216
CMO borrowings	25,096	(21,004)	(6,565)	(2,473)
Reverse repurchase agreements	5,083	(11,946)	(1,548)	(8,411)

Borrowings secured by investment securities	 (1,117)	714	(248)	(651)
Change in interest expense on borrowings on Mortgage Assets	 29,062	(32,236)	(8,361)	(11,535)
Change in net interest income	\$ (1,936) \$	15,597 \$	7,090 \$	20,751

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

The information required by this Item 8 is incorporated by reference to Impac Mortgage Holdings, Inc.'s Consolidated Financial Statements and Independent Auditors' Report beginning at page F-1 of this Form 10-K.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None.

PART III

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT

The information required by this Item 10 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2002 fiscal year.

ITEM 11. EXECUTIVE COMPENSATION

The information required by this Item 11 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2002 fiscal year.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The information required by this Item 12 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2002 fiscal year.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

The information required by this Item 13 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2002 fiscal year.

ITEM 14. CONTROLS AND PROCEDURES

As of December 31, 2002 the Chief Executive Officer, or "CEO," and Chief Financial Officer, or "CFO," performed an evaluation of the effectiveness and the and operation of the Company's disclosure controls and procedures as defined in Rule 13a – 14c under the Securities Exchange Act of 1934, as amended. Based on that evaluation, the CEO and CFO concluded that the Company's disclosure controls and procedures were effective as of December 31, 2002. There have been no significant changes in the Company's internal controls or in other factors that could significantly affect internal controls subsequent to December 31, 2002.

PART IV

ITEM 15. EXHIBITS, FINANCIAL STATEMENT SCHEDULES AND REPORTS ON FORM 8-K

(a) All schedules have been omitted because they are either not applicable, not required or the information required has been disclosed in the Consolidated Financial Statements and related Notes to Consolidated Financial Statements at page F-1, or otherwise included in this Form 10-K.

(b) Reports on Form 8-K

Current Report on Form 8-K/A, dated November 26, 2002, reporting Items 5 and 7, relating to the execution of an Amended and Restated Equity Distribution Agreement with UBS Warburg LLC.

Current Report on Form 8-K, dated November 27, 2002, furnished to the SEC reporting Item 9, relating to the posting of the Company's unaudited monthly fact sheet.(c) Exhibits

Exhibit Number	Description
3.1	Charter of the Registrant (incorporated by reference to the corresponding exhibit number to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7, 1995).
3.1(a)	Certificate of Correction of the Registrant (incorporated by reference to exhibit 3.1(a) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(b)	Articles of Amendment of the Registrant (incorporated by reference to exhibit 3.1(b) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(c)	Articles of Amendment for change of name to Charter of the Registrant (incorporated by reference to exhibit number 3.1(a) of the Registrant's Current Report on Form 8-K, filed February 11, 1998).
3.1(d)	Articles Supplementary and Certificate of Correction for Series A Junior Participating Preferred Stock of the Registrant (incorporated by reference to exhibit 3.1(d) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(e)	Articles Supplementary for Series B 10.5% Cumulative Convertible Preferred Stock of the Registrant (incorporated by reference to exhibit 3.1b of the Registrant's Current Report on Form 8-K, filed December 23, 1998).
3.1(f)	Articles Supplementary for Series C 10.5% Cumulative Convertible Preferred Stock of the Registrant (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending September 30, 2000).
3.1(g)	Certificate of Correction for Series C Preferred Stock of the Registrant (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending September 30, 2000).
3.1(h)	Articles Supplementary, filed with the State Department of Assessments and Taxation of Maryland on July 12, 2002, reclassifying Series C Preferred Stock of the Registrant (incorporated by reference to exhibit 9 of the Registrant's Form 8-A/A, Amendment No. 2, filed July 30, 2002).
3.1(i)	Articles of Amendment, filed with the State Department of Assessments and Taxation of Maryland on July 17, 2002, increasing authorized shares of Common Stock of the Registrant (incorporated by reference to exhibit 10 of the Registrant's Form 8-A/A, Amendment No. 2, filed July 30, 2002).
3.1(j)	Resolution to Change Principal Officer or Resident Agent, filed with the State Department of Assessments and Taxation of Maryland on September 11, 2002

3.2 Bylaws of the Registrant, as amended and restated (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending March 31, 1998).

4.1	Form of Stock Certificate of the Company (incorporated by reference to the corresponding exhibit number to the Registrant's Registration
	Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7, 1995).

- 4.2 Rights Agreement between the Registrant and BankBoston, N.A. (incorporated by reference to exhibit 4.2 of the Registrant's Registration Statement on Form 8-A as filed with the Securities and Exchange Commission on October 14, 1998).
- 4.2(a) Amendment No. 1 to Rights Agreement between the Registrant and BankBoston, N.A. (incorporated by reference to exhibit 4.2(a) of the Registrant's Registration Statement on Form 8-A/A as filed with the Securities and Exchange Commission on December 23, 1998).
- 10.1 1995 Stock Option, Deferred Stock and Restricted Stock Plan, as amended and restated (incorporated by reference to exhibit 10.1 of the Registrant's Quarterly Report on Form 10-Q for the period ending March 31, 1998).
- 10.2 Form of Indemnity Agreement between the Registrant and its Directors and Officers (incorporated by reference to exhibit 10.4 to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7, 1995).
- 10.3 Form of Amended and Restated Employment Agreement with ICI Funding Corporation (incorporated by reference to exhibit 10.8 to the Registrant's Quarterly Report on Form 10-Q, as amended, for the quarter ended June 30, 1998).
- 10.3(a) List of Officers and terms relating to Form of Amended and Restated Employment Agreement (incorporated by reference to exhibit 10.8(a) to the Registrant's Quarterly Report on Form 10-Q, as amended, for the quarter ended June 30, 1998).
- 10.3(b) Form of Amendment No. 1 to Amended and Restated Employment Agreement with Impac Funding Corporation (incorporated by reference to exhibit 10.1(a) of the Registrant's Current Report on Form 8-K, filed June 2, 1998).
- 10.3(c) List of Officers and terms relating to Form of Amendment No. 1 to the Amended and Restated Employment Agreement with Impac Funding Corporation (incorporated by reference to exhibit 10.1(b) of the Registrant's Current Report of Form 8-K, filed June 2, 1998).
- 10.4 Form of Loan Purchase and Administrative Services Agreement between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.9 to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7,1995).
- 10.5 Servicing Agreement effective November 11, 1995 between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.14 to the Registrant's Registration Statement on Form S-11, as amended (File No. 333-04011), filed with the Securities and Exchange Commission on May 17, 1996).
- 10.6 Impac Mortgage Holdings, Inc. 1996 Stock Option Loan Plan (incorporated by reference to exhibit 10.15 to the Registrant's Form 10-K for the year ended December 31, 1996).
- 10.7 Lease dated June 1, 1998 regarding 1401 Dove Street, Newport Beach California (incorporated by reference to exhibit 10.17 of the Registrant's 10-K for the year ended December 31, 1998).
- 10.7(a) Second Amendment to Lease dated October 1, 1999 between The Realty Associates Fund V, L.P., the Registrant and Impac Funding Corporation regarding 1401 Dove Street, Newport Beach California (incorporated by reference to exhibit number 10.4(d) of the Registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2000).

- 10.8 Office Lease, First Amendment to Office Lease, and Assignment, Assumption and Consent to Assignment of Lease with Property California OB One Corporation and Assignment to Impac Funding Corporation regarding 15050 Avenue of Science Suite 210 San Diego California. (incorporated by reference to exhibit number 10.10 of the Registrant's Annual Report on Form 10-k for the year ended December 31, 2001).
- 10.9 Employment Letter between Impac Funding Corporation and Ronald Morrison dated May 28, 1998 (incorporated by reference to exhibit 10.18 of the Registrant's 10-K for the year ended December 31, 1998).
- 10.10 Note dated June 30, 1999 between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.17 of the Registrant's 10-K for the year ended December 31, 1999).
- 10.11 Underwriting Agreement dated October 22, 2001 among the Registrant, Impac Funding Corporation and UBS Warburg LLC (incorporated by reference to exhibit 1.1 of Registrant's Current Report on Form 8-K dated October 22, 2001).
- 10.12 Underwriting Agreement dated February 7, 2002 among the Registrant, HBK MasterFund L.P. and UBS Warburg LLC (incorporated by reference to exhibit 1.1 of Registrant's Current Report on Form 8-K dated February 8, 2002).
- 10.13 Impac Mortgage Holdings, Inc. 2001 Stock Option Plan, Deferred Stock and Restricted Stock Plan (incorporated by reference to Appendix A of Registrant's Definitive Proxy Statement filed with the SEC on April 30, 2001).
- 10.14 Amendment to Impac Mortgage Holdings, Inc. 2001 Stock Option Plan, Deferred Stock and Restricted Stock Plan (incorporated by reference to Exhibit 4.1(a) of the Registrants Form S-8 filed with the SEC on March 1, 2002).
- 10.15 Underwriting Agreement dated August 22, 2002 among the Registrant, JMP Securities LLC and Sandler O'Neill & Partners, L.P. (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed August 23, 2002).
- 10.16 Underwriting Agreement dated February 20, 2003 among the Registrant, UBS Warburg LLC, JMP Securities LLC, RBC Dain Rauscher Inc., and Wedbush Morgan Securities, Inc. (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed February 21, 2003).
- 10.17 The Impac Companies Deferred Compensation Plan.
- 21.1 Subsidiaries of the Registrant (incorporated by reference to exhibit 21.1 of the Registrant's Quarterly Report on Form 10-Q for the period ended September 30, 2002).
- 23.1 Consent of KPMG LLP regarding the Registrant.
- 23.2 Consent of KPMG LLP regarding Impac Funding Corporation.
- 24 Power of Attorney (included on signature page).
- 99.1 Certifications of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Newport Beach, State of California, on the 28th day of March 2003.

IMPAC MORTGAGE HOLDINGS, INC.

by /s/ JOSEPH R. TOMKINSON

Joseph R. Tomkinson Chairman of the Board and Chief Executive Office

We, the undersigned directors and officers of Impac Mortgage Holdings, Inc., do hereby constitute and appoint Joseph R. Tomkinson and Richard J. Johnson, or either of them, our true and lawful attorneys and agents, to do any and all acts and things in our name and behalf in our capacities as directors and officers and to execute any and all instruments for us and in our names in the capacities indicated below, which said attorneys and agents, or either of them, may deem necessary or advisable to enable said corporation to comply with the Securities Exchange Act of 1934, as amended, and any rules, regulations, and requirements of the Securities and Exchange Commission, in connections with this report, including specifically, but without limitation, power and authority to sign for us or any of us in our names and in the capacities indicated below, any and all amendments to this report, and we do hereby ratify and confirm all that the said attorneys and agents, or either of them, shall do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1934, this report has been signed by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Joseph R. Tomkinson	Chairman of the Board, Chief Executive Officer and Director	March 28, 2003
Joseph R. Tomkinson		
/s/ Richard J. Johnson	Chief Financial Officer and Executive Vice President	March 28, 2003
Richard J. Johnson	vice riesident	
/s/ William S. Ashmore	President and Director	March 28, 2003
William S. Ashmore		
/s/ JAMES WALSH	Director	March 28, 2003
James Walsh		
/s/ Frank P. Filipps	Director	March 28, 2003
Frank P. Filipps		
/s/ Stephan R. Peers	Director	March 28, 2003
Stephan R. Peers		
/s/ William E. Rose	Director	March 28, 2003
William E. Rose		
/s/ Leigh J. Abrams	Director	March 28, 2003
Leigh J. Abrams	-	

CERTIFICATION

I, Joseph R. Tomkinson, certify that:

- 1. I have reviewed this yearly report on Form 10-K of Impac Mortgage Holdings, Inc.;
- 2. Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this annual report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the registrant and we have:
 - a. designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - b. evaluated the effectiveness of the registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this yearly report (the "Evaluation Date"); and
 - c. presented in this annual report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies in the design or operation of internal controls which could adversely affect the registrant's ability to record, process, summarize and report financial data and have identified for the registrant's auditors any material weaknesses in internal controls; and

b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls; and

6. The registrant's other certifying officers and I have indicated in this annual report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

/s/ Joseph R. Tomkinson

Joseph R. Tomkinson Chief Executive Officer

March 28, 2003

CERTIFICATION

I, Richard J. Johnson, certify that:

- 1. I have reviewed this yearly report on Form 10-K of Impac Mortgage Holdings, Inc.;
- 2. Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this annual report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the registrant and we have:
 - a. designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - b. evaluated the effectiveness of the registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this annual report (the "Evaluation Date"); and
 - c. presented in this annual report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies in the design or operation of internal controls which could adversely affect the registrant's ability to record, process, summarize and report financial data and have identified for the registrant's auditors any material weaknesses in internal controls; and

b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls; and

6. The registrant's other certifying officers and I have indicated in this annual report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

/s/ RICHARD J. JOHNSON

Richard J. Johnson Chief Financial Officer

March 28, 2003

INDEPENDENT AUDITORS' REPORT AND CONSOLIDATED FINANCIAL STATEMENTS

INDEX

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES

Independent Auditors' Report	F-2
Consolidated Balance Sheets at December 31, 2002 and 2001	F-3
Consolidated Statements of Operations and Comprehensive Earnings (Loss) for the three years ended December 31, 2002, 2001 and 2000	F-4
Consolidated Statements of Changes in Stockholders' Equity for the three years ended December 31, 2002, 2001 and 2000	F-6
Consolidated Statements of Cash Flows for the three years ended December 31, 2002, 2001 and 2000	F-7
Notes to Consolidated Financial Statements	F-9
IMPAC FUNDING CORPORATION AND SUBSIDIARIES	
Independent Auditors' Report	F-39
Consolidated Balance Sheets at December 31, 2002 and 2001	F-40
Consolidated Statements of Operations and Comprehensive Earnings (Loss) for the three years ended December 31, 2002, 2001 and 2000	F-41
Consolidated Statements of Changes in Shareholders' Equity for the three years ended December 31, 2002, 2001 and 2000	F-42
Consolidated Statements of Cash Flows for the three years ended December 31, 2002, 2001 and 2000	F-43
Notes to Consolidated Financial Statements	F-44

F-1

INDEPENDENT AUDITORS' REPORT

The Board of Directors Impac Mortgage Holdings, Inc.:

We have audited the accompanying consolidated balance sheets of Impac Mortgage Holdings, Inc. and subsidiaries as of December 31, 2002 and 2001 and the related consolidated statements of operations and comprehensive earnings (loss), changes in stockholders' equity and cash flows for each of the years in the three-year period ended December 31, 2002. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of Impac Mortgage Holdings, Inc. and subsidiaries as of December 31, 2002 and 2001 and the results of their operations and their cash flows for each of the years in the three-year period ended December 31, 2002 in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note A to the consolidated financial statements, the Company changed its method of accounting for derivative instuments and hedging activities in 2001.

KPMG LLP

Orange County, California January 29, 2003, except as to last paragraph to Note M to the consolidated financial statements, which is as of February 27, 2003

F-2

CONSOLIDATED BALANCE SHEETS

(dollar amounts in thousands, except share data)

		At December	31,	
		2002	2001	
ASSETS				
Cash and cash equivalents	\$	113,345 \$	51,887	
Investment securities available-for-sale		26,065	32,989	
Loans Receivable:				
CMO collateral		5,149,680	2,229,168	
Finance receivables, net of pledge balances		1,140,248	466,649	
Mortgages held-for-investment		57,536	20,078	
Allowance for loan losses		(26,602)	(11,692)	
Net loans receivable		6,320,862	2,704,203	
Accrued interest receivable		28,287	14,565	
Investment in Impac Funding Corporation		20,787	19,126	
Due from Impac Funding Corporation		14,500	14,500	
Other real estate owned		11,116	8,137	
Derivative assets		14,931	5,128	
Other assets		1,880	4,199	
Total assets	\$	6,551,773 \$	2,854,734	
LIABILITIES AND STOCKHOLDERS' EQUITY				
CMO borrowings	\$	5,041,751 \$	2,151,400	
Reverse repurchase agreements		1,168,029	469,491	
Borrowings secured by investment securities available-for-sale		7,134	12,997	
Accrued dividends payable		21,754	14,081	
Other liabilities		9,617	3,400	
Total liabilities		6,248,285	2,651,369	
Commitments and contingencies				
Stockholders' equity:				
Preferred stock, \$0.01 par value; 7,500,000 shares authorized; none issued and outstanding at December 31, 2002 and 2001 Series A junior participating preferred stock, \$0.01 par value; 2,500,000 shares authorized; none issued and outstanding at December 31, 2002 and 2001		_	_	
Common stock, \$0.01 par value; 200,000,000 and 50,000,000 shares authorized and 45,320,517 and 32,001,997 shares				
issued and outstanding at December 31, 2002 and 2001, respectively		453	320	
Additional paid-in capital		479,298	359,279	
Accumulated other comprehensive loss		(41,721)	(19,857)	
Notes receivable from common stock sales		_	(920)	
Net accumulated deficit:				
Cumulative dividends declared		(200,954)	(126,952)	
Retained earnings (accumulated deficit)		66,412	(8,505)	
Net accumulated deficit	_	(134,542)	(135,456)	
Total stockholders' equity		303,488	203,365	
Total liabilities and stockholders' equity	\$	6,551,773 \$	2,854,734	

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE EARNINGS (LOSS)

(in thousands, except per share data)

	For the y	For the year ended December 31		
	2002	2001	2000	
INTEREST INCOME:				
Mortgage Assets	\$ 221,978	\$ 153,951	\$ 144,735	
Other interest income	4,438	2,664	2,344	
Total interest income	226,416	156,615	147,079	
INTEREST EXPENSE:				
CMO borrowings	117,756	77,813	80,287	
Reverse repurchase agreements	23,584	33,370	42,433	
Other borrowings	3,467	829	1,376	
Total interest expense	144,807	112,012	124,096	
Net interest income	81,609	44,603	22,983	
Provision for loan losses	19,848	16,813	18,839	
Net interest income after provision for loan losses	61,761	27,790	4,144	
NON-INTEREST INCOME:				
Equity in net earnings (loss) of Impac Funding Corporation	17,073	10,912	(1,762	
Servicing fees	17,073	751	690	
Gain on sale of securities		312	050	
Other income	4,325	5,404	3,585	
	4,323	5,404	5,505	
Total non-interest income	21,582	17,379	2,513	
NON-INTEREST EXPENSE:				
Professional services	3,649	2,747	2,604	
Personnel expense	1,868	1,211	666	
General and administrative and other expense	1,716	2,303	2,230	
Write-down on investment securities available-for-sale	1,039	2,217	53,576	
(Gain) loss on sale of other real estate owned	154	(1,931)	1,814	
Mark-to-market loss - SFAS 133	—	3,821	_	
Total non-interest expense	8,426	10,368	60,890	
Earnings (loss) before extraordinary item and cumulative effect of change in accounting principle.	74,917	34,801	(54,233	
Extraordinary item		(1,006)		
Cumulative effect of change in accounting principle	—	(617)		
Net earnings (loss)	74,917	33,178	(54,233	
Less: Cash dividends on cumulative convertible preferred stock		(1,575)		
Net earnings (loss) available to common stockholders	74,917	31,603	(57,383	
Other comprehensive earnings (loss):				
Unrealized gains (losses) on securities:				
Unrealized holding gains (losses) arising during year	(21,752)	(21,094)	(795	
Less: Reclassification of gains (losses) included in income	(112)	1,805	7,806	
Net unrealized gains (losses) arising during year	(21,864)	(19,289)	7,011	
Comprehensive earnings (loss)	\$ 53,053	\$ 13,889	\$ (47,222	
		-,		

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE EARNINGS (LOSS)—(continued)

(in thousands)

	 For the year ended December 31,				
	 2002		2001		2000
Net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting principle:					
Net earnings (loss) per share—basic	\$ 1.87	\$	1.41	\$	(2.70)
Net earnings (loss) per share—diluted	\$ 1.84	\$	1.25	\$	(2.70)
Net earnings (loss) per share:					
Net earnings (loss) per share—basic	\$ 1.87	\$	1.34	\$	(2.70)
Net earnings (loss) per share—diluted	\$ 1.84	\$	1.19	\$	(2.70)
Dividends declared per common share	\$ 1.76	\$	0.69	\$	0.36

See accompanying notes to consolidated financial statements.

F-5

CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY

(dollar amounts in thousands, except share data)

	Number of Preferred Shares Outstanding	Preferred Stock	Number of Common Shares Outstanding	Common Stock	Additional Paid-In Capital	Accumulated Other Comprehensive Loss	Notes Receivable Common Stock Sales	Cumulative Dividends Declared	Retained Earnings (Accumulated Deficit)	Total Stockholders' Equity
Balance, December 31, 1999	1,200,000	\$ 12	21,400,906	\$ 214	\$ 327,632	\$ (7,579)	\$ (905)	\$ (93,080)	\$ 12,550	\$ 238,844
Dividends declared (\$0.36 per common share) Dividends declared on preferred		_		_	_	_	_	(7,743)	_	(7,743)
shares	_	_	_	_	_	_	_	(3,150)	_	(3,150)
Repurchase of common stock Payments on notes receivable from common stock sales	_	_	(990,950)	(10)	(2,282)		3	_	_	(2,292) 3
Net loss, 2000	_	_	_	_	_	_	_	_	(54,233)	(54,233)
Other comprehensive earnings	_					7,011		—		7,011
Balance, December 31, 2000	1,200,000	12	20,409,956	204	325,350	(568)	(902)	(103,973)	(41,683)	178,440
Dividends declared (\$0.69 per common share)	_	_	_	_	_	_	_	(21,404)	_	(21,404)
Dividends declared on preferred shares	_	_	_	_	_	_	_	(1,575)	_	(1,575)
Repurchase of common stock	_	_	(25,000)	_	(54)	_	_	_	_	(54)
Common stock offering Proceeds from exercise of stock	_	_	5,100,000	51	33,482	_	_	_	_	33,533
options	_	_	161,109	1	553	_	_	_	_	554
Conversion of preferred shares to common shares	(1,200,000)	(12)	6,355,932	64	(52)	_	_	_	_	_
Advances on notes receivable from common stock sales	_	_	_	_	_	_	(18)	_	_	(18)
Net earnings, 2001	_	_	_	_	_	_	_	_	33,178	33,178
Other comprehensive loss			_		_	(19,289)		_		(19,289)
Balance, December 31, 2001	_	_	32,001,997	320	359,279	(19,857)	(920)	(126,952)	(8,505)	203,365
Dividends declared (\$1.76 per common share)	_	_	_	_	_	_	_	(74,002)	_	(74,002)
Common stock offering	_	_	11,197,000	112	98,004	_	_	_	_	98,116
Proceeds from exercise of stock options	_	_	150,824	1	992	_	_	_	—	993
Sale of stock via Structured Equity Shelf	_	_	1,970,696	20	21,023	_	_	_	_	21,043
Advances on notes receivable from common stock sales	_	_	_	_	_	_	920	_	_	920
Net earnings, 2002	_	_	_	_	_	_	_	_	74,917	74,917
Other comprehensive loss						(21,864)		_		(21,864)
Balance, December 31, 2002	_	\$	45,320,517	\$ 453	\$ 479,298	\$ (41,721)	\$ —	\$ (200,954)	\$ 66,412	\$ 303,488

See accompanying notes to consolidated financial statements.

F-6

CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

	For the year ended December 31,				
		2002	2001	2000	
CASH FLOWS FROM OPERATING ACTIVITIES:					
Earnings (loss) before cumulative effect of change in accounting principle	\$	74,917 \$	33,795 \$	(54,233)	
Adjustments to reconcile net earnings (loss) to net cash provided by operating activities:					
Cumulative effect of change in accounting principle		_	(617)	_	
Equity in net (earnings) loss of Impac Funding Corporation		(17,073)	(10,912)	1,762	
Provision for loan losses		19,848	16,813	18,839	
Amortization of CMO premiums and deferred securitization costs		37,945	14,973	15,308	
Net change in accrued interest receivable		(13,722)	(1,577)	(1,779)	
Write-down of securitization costs			1,006	_	
Write-down of investment securities available-for-sale		1,039	2,217	53,576	
Gain (loss) on sale of other real estate owned		(154)	1,931	(1,814)	
Gain on sale of investment securities available-for-sale			(312)	—	
Net change in other assets and liabilities		(1,267)	(1,858)	(4,939)	
Net cash provided by operating activities		101,533	55,459	26,720	
CASH FLOWS FROM INVESTING ACTIVITIES:					
Net change in CMO collateral		(2,994,039)	(907,189)	312,171	
Net change in finance receivables		(673,599)	(61,817)	(208,966)	
Net change in mortgages held-for-investment		(44,235)	(16,091)	(433,340)	
Sale of investment securities available-for-sale		_	3,901	5,704	
Dividends from Impac Funding Corporation		12,870	8,894	—	
Principal reductions on investment securities available-for-sale		8,704	5,542	3,864	
Proceeds from sale of other real estate owned		12,455	5,722	18,091	
Net cash used in investing activities		(3,677,844)	(961,038)	(302,476)	
CASH FLOWS FROM FINANCING ACTIVITIES:					
Net change in reverse repurchase agreements and other borrowings		692,675	62,824	(150,955)	
Proceeds from CMO borrowings		3,876,008	1,500,921	943,558	
Repayment of CMO borrowings		(985,657)	(640,805)	(503,091)	
Dividends paid		(66,329)	(9,686)	(13,675)	
Retirement of senior subordinated debentures			(7,747)	_	
Proceeds from sale of common stock		119,159	33,533		
Repurchase of common stock			(54)	(2,292)	
Proceeds from exercise of stock options		993	554	_	
Reduction (advances) on notes receivable from common stock sales		920	(18)	3	
Net cash provided by financing activities		3,637,769	939,522	273,548	
Net change in cash and cash equivalents		61,458	33,943	(2,208)	
Cash and cash equivalents at beginning of year		51,887	17,944	20,152	
Cash and cash equivalents at end of year	\$	113,345 \$	51,887 \$	17,944	

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF CASH FLOWS —(continued)

(in thousands)

	 For the year ended December 31,			
	 2002	2001	2000	
SUPPLEMENTARY INFORMATION:				
Interest paid	\$ 140,935 \$	113,384 \$	125,391	
NON-CASH TRANSACTIONS:				
Accumulated other comprehensive gain (loss)	\$ (21,864) \$	(19,289) \$	7,011	
Transfer of mortgages held-for-investment to other real estate owned	1,839	2,522	3,179	
Transfer of CMO collateral to other real estate owned	13,441	7,993	8,300	
Transfer of finance receivables to other real estate owned	—	606	647	
Dividends declared and unpaid	21,754	14,081	788	

See accompanying notes to consolidated financial statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note A—Summary of Business and Significant Accounting Policies

1. Business and Financial Statement Presentation

Impac Mortgage Holdings, Inc. (IMH), a Maryland corporation incorporated in August 1995, and its subsidiaries, IMH Assets Corp. (IMH Assets), Impac Warehouse Lending Group, Inc. (IWLG), Impac Multifamily Capital Corporation (IMCC) and its affiliate Impac Funding Corporation (IFC), together with its wholly-owned subsidiaries Impac Secured Assets Corp. (ISAC) and Novelle Financial Services, Inc. (Novelle) are collectively (the Company). References to Impac Mortgage Holdings, Inc., or "IMH," are made to differentiate IMH, the publicly traded Company, as a separate entity from IMH Assets, IWLG, IMCC and IFC.

IMH is a mortgage real estate investment trust (REIT). Together with its subsidiaries and affiliate it is a nationwide acquirer and originator of nonconforming Alt-A mortgage loans (Alt-A mortgages). Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but have loan characteristics that make them non-conforming under those guidelines. Some of the principal differences between mortgages purchased by Fannie Mae and Freddie Mac and Alt-A mortgages are as follows:

- credit and income histories of the mortgagor;
- documentation required for approval of the mortgagor; and
- applicable loan-to-value ratios.

The Company operates three core businesses:

- long-term investment operations;
- mortgage operations; and
- warehouse lending operations.

The long-term investment operations invest primarily in adjustable and fixed rate Alt-A mortgages that are acquired and originated by the mortgage operations. This business primarily generates net interest income from its mortgage loan investment portfolio and, to a lesser extent, its investment securities portfolio. Investment in Alt-A mortgages is financed with collateralized mortgage obligations (CMO) financing, warehouse facilities and proceeds from the sale of capital stock.

The mortgage operations acquire, originate, sell and securitize primarily adjustable and fixed rate Alt-A mortgages. The mortgage operations generate income by securitizing and selling loans to permanent investors, including the long-term investment operations. This business also earns revenue from fees associated with mortgage servicing rights, master servicing agreements and interest income earned on mortgages held-for-sale. The mortgage operations use warehouse facilities provided by the warehouse lending operations to finance the acquisition and origination of mortgages.

The warehouse lending operations provide short-term financing to mortgage loan originators, including the mortgage operations, by funding mortgages from their closing date until they are sold to pre-approved investors. This business earns fees from warehouse transactions as well as net interest income from the difference between its cost of borrowings and the interest earned on warehouse advances.

The financial condition and operations of IMH have been presented in the consolidated financial statements for the three-year period ended December 31, 2002 and include the financial results of IMH, IMH Assets, IWLG and IMCC as stand-alone entities and the financial results of IMH's equity interest in net earnings (loss) in IFC as a stand-alone entity.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

IMH is entitled to 99% of the earnings or losses of IFC through its ownership of all of the non-voting preferred stock of IFC. As such, IMH records its investment in IFC using the equity method. Under this method, original investments are recorded at cost and adjusted by IMH's share of earnings or losses. Certain officers and directors of IMH own all of the common stock of IFC and are entitled to 1% of the earnings or losses of IFC. Gain on the sale of mortgages or securities by IFC to IMH are deferred and accreted into gain on sale over the estimated life of the mortgages or securities using the interest method. All significant intercompany balances and transactions with IMH's consolidated subsidiaries have been eliminated in consolidation. Interest income and expense on affiliated shortterm advances or borrowings has been earned or paid at the rate of 8.0% per annum. Certain amounts in the prior periods' consolidated financial statements have been reclassified to conform to the current presentation.

Management has made a number of estimates and assumptions relating to the reporting of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period to prepare these financial statements in conformity with accounting principles generally accepted in the United States of America. Actual results could differ from those estimates.

2. Cash and Cash Equivalents

For purposes of the consolidated statements of cash flows, cash and cash equivalents consist of cash and money market mutual funds. Investments with maturities of three months or less at date of acquisition are considered to be cash equivalents.

3. Investment Securities Available-for-Sale

Investment and mortgage-backed securities are classified as held-to-maturity, available-for-sale, and/or trading securities. Held-to-maturity investment and mortgage-backed securities are reported at amortized cost, available-for-sale securities are reported at fair value with unrealized gains and losses as a separate component of stockholders' equity and trading securities are reported at fair value with unrealized gains and losses reported at fair value with unrealized gains and losses on sale of investment securities available-for-sale are based on the specific identification method. As IMH qualifies as a Real Estate Investment Trust (REIT), and no income taxes are paid, the unrealized gains and losses are reported gross in stockholders' equity. Premiums or discounts obtained on investment securities are accreted or amortized to interest income over the estimated life of the investment securities using the interest method. Investments securities may be subject to credit, interest rate and/or prepayment risk.

From time to time, IMH purchases residuals created by IFC as a result of the sale of mortgages through securitizations. However, IMH has not purchased mortgage-backed securities, including residuals, from IFC since 1999. IFC sells a portfolio of mortgages to a special purpose entity, ISAC, which has been established for the limited purpose of buying and reselling the mortgages. IFC then transfers the same mortgages to a special purpose entity or owners trust (the Trust). The Trust issues interest-bearing asset-backed securities generally in an amount equal to the aggregate principal balance of the mortgages. IFC typically sells these certificates at face value and without recourse. Representations and warranties customary to the mortgage banking industry are provided by IFC. IMH or other investors purchase these certificates from the Trust and the proceeds from the sale of the certificates are used as consideration to purchase the underlying mortgages. In addition, IFC may provide a credit enhancement for the benefit of the investors in the form of additional collateral held by the Trust. An over-collateralization account is required to be maintained at specified levels.

4. CMO Collateral and Mortgages Held-for-Investment

IMH purchases from IFC primarily Alt-A mortgages to be held as long-term investment. Mortgages held-for-investment and CMO collateral are recorded at cost as of the date of purchase. Mortgages held-for-investment and CMO collateral include various types of adjustable and fixed rate mortgages secured by single-family residential and multi-family real estate properties originated by IMCC and, to a lesser extent, fixed rate second trust deeds secured by

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

single-family residential real estate properties. Premiums and discounts, which may result from the purchase or acquisition of mortgages in excess of the outstanding principal balance, are amortized to interest income over their estimated lives using the interest method as an adjustment to the carrying amount of the mortgage. Prepaid securitization costs related to the issuance of CMOs are amortized to interest expense over their estimated lives using the interest method. Mortgages held-for-investment and mortgages held as CMO collateral are continually evaluated for collectibility and, if appropriate, the mortgage may be placed on non-accrual status, generally when the mortgage is 90 days past due, and previously accrued interest reversed from income.

5. Finance Receivables

Finance receivables represent transactions with customers, including affiliated companies, involving residential real estate lending. As a warehouse lender, IWLG is a secured creditor of the mortgage bankers and brokers to which it extends credit and is subject to the risks inherent in that status including, the risk of borrower default and bankruptcy. Any claim of IWLG as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay. Finance receivables represent warehouse lines of credit with mortgage banking companies collateralized by mortgages on single-family residences. Finance receivables are stated at the principal balance outstanding. Interest income is recorded on the accrual basis in accordance with the terms of the mortgages.

6. Allowance for Loan Losses

An allowance is maintained for losses on mortgages held-for-investment, mortgages held as CMO collateral and finance receivables at an amount that management believes is sufficient to provide adequate protection against future losses in the mortgage loan investment portfolio. The allowance for losses is determined primarily on management's judgment of net loss potential changes in the nature and volume of the portfolio, value of the collateral and current economic conditions that may affect the borrowers' ability to pay. A provision is recorded for loans deemed to be uncollectible thereby increasing the allowance for loan losses. Subsequent recoveries on mortgages previously charged off are credited back to the allowance. Provision for loan losses is also determined based on the following:

- management's judgment of the net loss potential of mortgages in our mortgage loan investment portfolio based on prior loan loss experience;
- changes in the nature and volume of the mortgage loan investment portfolio;
- value of the collateral;
- delinquency trends; and
- current economic conditions that may affect the borrowers' ability to pay.

7. Premises and Equipment

Premises and equipment are stated at cost, less accumulated depreciation or amortization. Depreciation on premises and equipment is recorded using the straight-line method over the estimated useful lives of individual assets, typically, three to twenty years.

8. CMO Borrowings

The decision to issue CMOs is based on our current and future investment needs, market conditions and other factors. CMOs, which are primarily secured by Alt-A mortgages on single-family and multi-family residential real

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

property, are issued as a means of financing its long-term investment operations. CMOs are carried at their outstanding principal balances including accrued interest on such obligations. For accounting and tax purposes, mortgages financed through the issuance of CMOs are treated as assets and the CMOs are treated as debt when for accounting purposes the CMO qualifies as a financing arrangement. Each issue of CMOs is fully payable from the principal and interest payments on the underlying mortgages collateralizing such debt and any investment income on such collateral. CMOs typically are structured as one-month London Interbank Offered Rate (LIBOR) "floaters" and fixed rate securities with interest payable monthly. The maturity of each class of CMO is directly affected by the rate of principal prepayments on the related CMO collateral. Each CMO series is also subject to redemption according to specific terms of the respective indentures. As a result, the actual maturity of any class of a CMO series is likely to occur earlier than the stated maturities of the underlying mortgages.

When we issue CMOs for financing purposes, we seek an investment grade rating for our CMOs by a nationally recognized rating agency. To secure such a rating, it is often necessary to pledge collateral in excess of the principal amount of the CMOs to be issued, or to obtain other forms of credit enhancements such as additional mortgage insurance. The need for additional collateral or other credit enhancements depends upon factors such as the type of collateral provided, the interest rates paid, the geographic concentration of the mortgaged property securing the collateral and other criteria established by the rating agencies. The pledge of additional collateral reduces our capacity to raise additional funds through short-term secured borrowings or additional CMOs and diminishes the potential expansion of our mortgage loan investment portfolio. As a result, our objective is to pledge additional collateral for CMOs only in the amount required to obtain an investment grade rating for our CMOs by a nationally recognized rating agency. Our total loss exposure is limited to the equity invested in the CMOs at any point in time.

A net interest spread between interest income on mortgages securing CMOs and interest and other expenses associated with CMO financing is earned. Net interest spreads may be directly impacted by levels of early prepayment of underlying mortgages and, to the extent each CMO class has variable rates of interest, may be affected by changes in short-term interest rates.

9. Income Taxes

IMH operates so as to qualify as a REIT under the requirements of the Internal Revenue Code (the Code). Requirements for qualification as a REIT include various restrictions on ownership of IMH's stock, requirements concerning distribution of taxable income and certain restrictions on the nature of assets and sources of income. A REIT must distribute at least 90% of its taxable income to its stockholders of which 85% must be distributed within the taxable year in order to avoid the imposition of an excise tax and the remaining balance may extend until timely filing of its tax return in its subsequent taxable year. Qualifying distributions of its taxable income are deductible by a REIT in computing its taxable income. If in any tax year IMH should not qualify as a REIT, it would be taxed as a corporation and distributions to the stockholders would not be deductible in computing taxable income. If IMH were to fail to qualify as a REIT in any tax year, it would not be permitted to qualify for that year and the succeeding four years. As of December 31, 2002 IMH has estimated federal and state net operating loss tax carry-forwards of \$16.7 million, which expire in the year 2020, that are available to offset future taxable income.

10. Net Earnings (Loss) per Share

Basic net earnings (loss) per share are computed on the basis of the weighted average number of shares outstanding for the year divided by net earnings (loss) for the year. Diluted net earnings (loss) per share are computed on the basis of the weighted average number of shares and dilutive common equivalent shares outstanding for the year divided by net earnings (loss) for the year.

11. Stock Options

Stock options and awards may be granted to the directors, officers and key employees of the Company. The exercise price for any NQSO or ISO granted under the Stock Option Plans may not be less than 100% (or 110% in the case of ISOs granted to an employee who is deemed to own in excess of 10% of the outstanding common stock) of the

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

fair market value of the shares of common stock at the time the NQSO or ISO is granted. Grants under stock option plans are made and administered by the board of directors. IMH currently has a 1995 Stock Option, Deferred Stock and Restricted Stock Plan (1995 Plan). During 2001 the board of directors and stockholders approved a new Stock Option, Deferred Stock and Restricted Stock Plan (2001 Plan) (collectively, the Stock Option Plans). Each Stock Option Plan provides for the grant of qualified incentive stock options (ISOs), options not qualified (NQSOs), deferred stock, and restricted stock, and, in the case of the 2001 Plan, dividend equivalent rights and, in the case of the 1995 Plan, stock appreciation rights and limited stock appreciation rights awards (Awards).

In December 2002 the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation – Transition and Disclosure" (SFAS 148), an amendment of FASB Statement No. 123, "Accounting for Stock-Based Compensation," (FASB 123). SFAS 148 amends FASB 123 to provide alternative methods of transition for a voluntary change to the fair value method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of SFAS 123 to require prominent disclosures in both annual and interim financial statements. On December 15, 2002 IMH adopted the disclosure requirements of SFAS 148.

In November 1995 the FASB issued SFAS No. 123, "Accounting for Stock-Based Compensation" (SFAS 123). This statement establishes financial accounting standards for stock-based employee compensation plans. SFAS 123 permits management to choose either a new fair value based method or the current APB 25 intrinsic value based method of accounting for its stock-based compensation arrangements. SFAS 123 requires pro forma disclosures of net earnings (loss) computed as if the fair value based method had been applied in financial statements of companies that continue to follow current practice in accounting for such arrangements under APB 25. SFAS 123 applies to all stock-based employee compensation plans in which an employer grants shares of its stock or other equity instruments to employees except for employee stock ownership plans. SFAS 123 also applies to plans in which the employer incurs liabilities to employees in amounts based on the price of the employer's stock, i.e., stock option plans, stock purchase plans, restricted stock plans and stock appreciation rights. The statement also specifies the accounting for transactions in which a company issues stock options or other equity instruments for services provided by non-employees or to acquire goods or services from outside suppliers or vendors.

At December 31, 2002, the Company had a stock-based compensation plan. IMH applied APB Opinion No. 25 in accounting for its plans. No compensation cost has been recognized for fixed stock option plans. Had compensation cost for the Company's stock-based compensation plans been determined consistent with SFAS 123, the Company's net earnings and earnings per share would have been reduced to the pro forma amounts indicated below (in thousands of dollars, except per share amounts):

	For the year ended December 31,					
		2002		2001		2000
Net earnings (loss) before extraordinary item and cumulative effect of change in accounting principle as reported	\$	74,917	\$	34,801	\$	(54,233)
Less: Total stock-based employee compensation expense determined under fair value based method for all awards		(556)		(308)		(73)
Pro forma net earnings (loss)	\$	74,361	\$	34,493	\$	(54,306)
Net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting principle as reported:						
Basic	\$	1.87	\$	1.41	\$	(2.70)
Diluted	\$	1.84	\$	1.25	\$	(2.70)
Pro forma net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting principle:						
Basic	\$	1.85	\$	1.40	\$	(2.70)
Diluted	\$	1.82	\$	1.23	\$	(2.70)
Net earnings (loss) per share as reported:						
Basic	\$	1.87	\$	1.34	\$	(2.70)
Diluted	\$	1.84	\$	1.19	\$	(2.70)
Pro forma net earnings (loss):						
Basic	\$	1.85	\$	1.33	\$	(2.70)
Diluted	\$	1.82	\$	1.18	\$	(2.70)

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The derived fair value of the options granted during 2002, 2001 and 2000 was approximately \$1.19, \$1.18 and \$0.75, respectively. The fair value of options granted, which is amortized to expense over the option vesting period in determining the pro forma impact, is estimated on the date of grant using the Black-Scholes option pricing model with the following weighted average assumptions:

	For the year ended December 31,			
	2002	2001	2000	
Risk-free interest rate	1.45%	2.22%	4.80%	
Expected lives (in years)	3-10	3-10	1-3	
Expected volatility	35.07%	52.87%	62.64%	
Expected dividend yield	10.00%	10.00%	11.20%	

12. Derivative Instruments and Hedging Activities

In June 1998 the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards (SFAS) No. 133, "Accounting for Derivative Instruments and Hedging Activities," as amended by SFAS No. 137 and SFAS No. 138, collectively, (SFAS 133). SFAS 133 establishes accounting and reporting standards for derivative instruments, including a number of derivative instruments embedded in other contracts, collectively referred to as derivatives, and for hedging activities. It requires that an entity recognize all derivatives as either assets or liabilities in the statement of financial position and measure those instruments at fair value. If specific conditions are met, a derivative may be specifically designated as (1) a hedge of the exposure to changes in the fair value of a recognized asset or liability or an unrecognized firm commitment; (2) a hedge of the exposure to variable cash flows of a forecasted transaction; or (3) a hedge of the foreign currency exposure of a net investment in a foreign operation, an unrecognized firm commitment, an available for sale security or a foreign-currency-denominated forecasted transaction.

On January 1, 2001 IMH adopted SFAS 133 and the fair value of derivatives were reflected in financial condition and results of operations. On August 10, 2001 the Derivatives Implementation Group (DIG) of the FASB published DIG G20 which further interpreted SFAS 133. On October 1, 2001 IMH adopted the provisions of DIG G20 and net income and accumulated other comprehensive income were adjusted by the amount needed to reflect the cumulative impact of adopting the provisions of DIG G20.

Management follows a hedging program intended to limit its exposure to changes in interest rates primarily associated with cash flows on CMO borrowings. Management's primary objective is to hedge exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of adjustable rate CMO borrowings. Management also monitors on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Management's hedging program is formulated with the intent to offset the potential adverse effects of changing interest rates on cash flows on CMO borrowings resulting from the following:

- interest rate adjustment limitations on CMO collateral due to periodic and lifetime interest rate cap features; and
- mismatched interest rate adjustment periods between mortgages held as CMO collateral and CMO borrowings.

To mitigate exposure to the effect of changing interest rates on cash flows on CMO borrowings, IMH purchases derivatives in the form of interest rate cap agreements (Caps) interest rate floor agreements (Floors) and interest rate

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

swap agreements (Swaps). A Cap or Floor is a contractual agreement for which IMH pays a fee. If prevailing interest rates reach levels specified in the Cap or Floor agreement IMH may either receive or pay cash. A Swap is generally a contractual agreement that obligates one party to receive or make cash payments based on an adjustable rate index and the other party to receive or make cash payments based on a fixed rate. Swaps have the effect of fixing borrowing costs on a similar amount of Swaps and, as a result, can reduce the interest rate variability of borrowings. Management's objective is to lock in a reliable stream of cash flows when interest rates fall below or rise above certain levels. For instance, when interest rates rise, borrowing costs may increase at greater speeds than the underlying collateral supporting the borrowings. These derivatives hedge the variability of forecasted cash flows attributable to CMO borrowings and protect net interest income by providing cash flows at certain triggers during changing interest rate environments. In all hedging transactions counter-parties must have a highly rated credit rating as determined by various credit rating agencies.

Caps qualify as derivatives under provisions of SFAS 133. The hedging instrument is the specific LIBOR Cap that is hedging the LIBOR based CMO borrowings. The nature of the risk being hedged is the variability of the cash flows associated with the LIBOR borrowings. Prior to the adoption of DIG G20 management assessed the hedging effectiveness of its Caps utilizing only the intrinsic value of the Caps. DIG G20 allows management to utilize the terminal value of the Caps to assess effectiveness. DIG G20 allows for amortization of the initial fair value of the Caps over the life of the Caps based on the maturity date of the individual caplets. Upon adoption of DIG G20 net income and accumulated other comprehensive income were adjusted by the amount needed to reflect the cumulative impact of adopting the provisions of DIG G20. Subsequent to the adoption of DIG G20 Caps are considered effective hedges and are marked to market each reporting period with the entire change in market value being recognized in accumulated other comprehensive income.

Floors qualify as cash flow hedges under the provisions of SFAS 133. The hedging instrument is the specific LIBOR Floor that is hedging the LIBOR based CMO borrowings. The nature of the risk being hedged is the variability of the cash flows associated with the LIBOR borrowings. Prior to DIG G20 these derivatives were marked to market with the entire change in the market value of the intrinsic component recognized in accumulated other comprehensive income each reporting period. The time value component of these agreements were marked to market and recognized in non-interest expense in the statement of operations. Subsequent to the adoption of DIG G20 these derivatives are marked to market with the entire change in the market value recognized in accumulated other comprehensive income.

Effectiveness of derivatives is measured by the fact that the hedged item, CMO borrowings, and the derivative is based on one-month LIBOR. As both instruments are tied to the same index, the hedge is expected to be highly effective both at inception and on an ongoing basis. Management assesses the effectiveness and ineffectiveness of the hedging instruments at the inception of the hedge and at each reporting period. Based on the fact that, at inception, the critical terms of the hedges and forecasted CMO borrowings are the same, management has concluded that the changes in cash flows attributable to the risk being hedged are expected to be completely offset by the derivatives, subject to subsequent assessments that the critical terms have not changed.

13. Recent Accounting Pronouncements

In June 2001 FASB issued SFAS No. 143, "Accounting for Asset Retirement Obligations" (SFAS No. 143) requires the Company to record the fair value of an asset retirement obligation as a liability in the period in which it incurs a legal obligation associated with the retirement of tangible long-lived assets that result from the acquisition, construction, development, and/or normal use of the assets. The Company also records a corresponding asset that is depreciated over the life of the asset. Subsequent to the initial measurement of the asset retirement obligation, the obligation will be adjusted at the end of each period to reflect the passage of time and changes in the estimated future cash flows underlying the obligation. The Company is required to adopt SFAS No. 143 on January 1, 2003. The adoption of SFAS No. 143 did not expected to have a material effect on the Company's financial statements.

In April 2002 the FASB issued SFAS No. 145, "Rescission of FASB Statements No. 4, 44 and 64, Amendment of FASB Statement No. 13 and Technical Corrections" (SFAS 145). SFAS 145 amends existing guidance on reporting

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

gains and losses on the extinguishment of debt to prohibit the classification of the gain or loss as extraordinary, as the use of such extinguishments have become part of the risk management strategy of many companies. SFAS 145 also amends Statement No. 13 to require sale-leaseback accounting for certain lease modifications that have economic effects similar to sale-leaseback transactions. The provisions of the statement related to the rescission of Statement No. 4 is applied in fiscal year's beginning after May 15, 2002. Earlier application of these provisions is encouraged. The provisions of the statement related to Statement No. 13 were effective for transactions occurring after May 15, 2002, with early application encouraged. The adoption of SFAS 145 is not expected to have a material effect on the Company's financial statements.

In June 2002 the FASB issued SFAS No. 146, "Accounting for Costs Associated with Exit or Disposal Activities" (SFAS 146). SFAS 146 addresses financial accounting and reporting for costs associated with exit or disposal activities and nullifies Emerging Issues Task Force (EITF) Issue 94-3, Liability Recognition for Certain Employee Termination Benefits and Other Costs to Exit an Activity. The provisions of SFAS 146 are effective for exit or disposal activities that are initiated after December 31, 2002, with early application encouraged. The adoption of SFAS 146 is not expected to have a material effect on the Company's financial statements.

In November 2002 the FASB issued Interpretation No. 45, Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness to Others, an interpretation of FASB Statements No. 5, 57 and 107 and a rescission of FASB Interpretation No. 34. This Interpretation elaborates on the disclosures to be made by a guarantor in its interim and annual financial statements about its obligations under guarantees issued. The Interpretation also clarifies that a guarantor is required to recognize, at inception of a guarantee, a liability for the fair value of the obligation undertaken. The initial recognition and measurement provisions of the Interpretation are applicable to guarantees issued or modified after December 31, 2002 and are not expected to have a material effect on the Company's financial statements. The disclosure requirements are effective for financial statements of interim or annual periods ending after December 15, 2002.

In December 2002 the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation – Transition and Disclosure" (SFAS 148), an amendment of FASB Statement No. 123. SFAS 148 amends FASB Statement No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition for a voluntary change to the fair value method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of Statement No. 123 to require prominent disclosures in both annual and interim financial statements. Certain portions of the disclosure modifications are required for fiscal years ending after December 15, 2002 and are included in the notes to these consolidated financial statements. The Company has adopted the disclosure requirement of SFAS 148 and will continue to account for stock options using the intrinsic value method.

In January 2003 the FASB issued Interpretation No. 46, "Consolidation of Variable Interest Entities, an interpretation of ARB No. 51" (FIN 46). FIN 46 addresses the consolidation by business enterprises of variable interest entities as defined in the Interpretation. FIN 46 applies immediately to variable interests in variable interest entities created after January 31, 2003 and to variable interests in variable interest entities obtained after January 31, 2003. For nonpublic enterprises, such as IFC, with a variable interest in a variable interest entity created before February 1, 2003 the Interpretation is applied to the enterprise no later than the end of the first reporting period beginning after June 15, 2003. IMH is subject to a majority of the risk of loss from IFC and will therefore need to be included as a consolidated entity upon adoption of FIN 46. The result of this consolidation would be the elimination of certain inter-company balances and certain gains or losses. The adoption of FIN 46 will likely be immaterial on the results of operations of the Company as 99% of the earnings or losses of IFC are currently included in the Company's financial statements through its ownership of all of the non-voting preferred stock of IFC. However, the Company's balance sheet will reflect IFC's assets when consolidated. For more information about the financial condition and results of operations of Impac Funding Corporation refer to Note U—Impac Funding Corporation.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note B—Investment Securities Available-for-Sale

Investment securities available-for-sale are primarily secured by conventional, one-to-four family mortgages. The yield to maturity on each security depends on, among other things, the rate and timing of principal payments, including prepayments, repurchases, defaults and liquidations, the pass-through rate, and interest rate fluctuations. Interest in these securities is subordinated so that, in the event of a loss, payments to senior certificate holders will be made before IMH receives its payments. During 2000 IMH wrote-off \$52.6 million of investment securities available-for-sale that were investment securities secured by high loan-to-value second trust deeds, franchise loans receivable and certain sub-prime subordinated securities all of which were acquired prior to 1998. Subsequent to 1997 IMH has only acquired or invested in securities that are secured by mortgages underwritten and purchased by the mortgage operations. However, no investment securities have been purchased from IFC since 1999. During the years ended December 31, 2002 and 2001 IMH sold investment securities available-for-sale with a carrying value of none and \$3.6 million, respectively.

The amortized cost and estimated fair value of investment securities available-for-sale for the periods indicated are presented as follows:

	An	nortized Cost	Gross Unrealized Gain (in thous	Gross Unrealized Loss	Estimated Fair Value
As of December 31, 2002:					
Subordinated securities secured by mortgages	\$	17,710 \$	8,355 \$	_	\$ 26,065
As of December 31, 2001:					
Subordinated securities secured by mortgages	\$	26,661 \$	7,286 \$	958	\$ 32,989

To determine the fair value of investment securities available-for-sale, management must estimate future rates of mortgage prepayments, prepayment penalties to be received, delinquency rates, constant default rates and default loss severity and their impact on estimated cash flows. As of December 31, 2002 management used a 0.30% to 6.20% constant default rate estimate with a 10% to 130% loss severity estimate, which resulted in loss estimates of 0.03% to 8.10%. These estimates are based on historical loss data for comparable mortgages. Management estimates mortgage prepayments by evaluating historical prepayment performance of comparable mortgages and trends in the industry. As of December 31, 2002 management used a constant prepayment assumption of 23% to 50% to estimate the prepayment characteristics of the underlying collateral. Management determines the estimated fair value of the residuals by discounting the expected cash flows using a discount rate, which it believes is commensurate with the risks involved. As of December 31, 2002 management used a weighted average discount rate of approximately 9.85%.

Retained Interests in Securitizations

IMH uses certain assumptions and estimates in determining the fair value allocated to the retained interest at the time of initial sale and each subsequent sale in accordance with SFAS 140. These assumptions and estimates include projections for mortgage prepayment rates, constant default rates and loss severity commensurate with the risks involved and are reviewed periodically by management. Key economic assumptions used in measuring the retained interests for the periods indicated are as follows:

	At time of Securitization	At December 31, 2002	At December 31, 2001
Prepayment speed	16-50%	25-50%	18-50%
Loss severity	10-100%	10-130%	10-130%
Constant default rate	0.01-1%	1-6.20%	0.30-6.20%

The following table presents key economic assumptions and the sensitivity of the current fair value of residual cash flows on investment securities available-for-sale to immediate 10% and 20% adverse changes in those assumptions for the periods indicated (dollars in thousands):

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

	 At December 31,		
	 2002	2001	
Carrying amount of investment securities	\$ 21,591 \$	22,297	
Estimated fair value of investment securities	21,591	22,297	
Prepayment Speed Assumptions:	25-50%	15.20-60%	
Decline in fair value of 10% adverse change	\$ (491) \$	(812)	
Decline in fair value of 20% adverse change	(973)	(1,494)	
Loss Severity Assumptions:	10-130%	11-156%	
Decline in fair value of 10% adverse change	\$ (34) \$	(46)	
Decline in fair value of 20% adverse change	(68)	(100)	
Constant Default Rate Assumptions:	1-6.20%	0.33-7.44%	
Decline in fair value of 10% adverse change	\$ (541) \$	(706)	
Decline in fair value of 20% adverse change	(1,079)	(926)	

These sensitivities are hypothetical and should be used with caution. As the figures indicate, changes in fair value based on a 10% and 20% variation in assumptions generally cannot be extrapolated because the relationship of the change in the assumption to the change in fair value may not be linear. Also, the effect of a variation in a particular assumption on the fair value of the retained interest is calculated without changing any other assumption; in reality, changes in one factor may result in changes in another, for example, increases in market interest rates may result in lower prepayments and increased credit losses, which might magnify or counteract the sensitivities.

Quantitative information about delinquencies, credit losses and cash flows related to retained interests for the indicated periods are presented as follows (dollars in thousands):

	 At December 31,		
	 2002	2001	
Net credit losses	\$ 994 \$	586	
Actual losses to date	3.13%	3.01%	
Projected losses over life	1.09%	1.14%	
Cash flows received on retained interests	\$ 4,636 \$	2,224	

Note C—Mortgages Held-for-Investment

Mortgages held-for-investment include various types of adjustable and fixed rate loans secured by mortgages on single-family residential and multi-family real estate properties. During the years ended December 31, 2002 and 2001 IMH purchased \$3.9 billion and \$1.5 billion, respectively, of mortgages from IFC. As of December 31, 2002 and 2001 approximately 45% and 28%, respectively, of mortgages held-for-investment were collateralized by properties located in California. Mortgages held-for-investment for the periods indicated consisted of the following:

	 At December 31,		
	 2002	2001	
	(in thousand	ls)	
Adjustable rate mortgages secured by single-family residential real estate	\$ 15,626 \$	6,875	
Adjustable rate mortgages secured by multi-family residential real estate	13,392	—	
Fixed rate mortgages secured by second trust deeds on single-family residential real estate	27,880	20,211	
Unamortized net premiums (discounts) and deferred costs on mortgages	638	(8)	
	 ·		
Total mortgages held-for-investment	\$ 57,536 \$	20,078	

As of December 31, 2002 and 2001 there were \$6.0 million and \$9.3 million, respectively, of mortgages held-for-investment, which were not accruing interest due to the delinquent nature of the mortgages. If interest on such loans had

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

been accrued for the years ended December 31, 2002, 2001 and 2000, interest income would have increased by \$328,000, \$563,000 and \$1.0 million, respectively.

Note D—CMO Collateral

IMH Assets earns the net interest spread between the interest income on the mortgages securing CMOs and the interest and other expenses associated with CMO financing. Interest income is reduced by the amortization of loan premiums while interest expense includes the amortization of securitization costs, which are incurred in obtaining CMO financing. The net interest spread on CMOs may be directly impacted by mortgage prepayment levels and, to the extent each CMO class has variable rates of interest, by changes in short-term interest rates. CMO collateral includes various types of adjustable and fixed rate mortgages secured by single-family residential and multi-family real estate properties and, to a lesser extent, fixed rate second trust deeds secured by single-family residential real estate properties. CMO collateral for the periods indicated consisted of the following:

	 At December 31,		
	2002	2001	
	(in thousand	s)	
Adjustable and fixed rate mortgages secured by single-family residential real estate	\$ 5,054,486 \$	2,161,645	
Adjustable rate mortgages secured by multi-family residential real estate	12,403	—	
Fixed rate second trust deeds secured by single-family residential real estate	15,319	24,916	
Unamortized net premiums on mortgages	75,987	35,397	
Capitalized securitization costs	21,816	11,582	
Fair value of derivative instruments	(30,331)	(4,372)	
Total CMO collateral	\$ 5,149,680 \$	2,229,168	

The following table presents selected information about CMO collateral for the periods indicated:

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

		At December 31,			
	2002	2001	2000		
Percent Alt-A mortgages	99	95	90		
Percent ARMs	85	87	65		
Percent hybrid ARMs	35	68	65		
Weighted average months until hybrid ARMs interest rate adjustment	14	15	16		
Weighted average coupon	6.57	7.92	9.33		
Weighted average margin	3.01	3.42	4.15		
Weighted average original LTV	82	83	85		
Weighted average original credit score	683	678	675		
Percent with active prepayment penalty	76	54	36		
Weighted average months until prepayment penalty expiration	24	22	27		
Prior 12-month prepayment rate	25	34	25		
Lifetime prepayment rate	33	34	31		
Percent of mortgages in California	63	63	53		
Percent purchase transactions	62	70	74		
Percent owner occupied	93	94	96		
Percent first lien	99	99	95		

Note E—Finance Receivables

The terms of IWLG's affiliated warehouse lines are based on Bank of America's prime rate, which was 4.25% and 4.75% as of December 31, 2002 and 2001, respectively, with advance rates between 75% and 99% of the fair value of the mortgages outstanding. The terms of IWLG's non-affiliated warehouse lines, including the maximum warehouse line amount and interest rate, are determined based upon the financial strength, historical performance and other qualifications of the borrower. The warehouse lines have maturities that range from on-demand to one year and are generally collateralized by mortgages on single-family residential real estate. As of December 31, 2002 and 2001 IWLG had \$1.5 billion and \$1.7 billion, respectively, of warehouse lines of credit available to 64 and 57 borrowers, respectively, of which \$1.1 billion and \$466.6 million, respectively, was outstanding. IWLG finances its warehouse lending operations through reverse repurchase agreements and equity. Finance receivables, net of pledge balances, for the periods indicated consisted of the following:

 At December 31,		
 2002	2001	
(in thousands)	
\$ 664,021 \$	300,571	
363,033	106,940	
87,455	34,907	
25,739	24,231	
\$ 1,140,248 \$	466,649	
	2002 (in thousands \$ 664,021 \$ 363,033 87,455 25,739	

Note F—Allowance for loan losses

Activity for allowance for loan losses for the periods indicated was as follows:

		For the year ended December 31,				
		2002 2001		2002 2001		2000
		(in	thousands)			
Beginning balance	\$	11,692 \$	5,090 \$	4,029		
Provision for loan losses		19,848	16,813	18,839		
Charge-offs, net of recoveries		(4,938)	(9,339)	(16,496)		
Loss on sale of delinquent loans		—	(872)	(1,282)		
		·	·			
Ending balance	\$	26,602 \$	11,692 \$	5,090		

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note G—Reverse repurchase agreements

IMH and IWLG enter into reverse repurchase agreements with major brokerage firms to finance its warehouse lending operations and to fund the purchase of mortgages and mortgage-backed securities. Mortgages and mortgage-backed securities underlying reverse repurchase agreements are delivered to dealers that arrange the transactions. Reverse repurchase agreements are uncommitted lines, which may be withdrawn at any time by the lender, with interest rates that range from one-month LIBOR plus 0.85% to 2.00% or Eurodollar rates plus 75 to 150 basis points, depending on the type of collateral provided.

A reverse repurchase agreement, although structured as a sale and repurchase obligation, acts as a financing vehicle under which IMH and IWLG effectively pledges mortgages and mortgage-backed securities as collateral to secure a short-term loan. Generally, the other party to the agreement makes the loan in an amount equal to a percentage of the market value of the pledged collateral. At the maturity of the reverse repurchase agreement, IMH is required to repay the loan and correspondingly it receives its collateral. Under reverse repurchase agreements, IMH retains the beneficial ownership, including the right to distributions on the collateral and the right to vote on matters as to which certificate holders vote. Upon payment default, the lending party may liquidate the collateral. IMH's borrowing agreements require it to pledge cash, additional mortgages or additional securities backed by mortgages in the event the market value of existing collateral declines. IMH may be required to sell assets to reduce its borrowings to the extent that cash reserves are insufficient to cover such deficiencies in collateral. Interest expense on reverse repurchase agreements to fund the purchase of mortgages for the years ended December 31, 2002, 2001 and 2000 was \$23.6 million, \$30.8 million and \$39.2 million, respectively. The following tables present information regarding reverse repurchase agreements for the periods indicated (dollars in thousands):

	Type of Collateral	Committed]	Reverse Repurchase Liability	Underlying Collateral	Maturity Date
As of December 31, 2001	Mortgages	No	\$	469,491 \$	489,113	N/A
As of December 31, 2002	Mortgages	No	\$	1,168,029 \$	1,207,745	N/A

As of December 31, 2002 and 2001 reverse repurchase agreements includes accrued interest payable of \$1.5 million and \$488,000, respectively. The following table presents certain information on reverse repurchase agreements for the periods indicated:

	 For the year ended December 31,		
	 2002	2001	
	(dollars in the	ousands)	
Maximum month-end outstanding balance during period	\$ 1,168,029 \$	\$ 663,306	
Average balance outstanding for period	814,248	580,605	
Weighted average rate for period	2.90%	5.31%	

Note H—CMO Borrowings

Interest rates on adjustable rate CMOs range from a low of 0.26% over one-month LIBOR on "AAA" credit rated bonds to a high of 4.13% over one-month LIBOR on "BBB" credit rated bonds. As of December 31, 2002 interest rates on adjustable rate CMOs ranged from 0.26% to 2.75% over one-month LIBOR, or 1.64% to 4.13%. Interest rates on fixed rate CMOs range from 5.25% to 12.00%, depending on the class of CMOs issued. IMH Assets completed \$3.9 billion and \$1.5 billion in CMOs during the years ended December 31, 2002 and 2001, respectively. For the years ended December 31, 2002 interest expense on CMO borrowings was \$117.8 million, \$77.8 million and \$80.3 million, respectively.

The following table presents CMOs issued, CMOs outstanding for the periods indicated and certain interest rate information on each CMO (dollars in millions):

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

		CMOs Outstanding as of		Dange of	Range of		Range of Interest Rate
Issuance Name	Issuance Amount	12/31/02	12/31/01	Fixed Interest Rates (%)	Margins Over One-Month LIBOR (%)	Optional Redemption Date	Margins After Adjustment Date (%)
Impac CMB Trust Series 1998-1	\$ 362.8	\$ 45.8	\$ 91.7	6.65-7.25	N/A	N/A	N/A
1998-2	220.2	33.6	62.2	6.70-7.25	N/A	N/A	N/A
1999-2	115.0	—	36.1	N/A	0.37	7/2006(1)	0.74
2000-1	452.0	101.3	203.8	N/A	0.32-2.40	2/2010(2)	0.64-3.60
2000-2	491.6	145.0	303.2	N/A	0.26-1.90	12/2010(2)	0.52-2.85
2001-1	357.8	197.5	325.5	N/A	0.28-2.00	6/2011(2)	0.56-3.00
2001-2	400.5	264.6	386.6	N/A	0.28-1.90	9/2011(2)	0.56-2.85
Impac CMB Trust Series 2001-3	397.0	300.5	394.8	N/A	0.37-2.00	11/2011(2)	0.74-3.00
Impac CMB Trust Series 2001-4	345.6	269.4	345.6	N/A	0.42-2.30	1/2012(2)	0.84-3.45
Impac CMB Trust Series 2002-1	495.0	425.2	_	N/A	0.32-2.25	3/2012(2)	0.64-3.38
Impac CMB Trust Series 2002-2	496.3	444.8	_	N/A	0.28-2.10	5/2012(2)	0.56-3.15
2002-3	745.5	692.5	_	N/A	0.27-2.20	7/2012(2)	0.54-3.30
Impac CMB Trust Series 2002-4F	199.0	194.1		6.00-11.75	N/A	N/A	N/A
Impac CMB Trust Series 2002-5	497.5	483.4		N/A	0.37-2.35	9/2012(1)	0.74-3.53
Impac CMB Trust Series 2002-6F	199.0	200.3	_	5.50-11.75	N/A	N/A	N/A
Impac CMB Trust Series 2002-7	597.0	590.9		N/A	0.43-2.75	11/2012(1)	0.86-4.13
Impac CMB Trust Series 2002-8	447.8	446.2	_	N/A	0.46-2.45	12/2012(1)	0.92-3.68
Impac CMB Trust Series 2002-9F	199.0	202.0	_	5.25-12.00	N/A	N/A	N/A
	7,018.6	5,037.1	2,149.5				
Accrued interest		4.7	1.9				
	\$ 7,018.6	\$ 5,041.8	\$ 2,151.4				
	Impac CMB Trust Series 1998-1Impac CMB Trust Series 1998-2Impac CMB Trust Series 2000-1Impac CMB Trust Series 2000-2Impac CMB Trust Series 2000-2Impac CMB Trust Series 2001-1Impac CMB Trust Series 2001-2Impac CMB Trust Series 2001-3Impac CMB Trust Series 2001-3Impac CMB Trust Series 2001-4Impac CMB Trust Series 2001-4Impac CMB Trust Series 2002-1Impac CMB Trust Series 2002-1Impac CMB Trust Series 2002-2Impac CMB Trust Series 2002-3Impac CMB Trust Series 2002-3Impac CMB Trust Series 2002-5Impac CMB Trust Series 2002-6FImpac CMB Trust Series 2002-7Impac CMB Trust Series 2002-7Impac CMB Trust Series 2002-7Impac CMB Trust Series 2002-8Impac CMB Trust Series 2002-9F	Issuance Name Amount Impac CMB Trust Series 1998-1 362.8 Impac CMB Trust Series 1998-2 220.2 Impac CMB Trust Series 1999-2 220.2 Impac CMB Trust Series 2000-1 452.0 Impac CMB Trust Series 2000-1 452.0 Impac CMB Trust Series 2000-2 491.6 Impac CMB Trust Series 2001-1 357.8 Impac CMB Trust Series 2001-2 400.5 Impac CMB Trust Series 2001-3 397.0 Impac CMB Trust Series 2001-4 345.6 Impac CMB Trust Series 2002-1 495.0 Impac CMB Trust Series 2002-1 495.0 Impac CMB Trust Series 2002-1 495.0 Impac CMB Trust Series 2002-2 496.3 Impac CMB Trust Series 2002-3 745.5 Impac CMB Trust Series 2002-4F 199.0 Impac CMB Trust Series 2002-5 497.5 Impac CMB Trust Series 2002-6F 199.0 Impac CMB Trust Series 2002-8 447.8 2002-9F 199.0 Impac CMB Trust Series 2002-9F 199.0 Impac CMB Trust Series 2002-9F 199.0	Issuance Name Issuance	Issuance Name Issuance	Impac CMB Trust Series 1999-1 S 362.8 \$ 45.8 \$ 91.7 Range of Fixe of Interest Rates (%) Impac CMB Trust Series 1999-2 \$ 362.8 \$ 45.8 \$ 91.7 6.655.7.25 Impac CMB Trust Series 1999-2 220.2 33.6 62.2 6.70-7.25 Impac CMB Trust Series 2000-1 215.0 36.1 N/A Impac CMB Trust Series 2000-2 491.6 145.0 303.2 N/A Impac CMB Trust Series 2000-2 491.6 145.0 303.2 N/A Impac CMB Trust Series 2001-3 397.0 300.5 394.8 N/A Impac CMB Trust Series 2001-3 397.0 300.5 394.8 N/A Impac CMB Trust Series 2001-4 345.6 269.4 345.6 N/A Impac CMB Trust Series 2001-3 397.0 300.5 394.8 N/A Impac CMB Trust Series 2002-1 495.0 425.2 N/A Impac CMB Trust Series 2002-1 496.3 444.8 N/A Impac	Impac CMB Trust Series 1998-1 S 362.8 S 45.8 S 91.7 Range of Interest Rate Marginest Series 1998-1 Range of Interest Rate Marginest Series 1998-1 Range of Interest Rate Marginest Series 1998-2 Range of Interest Rate Marginest Series 1998-2 Range of Interest Rate Marginest Series 2002 Range of Interest Series 2003 Range of Interest Series 2003 Range of Interest Series 2003 Range of Interest Series 2004 Range of Interest Series 2003 Range of Interest Series 2004 Range of Interest Series 2005 Range of Interest Series 2004 Range of Interest Series 2005 Range of Interest Series 2005 <thrange of<br="">Interest Series 2005 Range of In</thrange>	$\begin{tabular}{ c c c c c } \hline lic c c c c c c c c c c c c c c c c c c $

(1) The optional redemption date is determined upon the earlier of (a) the payment date on which the sum of the aggregate principal balance of the bonds is less than or equal to 25% of the original bond balance at cut-off date or (b) the date indicated.

(2) The optional redemption date is determined upon the earlier of (a) the payment date on which the sum of the aggregate principal balance of the bonds is less than or equal to 20% of the original bond balance at cut-off date or (b) the date indicated.

Certain CMOs are guaranteed to certificate holders by a mortgage loan insurer, which give the CMOs the highest rating established by a nationally recognized rating agency. Each issue of CMOs is fully payable from the principal and

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

interest payments on the underlying mortgages collateralizing such debt, any cash or other collateral required to be pledged as a condition to receiving the desired rating on the debt and any investment income on such collateral. CMOs typically are structured as adjustable rate securities that are indexed to one-month LIBOR and fixed rate securities with interest on adjustable and fixed rate CMOs payable to certificate holders monthly.

Note I—Borrowings Secured by Investment Securities Available-for-Sale

In October 1999 IMH completed a re-securitization of a portion of its investment securities available-for-sale portfolio, which raised additional cash liquidity for IMH of approximately \$23.4 million, at the time of execution, after repaying reverse repurchase agreements collateralized by the investment securities available-for-sale. As of December 31, 2002 and 2001 there was \$7.1 million and \$13.0 million, respectively, outstanding on the borrowings, which were secured by \$21.6 million and \$23.9 million, respectively, of investment securities available-for-sale. The outstanding borrowings are principal only notes with remaining discounts of \$1.9 million and \$3.5 million as of December 31, 2002 and 2001, respectively. The notes represent senior or subordinated interests in trust funds primarily consisting of a pool of mortgages and are non-recourse obligations. The terms of the notes are dependent on the cash flows from the underlying certificates.

Note J—Segment Reporting

Management internally reviews and analyzes its operating segments as follows:

- the long-term investment operations, conducted by IMH, IMH Assets and IMCC, invests primarily in Alt-A single-family residential and multi-family mortgages and mortgage-backed securities secured by or representing interests in such mortgages and, to a lesser extent, in second mortgages;
- the warehouse lending operations, conducted by IWLG, provides warehouse and repurchase financing to affiliated companies and to approved mortgage banks, most of which are correspondents of IFC, to finance mortgages; and
- the mortgage operations, conducted by IFC and NFS, acquires and originates Alt-A mortgages and B/C mortgages from its network of third party correspondents, mortgage brokers and retail customers.

The following table presents reporting segments as of and for the year ended December 31, 2002 (in thousands):

	_	Long-Term Investment Operations	Warehouse Lending Operations	(a) Eliminations	Consolidated
Balance Sheet Items:					
CMO collateral	\$	5,149,680 \$	— 5	5 — 4	5,149,680
Total assets		5,597,049	1,266,227	(311,503)	6,551,773
Total stockholders' equity		477,120	91,261	(264,893)	303,488
Income Statement Items:					
Interest income	\$	187,799 \$	43,017 \$	5 (4,400)\$	5 226,416
Interest expense		125,018	24,189	(4,400)	144,807
Equity in net earnings of IFC (b)				17,073	17,073
Net earnings		39,345	18,499	17,073	74,917

The following table presents reporting segments as of and for the year ended December 31, 2001 (in thousands):

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

	Long-Term Investment Operations	Warehouse Lending Operations	(a) Eliminations	Consolidated
Balance Sheet Items:				
CMO collateral	\$ 2,229,168 \$	— \$	— \$	2,229,168
Total assets	2,490,564	543,536	(179,366)	2,854,734
Total stockholders' equity	309,794	72,763	(179,192)	203,365
Income Statement Items:				
Interest income	\$ 122,502 \$	42,463 \$	(8,350) \$	156,615
Interest expense	89,485	30,877	(8,350)	112,012
Equity in net loss of IFC (b)	—	—	10,912	10,912
Net earnings	11,799	10,467	10,912	33,178

The following table presents reporting segments as of and for the year ended December 31, 2000 (in thousands):

	Long-Term Investment Operations	Warehouse Lending Operations	(a) Eliminations	Consolidated
Balance Sheet Items:				
CMO collateral	\$ 1,372,996 \$		\$ _ \$	1,372,996
Total assets	1,590,776	461,211	(153,149)	1,898,838
Total stockholders' equity	262,416	62,296	(146,272)	178,440
Income Statement Items:				
Interest income	\$ 104,591 \$	53,940	\$ (11,452) \$	147,079
Interest expense	96,325	39,223	(11,452)	124,096
Equity in net earnings of IFC (b)	—	—	(1,762)	(1,762)
Net earnings (loss)	(66,163)	13,612	(1,682)	(54,233)
Income Statement Items: Interest income Interest expense Equity in net earnings of IFC (b)	\$ 104,591 \$ 96,325 —	53,940 39,223 —	\$ (11,452) \$ (11,452) (1,762)	147,0 124,0 (1,7)

(a) Elimination of inter-segment balance sheet and income statement items.

(b) IMH accounts for the mortgage operations using the equity method as it is an unconsolidated subsidiary of IMH.

Note K—Fair Value of Financial Instruments

The estimated fair value amounts have been determined by management using available market information and appropriate valuation methodologies, however, considerable judgment is necessarily required to interpret market data to develop the estimates of fair value. Accordingly, the estimates presented are not necessarily indicative of the amounts that could be realized in a current market exchange. The use of different market assumptions and/or estimation methodologies may have a material effect on the estimated fair value amounts.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

	December 31, 2002			December 31,	, 2001
		Carrying Estimated Carrying Amount Fair Value Amount			Estimated Fair Value
			(in thousand	ls)	
Assets					
Cash and cash equivalents	\$	113,345 \$	113,345 \$	51,887 \$	51,887
Investment securities available-for-sale		26,065	26,065	32,989	32,989
CMO collateral, including allocated derivative assets		5,149,680	5,213,546	2,229,168	2,339,558
Finance receivables		1,140,248	1,140,248	466,649	466,649
Mortgages held-for-investment		57,536	56,490	20,078	12,991
Due from affiliates		14,500	14,500	14,500	14,500
Derivative assets		14,931	14,931	5,128	5,128
<u>Liabilities</u>					
CMO borrowings, excluding accrued interest	\$	5,037,026 \$	5,058,705 \$	2,149,545 \$	2,147,391
Reverse repurchase agreements, excluding accrued interest		1,168,029	1,168,029	469,003	469,003
Borrowings secured by investment securities		7,134	6,970	12,997	12,607

The fair value estimates as of December 31, 2002 and 2001 are based on pertinent information available to management as of that date. Although management is not aware of any factors that would significantly affect the estimated fair value amounts, such amounts have not been comprehensively revalued for purposes of these consolidated financial statements since those dates and, therefore, current estimates of fair value may differ significantly from the amounts presented.

The following describes the methods and assumptions used by management in estimating fair values:

Cash and Cash Equivalents

Fair value approximates carrying amounts as these instruments are demand deposits and money market mutual funds and do not present unanticipated interest rate or credit concerns.

Investment Securities Available-for-Sale

Fair value is estimated using a bond model, which incorporates certain assumptions such as prepayment, yield and losses.

CMO Collateral

Fair value is estimated based on quoted market prices from dealers and brokers for similar types of mortgages.

Finance Receivables

Fair value approximates carrying amounts due to the short-term nature of the assets and do not present unanticipated interest rate or credit concerns.

Mortgages Held-for-Investment

Fair value is estimated based on estimates of proceeds that could be received from the sale of the underlying collateral of each mortgage.

Due From/To Affiliates

Fair value approximates carrying amount because of the short-term maturity of the liabilities and does not present unanticipated interest rate or credit concerns.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

CMO Borrowings

Fair value of fixed rate borrowings is estimated based on the use of a bond model, which incorporates certain assumptions such as prepayment, yield and losses.

Reverse Repurchase Agreements

Fair value approximates carrying amounts due to the short-term nature of the liabilities and do not present unanticipated interest rate or credit concerns.

Borrowings Secured by Investment Securities Available-for-Sale

Fair value is estimated based on quoted market prices from dealers or brokers.

Derivative Assets and Liabilities

Fair value is estimated based on quoted market prices from dealers or brokers.

Short-term Commitments to Extend Credit

Fees associated with warehouse lines of credit are not collected. Accordingly, these commitments do not have an estimated fair value.

Note L-401(k) Plan

IWLG and IMCC do not have their own 401(k) plan. Therefore, employees of IWLG and IMCC participate in the 401(k) plan of IFC. Under the 401(k) plan, employees may contribute up to 25% of their salaries, pursuant to certain restrictions. IWLG and IMCC will match 50% of the first 4% of employee contributions. Additional contributions may be made at the discretion of management. IWLG's and IMCC's matching and discretionary contributions were not material for any period presented.

Note M—Related Party Transactions

Cost Allocations

IMH and IWLG pay the mortgage operations for management and operating services based upon usage which management believes is reasonable. Total cost allocations incurred by IMH and IWLG for the years ended December 31, 2002, 2001 and 2000 were \$2.5 million, \$1.7 million and \$1.5 million, respectively.

IMH and IFC entered into a premises operating sublease agreement to rent approximately 74,000 square feet of office space in Newport Beach, California, for a ten-year term, which expires in May 2008. IFC pays monthly rental expenses and allocates the cost to subsidiaries and affiliated companies on the basis of square footage occupied. Total expense allocated to IMH and IWLG for the years ended December 31, 2002, 2001, and 2000 was \$147,000, \$121, 000 and \$113,000 respectively.

Credit Arrangements

IWLG maintains a \$600.0 million uncommitted warehouse financing facility with IFC. Advances under the warehouse facility bears interest at Bank of America's prime rate minus 0.50%. As of December 31, 2002 and 2001

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

amounts outstanding on IWLG's warehouse line to the mortgage operations, excluding pledge balances, were \$491.4 million and \$174.1 million, respectively. Interest income recorded by IWLG related to the warehouse line provided to IFC for the years ended December 31, 2002, 2001 and 2000 was \$18.0 million, \$18.4 million and \$28.1 million, respectively.

During the normal course of business IMH and subsidiaries may advance or borrow funds on a short-term basis with affiliated companies. Advances to affiliates are reflected as "Due from affiliates" while borrowings are reflected as "Due to affiliates" on IMH's balance sheet. These short-term advances and borrowings bear interest at a fixed rate of 8% per annum. As of December 31, 2002 and 2001 due to affiliates and due from affiliates was none. Interest income recorded by IMH and subsidiaries related to short-term advances with affiliates for the years ended December 31, 2002, 2001 and 2000 was \$1.2 million, \$122,000 and \$90,000, respectively. Interest expense recorded by IMH and subsidiaries related to short-term borrowings with affiliates for the years ended December 31, 2002, 2001 and 2000 was \$1.6 million, \$250,000 and \$25,000, respectively.

During 1999 IMH advanced \$14.5 million in cash to IFC at an interest rate of 9.50% per annum due June 30, 2004 in exchange for an interest only note to fund the operations of IFC and other strategic opportunities deemed appropriate by IFC. At December 31, 2002 and 2001 the amount outstanding on the note was \$14.5 million. Interest income recorded by IMH related to the note for the years ended December 31, 2002, 2001 and 2000 was \$1.4 million.

Transactions with the Mortgage Operations

During the years ended December 31, 2002 and 2001 IMH purchased mortgages from the mortgage operations having a principal balance of \$3.9 billion and \$1.5 billion, respectively. The mortgages were purchased with premiums of \$71.7 million and \$24.4 million, respectively. Servicing rights on all mortgages purchased by IMH were retained by IFC.

On February 20, 2001 IFC purchased \$5.0 million of IMH's Preferred Stock from LBP, Inc. (LBPI) at cost plus accumulated dividends. On March 27, 2001 IFC purchased an additional \$5.0 million of IMH's Preferred Stock from LBPI for \$5.25 million plus accumulated dividends. In August 2001 the Preferred Stock was converted into 2,118,644 shares of IMH common stock. Subsequently, IFC sold 1.7 million shares of IMH common stock for a gain of \$3.5 million. As of December 31, 2002 IFC did not own any shares of IMH common stock

The mortgage operations acts as a servicer of mortgages acquired on a "servicing-released" basis by IMH from the mortgage operations pursuant to the terms of a Servicing Agreement, which became effective on November 20, 1995. The mortgage operations subcontracts all of its servicing obligations under such loans to independent third parties pursuant to sub-servicing agreements.

Indebtedness of Management

In connection with the exercise of stock options by certain directors and employees of the Company, IMH made loans secured by IMH stock. The loans were made for a five-year term. In January 2002 these loans were paid in full. As of December 31, 2002 and 2001 total notes receivable from common stock sales was none and \$920,000, respectively. Interest income recorded by IMH related to the loans for the years ended December 31, 2002, 2001 and 2000 was \$7,000, \$39,000 and \$50,000, respectively.

On December 10, 2001 IFC provided William B. Ashmore, President of IFC, with a \$600,000 adjustable rate mortgage to provide financing with an initial rate of 4.13%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. As of December 31, 2002 and 2001 the outstanding principal balance on the mortgage was \$587,000 and \$600,000, respectively.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

On November 15, 2001 IFC provided Richard J. Johnson, Executive Vice President and Chief Financial Officer of IMH, with a \$140,000 fixed rate mortgage to provide financing with an interest rate of 6.50%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. Subsequently, this loan has been sold to a non-affiliate.

On February 27, 2003 IFC provided William B. Ashmore, President of IFC, with a \$295,600 fixed rate mortgage to provide financing with an initial rate of 5.50%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons.

Note O—Commitments and Contingencies

IMH is a party to financial instruments with off-balance sheet risk in the normal course of business. Such instruments include short-term commitments to extend credit to borrowers under warehouse lines of credit, which involve elements of credit risk, lease commitments, and exposure to credit loss in the event of nonperformance by the counter-parties to the various agreements associated with loan purchases. Unless noted otherwise, IMH does not require collateral or other security to support such commitments. Management uses the same credit policies in making commitments and conditional obligations as it does for on-balance sheet instruments.

Short-Term Loan Commitments

IWLG's warehouse lending program provides secured short-term revolving financing to small-and medium-size mortgage originators and affiliated companies to finance mortgages from the closing of the mortgages until sold to permanent investors. As of December 31, 2002 and 2001 IWLG had 64 and 57 committed lines of credit, respectively, extended in the aggregate principal amount of \$1.1 billion and \$1.7 billion, respectively.

Lease Commitments

IMH entered into a premises operating sublease agreement for approximately 74,000 square feet of office space in Newport Beach, California, which expires in May 2008. Minimum premises rental commitments under non-cancelable leases are as follows (in thousands):

Year 2003	\$ 2,026
Year 2004	1,990
Year 2005	2,035
Year 2006	2,080
Year 2007	2,124
Year 2008 and thereafter	2,143
Total lease commitments	\$ 12,398

Rent expense associated with the premises operating lease is allocated between IMH, IWLG and IFC based on square footage. IMH and IWLG's combined portion of premises rental expense for the years ended December 31, 2002, 2001 and 2000 was \$147,000, \$121,000 and \$113,000, respectively.

Loan Purchase Commitments

In the ordinary course of business, the mortgage operations is exposed to liability under representations and warranties made to purchasers and insurers of mortgages and the purchasers of servicing rights. Under certain circumstances, the mortgage operations is required to repurchase mortgages if there had been a breach of representations or warranties. IMH has guaranteed the performance obligation of the mortgage operations under such

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

representation and warranties related to mortgages included in securitizations. However, IMH does not anticipate nonperformance by such borrowers or counterparties.

Legal Proceedings

On September 1, 2000, a complaint captioned Michael P. and Shellie Gilmor v. Preferred Credit Corporation and Impac Funding Corporation, et al. was filed in the Circuit Court for Clay County, Missouri, Case No. CV100-4263-CC, as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loans Act and Merchandising Practices Act. In July 2001, the Missouri complaint was amended to include IMH and other Impac-related entities. A plaintiffs class was certified on January 2, 2003. Plaintiffs have also filed a motion to certify a defendant class, which motion will be held in abeyance pending resolution by the Missouri Supreme Court of certain issues relating to the applicable statute of limitations. On August 2, 2001, a complaint captioned Frazier, et al. v. Preferred Credit, et al., was filed in the Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004762-01. This case was subsequently removed to federal court, Case No. 01-2714 (W.D. Tenn). This was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration of the ruling. Plaintiffs' motion was denied in its entirety by order dated March 7, 2003. This case is currently dismissed as to all defendants. On August 8, 2001, a complaint captioned Mattie L. Street v. PSB Lending Corp., et al., was filed in the Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004888-01. This case was subsequently removed to federal court, Case No. 01-2751 (W.D. Tenn.). The Street action was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration. Plaintiffs' motion was denied in its entirety by order dated March 6, 2003. This case is currently dismissed as to all defendants. On October 2, 2001, a complaint captioned Deborah Searcy, Shirley Walker, et al. vs. Impac Funding Corporation, Impac Mortgage Holdings, Inc. et al. was filed in the Wayne County Circuit Court, State of Michigan, as a purported class action lawsuit alleging that the defendants violated Michigan's Secondary Mortgage Loan Act, Credit Reform Act and Consumer Protection Act. In February 2003, the Court granted a motion to accept for filing an amended complaint adding certain defendants, including an Impac-related entity, and dropping others, including certain Impac-related entities. Service of the amended complaint has not yet occurred. On October 10, 2001, a complaint captioned Hayes v Impac Funding Corporation, et al. was filed in the Circuit Court of Vanderburgh County, Indiana as Case No. 82C01-0110-CP580. This was stated as a purported class action lawsuit alleging a violation of the Indiana Uniform Consumer Credit Code when the loans were originated. This matter was dismissed and the plaintiffs have noticed an appeal to the Indiana Court of Appeals. This appeal remains pending. On November 30, 2001, a complaint captioned Garry Lee Skinner and Judy Cooper Skinner, et al. v. Preferred Credit, et al. was filed in the Superior Court of Durham County, North Carolina as Case No. 1CV-05596. This is stated as a purported class action alleging a violation of the North Carolina Interest Statutes and Unfair and Deceptive Trade Practices Act when the secondary mortgage loans were originated by the defendants. A motion to dismiss the complaint has been filed. This motion remains pending.

All of the purported class action lawsuits are similar in nature in that they allege that the mortgage loan originators violated the respective state's statutes by charging excessive fees and costs when making second mortgage loans on residential real estate. The complaints allege that IFC was a purchaser, and is a holder, along with other Impac-related entities, of second mortgage loans originated by other lenders. The plaintiffs in the lawsuits are seeking damages that include disgorgement, restitution, rescission, actual damages, statutory damages, exemplary damages, pre-judgement interest and punitive damages. No material amount of damages are specified in the complaints.

The Company believes that it has meritorious defenses to the above claims and intends to defend these claims vigorously. Nevertheless, litigation is uncertain and the Company may not prevail in the lawsuits and can express no opinion as to its ultimate outcome.

The Company is a party to other litigation and claims, which are normal in the course of its operations. While the results of such other litigation and claims cannot be predicted with certainty, the Company believes the final outcome of such other matters will not have a material adverse effect on the Company.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note P—Derivative Instruments and Hedging Activities

The long-term investment operations primarily acquires for long-term investment six-month LIBOR ARMs and hybrid ARMs. Six-month LIBOR ARMs are generally subject to periodic and lifetime interest rate caps. This means that the interest rate of each ARM is limited to upwards or downwards movements on its periodic interest rate adjustment date, generally six months, or over the life of the mortgage. Periodic caps limit the maximum interest rate change, which can occur on any interest rate change date to generally a maximum of 1% per semiannual adjustment. Also, each ARM has a maximum lifetime interest rate cap. Generally, borrowings are not subject to the same periodic or lifetime interest rate limitations. During a period of rapidly increasing or decreasing interest rates, financing costs would increase or decrease at a faster rate than the periodic interest rate adjustments on mortgages would allow, which could effect net interest income. In addition, if market rates were to exceed the maximum interest rate periods generally ranging from two to three years, and to a lesser extent five years, which subsequently convert to six-month LIBOR ARMs. During a rapidly increasing or decreasing interest rate adjustment financing costs could increase or decrease more rapidly than would interest rates on mortgages, which would remain fixed until their next interest rate adjustment date.

The mortgage loan investment portfolio is also subject to basis risk as the basis, or interest rate index, of mortgages and borrowings are tied to different interest indices. For instance, six-month LIBOR ARMs are indexed to six-month LIBOR while CMO and reverse repurchase borrowings are indexed to one-month LIBOR. Therefore, interest income would be received based on six-month LIBOR plus a spread but interest expense would be paid based on one-month LIBOR plus a spread. These indices typically move in unison but may vary in unequal amounts.

As mortgages are acquired from the mortgage operations, IMH purchases Caps at interest rate strike prices that are tied to a forward yield curve to hedge forecasted cash flows on CMO borrowings. IMH purchases Caps at strike prices over the one-month LIBOR forward yield curve to correspond to forecasted CMO borrowing costs, which are indexed to one-month LIBOR. A Cap allows the yield on an ARM to rise in an increasing interest rate environment just as the cost of related CMO borrowings would rise as we receive cash payments if one-month LIBOR were to reach the contractual strike price of the Cap. The Cap would provide protection for (1) periodic and lifetime cap limitations on the mortgage that are absent on the related borrowings, and (2) mismatched interest rate adjustment periods. Simultaneously with the purchase of Caps, IMH may also sell Floors at interest rate strike prices that are below the one-month LIBOR forward yield curve. The sale of a Floor allows IMH to offset the cost of the Cap. Yields on ARMs would decrease in a decreasing interest rate environment just as the cost of related borrowings could decrease as payments would be required if one-month LIBOR were to reach the contractual strike price of the Floor. IMH also purchases Swaps whereby it receives cash payments based on one-month LIBOR and makes cash payments at a fixed rate. Swaps have the effect of fixing outstanding borrowing on a similar notional amount of Swaps and, as a result, can reduce the interest rate variability of borrowings. The purchase of derivatives protects net interest income during periods of increasing interest rates while maintaining a steady stream of cash flows if interest rates decline.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The following table summarizes the notional amount (in thousands), weighted average strike price and low and high strike prices of Caps, Floors and Swaps that were acquired to hedge the variability of interest rates and corresponding cash flows on CMO borrowings for the interest due dates indicated:

	Interest Rate Caps (1) Interest Ra					e Floors			Interest Rate Swaps			
Interest Due Date		Notional Balance	Weighted Average Strike	Low Strike	Notional Balance	Weighted Average Strike	Low Strike	High Strike	Notional Balance	Weighted Average Fixed Rate	Low Fixed Rate	High Fixed Rate
Jan 2003	\$	3,013,300	4.16%	1.90%\$	1,624,596	3.63%	2.62%	4.68%\$	935,503	2.65%	2.05%	5.18%
Feb 2003		2,929,466	4.08	1.79	1,585,277	3.56	2.66	4.50	941,384	2.64	2.05	5.18
Mar 2003		2,825,653	4.10	1.84	1,547,650	3.60	1.84	5.23	936,524	2.64	2.05	5.18
Apr 2003		2,720,175	4.34	2.03	1,502,007	3.83	1.84	5.44	930,935	2.64	2.05	5.18
May 2003		2,596,248	4.33	2.03	1,457,946	3.83	1.84	5.44	924,618	2.64	2.05	5.18
June 2003		2,484,014	4.37	2.12	1,414,711	3.83	3.32	4.80	887,907	2.56	2.05	5.18
July 2003		2,376,644	4.60	2.39	1,330,292	3.97	1.84	5.66	861,991	2.51	2.05	5.18
Aug 2003		2,239,429	4.56	2.39	1,211,940	3.93	2.09	5.40	854,410	2.51	2.05	5.18
Sep 2003		2,112,970	4.61	2.54	1,102,832	3.94	2.34	5.03	846,117	2.51	2.05	5.18
Oct 2003		2,022,564	4.81	2.88	1,036,466	4.11	2.59	5.55	837,126	2.51	2.05	5.18
Nov 2003		1,771,388	4.89	2.88	911,333	4.16	4.07	5.18	827,455	2.51	2.05	5.18
Dec 2003		1,616,238	4.99	3.01	825,602	4.22	3.09	5.59	817,118	2.51	2.05	5.18
Jan 2004		1,422,535	5.23	3.42	743,881	4.37	4.47	5.27	806,135	2.51	2.05	5.18
Feb 2004		1,290,226	5.07	3.31	702,961	4.36	4.33	5.30	794,527	2.51	2.05	5.18
Mar 2004		1,051,052	5.11	3.43	623,460	4.36	3.59	5.67	782,316	2.51	2.05	5.18
Apr 2004		717,354	5.18	3.74	355,711	4.23	3.59	5.75	769,528	2.51	2.05	5.18
May 2004		590,377	5.10	3.74	310,753	4.23	4.54	5.43	756,186	2.51	2.05	5.18
June 2004		509,406	5.12	3.83	242,025	4.40	3.59	5.77	738,558	2.50	2.05	3.70
July 2004		288,158	5.32	4.05	151,673	4.66	3.59	5.82	724,248	2.50	2.05	3.70
Aug 2004		165,490	5.71	4.05	87,774	4.63	3.59	5.55	709,469	2.50	2.05	3.70
Sep 2004		118,649	6.23	4.68	79,769	4.59	3.59	4.91	536,647	2.46	2.05	3.70
Oct 2004		113,771	6.29	4.80	75,640	4.62	3.59	5.51	504,068	2.46	2.05	3.70
Nov 2004		102,987	6.36	4.80	65,592	4.60	3.59	5.51	300,860	2.50	2.05	3.70
Dec 2004		77,605	6.42	4.87	59,003	4.61	3.59	5.55	182,480	2.70	2.10	3.70
Average Notional Balance of Derivatives	\$	16,015		\$	15,261			\$	9,873			
Weighted Average Strike			4.62%			3.95%				2.60%		

(1) The high strike price for each period indicated is 10.25%.

The strike prices of Caps and Floors are based on one-month LIBOR as that is the interest rate index that is used to set interest rates on adjustable rate CMO borrowings. As of December 31, 2002 one-month LIBOR was 1.38%. Since one-month LIBOR was below the majority of strike prices on Caps during 2002, we did not receive significant cash from these derivative instruments in 2002. Conversely, we made cash payments on many Floors during 2002 as we paid the interest rate differential between the Floor strike price and the one-month LIBOR rate, which was generally lower than Floor strike prices. We made cash payments on many Floors as we hold short positions on the majority of Floors. We originally sold Floors to offset our cost of acquiring long positions in Caps. If one-month LIBOR rises and approaches the Floor strike prices, we will pay less on the Floors.

We primarily acquired Caps and Floors during the first half of 2002 and throughout 2001 to protect net interest income during a potentially rising interest rate environment. With market volatility and uncertainty and the associated risk involved which was occurring during 2001 and that was further exacerbated by the terrorist events of September 11, 2001, we felt it was prudent, at the time, to protect net interest margins from possible extreme interest rate increases by acquiring Caps and Floors. However, because interest rates continued to decline to all-time lows during 2002, we began

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

to acquire Swaps to essentially convert our adjustable rate CMO borrowings with fixed rate derivative instruments. For instance, we receive one-month LIBOR on the Swaps, which offsets our interest expense on adjustable rate CMO borrowings, and we pay a fixed interest rate. We continued to acquire Swaps to manage interest rate risk during the first quarter of 2003.

On January 1, 2001 IMH adopted SFAS 133 and the fair market value of derivatives were reflected in financial condition and results of operations. On August 10, 2001 the Derivatives Implementation Group (DIG) of the FASB published DIG G20, which further interpreted SFAS 133. On October 1, 2001 IMH adopted the provisions of DIG G20 and net income and accumulated other comprehensive income were adjusted by the amount needed to reflect the cumulative impact of adopting the provisions of DIG G20.

The following table presents certain information related to derivatives and the related component in the financial statements as of December 31, 2002 (dollars in thousands):

	D	Fair Value of erivatives	Index	 Related Amount in OCI	Unamortized Derivative Instruments	Related Amount in Derivative Asset Account	Related Amount in CMO Collateral
Derivatives not associated with CMOs	\$	(15,515)	1 mo. LIBOR	\$ (9,693)\$	(5,822)\$	(15,515)\$	_
Cash in margin account		30,446	N/A		_	30,446	_
Derivatives associated with CMOs		(30,332)	1 mo. LIBOR	(39,464)	9,132	_	(30,332)
99% of OCI activity at IFC		—	Fannie Mae	(1,035)	—	—	_
				· · · ·			
Totals	\$	(15,401)		\$ (50,192) \$	3,310 \$	14,931 \$	(30,332)

The following table presents certain information related to derivative instruments and hedging activities as of December 31, 2001 (dollars in thousands):

	D	Fair Value of erivatives	Index		Related Amount in OCI	Unamortized Derivative Instruments	Related Amount in Derivative Asset Account	Related Amount in CMO Collateral
Derivatives not associated with CMOs	\$	(13,659)	1 mo. LIBOR	\$	(15,240) \$	1,581	\$ (13,659) \$	
Cash in margin account		18,787	N/A		_	_	18,787	—
Derivatives associated with CMOs		(4,372)	1 mo. LIBOR		(12,722)	8,350	_	(4,372)
99% of OCI activity at IFC		(158)	Fannie Mae		(90)	—	—	—
Totals	\$	598		\$	(28,052) \$	9,931	\$ 5,128 \$	(4,372)
	_			_				

Note Q—Stock Option Plans

The Stock Option Plans are administered by the board of directors (the Administrator). ISOs may be granted to the officers and key employees of the Company. NQSOs and Awards may be granted to directors, officers and key employees of the Company. The exercise price for any NQSO or ISO granted under the Stock Option Plans may not be less than 100% (or 110% in the case of ISOs granted to an employee who is deemed to own in excess of 10% of the outstanding common stock) of the fair market value of the shares of common stock at the time the NQSO or ISO is granted.

The total number of shares initially reserved and available for issuance under the 2001 Plan was 1.0 million shares. However, on the beginning of each calendar year the maximum number of shares available for issuance may increase by 3.5% of the total number of shares of stock outstanding or a lesser amount determined by the board of directors. Pursuant to this provision, in January of 2003 and 2002, the 2001 Plan increased by an additional 1.5 million and 1.1 million shares, respectively, available for grant. Unless previously terminated by the board of directors, no options or

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Awards may be granted under the 2001 Plan after March 27, 2011. The total number of shares reserved and available for issuance under the 1995 Plan is 67,390 and under the 2001 Plan is 315,319. Unless previously terminated by the board of directors, no options or awards may be granted under the 1995 Plan after August 31, 2005.

Options granted under the Stock Option Plans will become exercisable in accordance with the terms of the grant made by the Administrator. Awards will be subject to the terms and restrictions of the award made by the Administrator. The Administrator has discretionary authority to select participants from among eligible persons and to determine at the time an option or Award is granted and, in the case of options, whether it is intended to be an ISO or a NQSO, and when and in what increments shares covered by the option may be purchased. As of December 31, 2002 and 2001 options to purchase 934,990 shares and 761,163 shares, respectively, were exercisable and 382,709 shares and 225,390 shares, respectively, were reserved for future grants under the Stock Option Plans. Option transactions for the periods indicated are summarized as follows:

	For the year ended December 31,									
	2002		2001		2000					
	Number of Shares	Weighted- Average Exercise Price	Number of Shares	Weighted- Average Exercise Price	Number of Shares	Weighted- Average Exercise Price				
Options outstanding at beginning of year	1,634,501 \$	5.99	253,856 \$	4.15	674,891 \$	9.89				
Options granted.	962,750	10.88	1,571,000	6.06	112,500	3.56				
Options exercised	(150,824)	6.59	(161,109)	4.03		—				
Options forfeited/cancelled			(29,246)	4.34	(533,535)	11.28				
Options outstanding at end of year	2,446,427 \$	7.88	1,634,501 \$	5.99	253,856 \$	4.15				

The following table presents information about fixed stock options outstanding at December 31, 2002:

		S	tock Options Outstanding		Options Ex	kercisable
1	Exercise Price Range	Number Outstanding	Weighted Average Remaining Contractual Life (mos.)	Weighted Average Exercise Price (\$)	Number Exercisable	Weighted Average Exercise Price (\$)
	\$ 2.70-4.00	48,335	4.23	\$ 3.82	24,999	\$ 3.81
	4.18-4.18	652,500	8.24	4.18	652,500	4.18
	4.44-7.60	101,170	7.99	6.40	97,500	6.37
	7.68-7.68	681,672	2.56	7.68	159,991	7.68
	9.38-9.38	11,250	9.24	9.38	_	_
	9.40-9.40	11,250	9.24	9.40	—	—
	9.42-9.42	11,250	9.24	9.42	_	—
	10.54-10.54	50,000	3.33	10.54	—	—
	10.95-10.95	872,500	3.58	10.95	_	—
	11.80-11.80	6,500	2.73	11.80		—
		2,446,427	4.80	7.88	934,990	5.00

In December 2002 the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation – Transition and Disclosure" (SFAS 148), an amendment of FASB Statement No. 123. SFAS 148 amends FASB Statement No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition for a voluntary change to the fair value method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of Statement No. 123 to require prominent disclosures in both annual and interim financial statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note R—Stockholders' Equity

During 2000 the board of directors authorized the repurchase up to \$3.0 million of common stock, \$0.01 par value, in open market purchases from time to time at the discretion of management; the timing and extent of the repurchases were dependent on market conditions. During 2001 and 2000 IMH repurchased 1,015,950 shares of its common stock for \$2.3 million. All repurchased shares were canceled. During 2002 IMH did not repurchase any shares of common stock.

On October 7, 1998 the board of directors adopted a stockholder rights plan in which preferred stock purchase rights were distributed as a dividend at the rate of one right for each outstanding share of common stock. The dividend distribution was made on October 19, 1998 payable to stockholders of record on that date. The rights are attached to IMH's common stock. The rights will be exercisable and trade separately only in the event that a person or group acquires or announces the intent to acquire 10 percent or more of IMH's common stock (unless the offer to acquire the shares is approved by a majority of the board of directors who are not affiliates of the acquirer). Each right will entitle stockholders to buy one-hundredth of a share of a new series of junior participating preferred stock at an exercise price of \$30.00. If IMH is acquired in a merger or other transaction after a person has acquired 10 percent or more of IMH outstanding common stock, each right will entitle the stockholder to purchase, at the right's then-current exercise price, a number of the acquiring company's common shares having a market value of twice such price. In addition, if a person or group acquires 10 percent or more of shares of IMH's common stock having a market value of twice such price. Following the acquisition by a person of 10 percent or more of the IMH's common stock and before an acquisition of 50 percent or more of the common stock, the board of directors may exchange the rights, other than the rights owned by such person, at an exchange ratio of one share of common stock per right. Before a person or group acquires beneficial ownership of 10 percent or more of IMH's common stock, the rights are redeemable for \$0.0001 per right at the option of the board of directors. The rights will expire on October 19, 2008. The rights distribution is not taxable to stockholders. The rights are intended to enable all stockholders to realize the long-term value of their investment in the Company.

Note S—Reconciliation of Earnings (Loss) Per Share

The following table presents the computation of basic and diluted net earnings (loss) per share, as if all stock options and cumulative convertible preferred stock (Preferred Stock) was outstanding for the periods indicated (in thousands, except per share data):

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

		For t	he yea	r ended Decembe	er 31,	
		2002		2001		2000
Numerator for basic earnings per share:						
Earnings (loss) before extraordinary item and cumulative effect of change in accounting principle	\$	74,917	\$	34,801	\$	(54,233)
Extraordinary item		—		(1,006)		—
Cumulative effect of change in accounting principle				(617)		
Earnings (loss) after extraordinary item and cumulative effect change in accounting principle		74,917		33,178		(54,233)
Less: Cash dividends on cumulative convertible preferred stock				(1,575)		(3,150)
Net earnings (loss) available to common stockholders	\$	74,917	\$	31,603	\$	(57,383)
Denominator for basic earnings per share:						
Basic weighted average number of common shares outstanding during the period		40,099		23,510		21,270
Denominator for diluted earnings per share:						
Diluted weighted average number of common shares outstanding during the period		40,099		27,730		21,270
Net effect of dilutive stock options		674		222		
Diluted weighted average common shares		40,773		27,952		21,270
Net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting principle:						
Basic	\$	1.87	\$	1.41	\$	(2.70)
Diluted	\$	1.84	\$	1.25	\$	(2.70)
Net earnings (loss) per share:						
Basic	\$	1.87	\$	1.34	\$	(2.70)
Diluted	\$	1.84	\$	1.19	\$	(2.70)
	-		_			

The anti-dilutive effects of stock options outstanding as of December 31, 2002, 2001 and 2000 was 405,975, 348,475 and 669,391, respectively. The antidilutive effects of outstanding Preferred Stock as of December 31, 2002, 2001 and 2000 was none, none and 6,355,932, respectively. The dilutive effect of 6,355,932 of Preferred Stock as of December 31, 2000 was not used to calculate net earnings (loss) per share because IMH had a net loss during 2000. Terms of the Preferred Stock acquisition provided for a downward adjustment of the conversion price if, among other things, certain earnings levels were not attained by IMH through June 30, 1999. The change in the Preferred Stock conversion price during 2000 from \$4.95 to \$4.72 per share resulted in 6,355,932 in common stock equivalent shares outstanding as of December 31, 2000.

Note T—Quarterly Financial Data (unaudited)

Selected quarterly financial data for 2002 follows (in thousands, except per share data):

		For the Three Months Ended,						
	Dee	December 31, September 30, June 30,		March 31,				
Net interest income	\$	25,299	\$ 23,109	\$ 17,554 \$	15,647			
Provision for loan losses		6,546	5,361	4,234	3,707			
Non-interest income		5,546	3,977	6,407	5,652			
Non-interest expense		2,190	2,291	2,002	1,943			
Net earnings		22,109	19,434	17,725	15,649			
Net earnings per share – diluted (1)		0.50	0.47	0.45	0.44			
Dividends declared per share		0.48	0.45	0.44	0.43			

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Selected quarterly financial data for 2001 follows (in thousands, except per share data):

	For the Three Months Ended,						
	D	ecember 31,	September 30,	June 30,	Ma	rch 31,	
Net interest income	\$	13,772 \$	11,387	\$ 10,551	\$	8,893	
Provision for loan losses		6,255	2,615	3,905		4,038	
Non-interest income		6,104	4,361	4,789		2,125	
Non-interest expense, including extraordinary item and Cumulative effect of change in							
accounting principle		(1,341)	4,842	2,652		5,838	
Net earnings		14,962	8,291	8,783		1,142	
Net earnings per share – diluted (1)		0.48	0.31	0.33		0.04	
Dividends declared per share		0.44	0.25	—		_	

⁽¹⁾ Diluted earnings per share are computed independently for each of the quarters presented. Therefore, the sum of the quarterly earnings per share may not equal the total for the year.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note U—Impac Funding Corporation

The following condensed financial information summarizes the financial condition and results of operations of Impac Funding Corporation for the periods indicated:

Condensed Consolidated Balance Sheets

(in thousands)

	At December	31,
	2002	2001
ASSETS		
Cash and cash equivalents	\$ 22,773 \$	28,612
Securities available-for-sale	129	3,394
Mortgages held-for-sale	495,877	174,172
Mortgage servicing rights	8,274	8,468
Premises and equipment, net	4,948	5,333
Accrued interest receivable	430	130
Other assets	 43,155	19,693
Total assets	\$ 575,586 \$	239,802
LIABILITIES AND SHAREHOLDERS' EQUITY	 	
Borrowings from IWLG	\$ 491,383 \$	174,136
Due to affiliates	14,500	14,500
Deferred revenue	5,088	4,479
Accrued interest expense	1,068	453
Other liabilities	 42,550	26,914
Total liabilities	554,589	220,482
Shareholders' equity:	 	
Preferred stock	18,053	18,053
Common stock	182	182
Retained earnings	25,969	8,722
Cumulative dividends declared	(21,984)	(8,984)
Accumulated other comprehensive income (loss)	 (1,223)	1,347
Total shareholders' equity	20,997	19,320
Total liabilities and shareholders' equity	\$ 575,586 \$	239,802

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Condensed Consolidated Statements of Operations (in thousands)

	For th	e year ended Decemb	er 31,
	2002	2001	2000
Net interest income:			
Total interest income	\$ 30,393	\$ 24,175	\$ 28,649
Total interest expense	22,125	20,865	30,056
Net interest income (expense)	 8,268	3,310	(1,407)
Non-interest income:			
Gain on sale of loans	71,064	46,949	19,727
Loan servicing income (expense)	(1,863)	2,140	6,286
Other income	2,140	5,005	1,105
Total non-interest income	 71,341	54,094	27,118
Non-interest expense:			
General and administrative and other expense	42,048	29,257	19,634
Amortization of mortgage servicing rights	4,059	4,519	5,179
Provision for repurchases	2,806	3,498	371
Impairment of mortgage servicing rights	855	825	_
Write-down of securities available-for-sale	 —	_	1,537
Total non-interest expense	49,768	38,099	26,721
Earnings (loss) before income taxes	 29,841	19,305	(1,010)
Income taxes	12,595	8,300	770
Earnings (loss) before cumulative effect of change in accounting principle	 17,246	11,005	(1,780)
Cumulative effect of change in accounting principle	 	17	
Net earnings (loss)	\$ 17,246	\$ 11,022	\$ (1,780)

INDEPENDENT AUDITORS' REPORT

The Board of Directors Impac Funding Corporation:

We have audited the accompanying consolidated balance sheets of Impac Funding Corporation and subsidiaries as of December 31, 2002 and 2001 and the related consolidated statements of operations and comprehensive earnings (loss), changes in shareholders' equity and cash flows for each of the years in the three-year period ended December 31, 2002. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of Impac Funding Corporation and subsidiaries as of December 31, 2002 and 2001 and the results of their operations and their cash flows for each of the years in the three-year period ended December 31, 2002 in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note A to the consolidated financial statements, the Company changed its method of accounting for derivative instuments and hedging activities in 2001.

KPMG LLP

Orange County, California January 29, 2003, except as to the last paragraph to Note J to the consolidated financial statements, which is as of February 27, 2003

IMPAC FUNDING CORPORATION AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

(dollar amounts in thousands)

Securities available-for-sale 129 3,394 Mortgages held-for-sale 495,877 174,172 Mortgage servicing rights 8,274 8,468 Premises and equipment, net 4,948 5,333 Accrued interest receivable 430 130 Other assets 43,155 19,693 Total assets 575,586 \$ 239,802 Entropy of the second secon		At December 31,		er 31,
Cash and cash equivalents\$22,773\$22,612Securities available-for-sale1293,394Morrgages held-for-sale495,877174,172Morrgages held-for-sale4,9485,333Accured interest receivable43,15519,693Other assets43,15519,693Total assets\$575,586\$233,802LIABILITIES AND SHAREHOLDERS' EQUITYBorrowings from IWLG5491,383\$174,136Due to affiliates14,50014,50014,50014,500Defored revenue5,0884,479442,55026,914Accured interest expense1,068453443,5526,914Total labilities5,5489\$220,48226,914Commitments and contingencies55,589220,48226,914Shareholders' equity:7174,17218,05318,053Commitments and contingencies11,000 shares astudorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Commitments and contingencies25,9588,722182Shareholders' equity:25,9588,722182Preferred stevide dividends declared(21,944)(8,984)Accured indired declared(21,944)18,053Commuted widends declared(21,944)18,053Total isareholders' equity19,30018,053Total asterholders' equity19,30018,053Commit et widends declared(21,944)			2002	2001
Securities available-for-sale 129 3,394 Mortgages held-for-sale 495,877 174,172 Mortgage servicing rights 8,274 8,468 Premises and equipment, net 4,948 5,333 Accrued interest receivable 430 130 Other assets 431,55 19,693 Total assets 575,586 \$ 239,802 LIABILITIES AND SHAREHOLDERS' EQUITY Borrowings from IWLG \$ 491,383 \$ 174,136 Due to affiliates 14,500 14,500 Accrued interest expense 1,068 43,793 Accrued interest expense 1,068 433 Other liabilities 554,589 220,482 Commitments and contingencies Shareholders' equity: 7 7 Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 182 182 Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outsta	ASSETS			
Mortgages held-for-sale 495,877 174,172 Mortgage servicing rights 8,274 8,648 Premises and equipment, net 4,948 5,333 Accrued interest receivable 430 130 Other assets 43,155 19,693 Total assets 431,55 19,693 Due to affiliates 491,383 \$ 174,136 Due to affiliates 14,500 14,500 De to affiliates 14,500 14,500 Det to affiliates 10,668 443,550 Other liabilities 554,589 220,691 Total liabilities 554,589 220,482 Commitments and contingencies 554,589 220,482 Commitments and contingencies 18,053 18,053 Shareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Commutative dividends declared (21,984) (8,984) Accumulated other comprehensive income (loss) (1,222) 1,347	Cash and cash equivalents	\$	22,773 \$	28,612
Morgage servicing rights 8,274 8,468 Premises and equipment, net 4,948 5,333 Accrued interest receivable 43,155 19,693 Other assets 43,155 19,693 Total assets 575,586 \$ 239,802 ILABILITIES AND SHAREHOLDERS' EQUITY Borrowings from IWLG \$ 491,383 \$ 174,136 Due to affiliates 14,500 14,500 Deferred revenue 5,088 4,479 Accrued interest expense 1,068 433 Other liabilities 220,482 26,914 Total liabilities 554,589 220,482 Commitments and contingencies 554,589 220,482 Shareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Commot stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 201 182 182 Commutative dividends declared (21,954) (8,944) 45,255 Commutative dividends declared (25,968 8,722	Securities available-for-sale		129	3,394
Premises and equipment, net4,9485,333Accrued interest receivable430130Other assets43,15519,693Total assets\$ 575,568\$ 239,802ILIABILITIES AND SHAREHOLDERS' EQUITYBorrowings from IWLG\$ 491,383\$ 174,136Due to affiliates14,50014,500Deferred revenue5,0884,479Accrued interest expense1,0684453Other liabilities10,068453Other liabilities554,589220,482Commitments and contingencies554,589220,482Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Retained earnings25,5688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity19,32019,320	Mortgages held-for-sale		495,877	174,172
Accrued interest receivable430130Other assets43,15519,693Total assets\$ 575,58\$ 239,802LIABILITIES AND SHAREHOLDERS' EQUITYBorrowings from IWLG\$ 491,333\$ 174,136Due to affiliates14,50014,500Deferred revenue5,0884,479Accrued interest expense1,0684433Other liabilities26,9144533Other liabilities554,589220,482Total liabilities554,589220,482Comminments and contingencies55Shareholders' equity:18,05318,053Preferred stock, no par value; 10,000 shares issued and outstanding at December 31, 2002 and 200118218,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118218,053Commode declared25,9688,72226,914182Accumulated other comprehensive income (loss)(1,222)1,33718,053Total shareholders' equity20,99719,32019,320	Mortgage servicing rights		8,274	8,468
Other assets43,15519,693Total assets\$ 575,586\$ 239,802LIABILITIES AND SHAREHOLDERS' EQUITYBorrowings from IWLG\$ 491,383\$ 174,136Due to affiliates14,50014,500Deferred revenue5,0884,479Accrued interest expense1,068453Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies55Shareholders' equity:18,05318,053Preferred stock, no par value; 10,000 shares issued and outstanding at December 31, 2002 and 200118182Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accurulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Premises and equipment, net		4,948	5,333
Total assets S 575,586 \$ 239,802 LIABILITIES AND SHAREHOLDERS' EQUITY Borrowings from IWLG \$ 491,383 \$ 174,136 Due to affiliates 14,500 14,500 14,500 Deferred revenue 5,088 4,479 Accrued interest expense 1,068 453 Other liabilities 42,550 26,914 Total liabilities 554,589 220,482 Commitments and contingencies 5 54,589 220,482 Shareholders' equity:	Accrued interest receivable		430	130
LIABILITIES AND SHAREHOLDERS' EQUITYBorrowings from IWLG\$ 491,383\$ 174,136Due to affiliates14,50014,500Deferred revenue5,0884,479Accrued interest expense1,0684333Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies518,053Shareholders' equity:18,05318,053Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Common stock, no par value; 10,000 shares outper 31, 2002 and 200125,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Other assets		43,155	19,693
Borrowings from IWLG \$ 491,383 \$ 174,136 Due to affiliates 14,500 14,500 Deferred revenue 5,088 4,479 Accrued interest expense 1,068 4233 Other liabilities 42,550 220,482 Total liabilities 554,589 220,482 Commitments and contingencies 554,589 220,482 Shareholders' equity: 554,589 18,053 Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 182 182 Retained earnings 25,968 8,722 1,984 182 Quulative dividends declared (21,984) (8,984) 1,222 1,347 Total shareholders' equity 20,997 19,320 19,320	Total assets	\$	575,586 \$	239,802
Borrowings from IWLG \$ 491,383 \$ 174,136 Due to affiliates 14,500 14,500 Deferred revenue 5,088 4,479 Accrued interest expense 1,068 4233 Other liabilities 42,550 220,482 Total liabilities 554,589 220,482 Commitments and contingencies 554,589 220,482 Shareholders' equity: 554,589 18,053 Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 182 182 Retained earnings 25,968 8,722 1,984 182 Quulative dividends declared (21,984) (8,984) 1,222 1,347 Total shareholders' equity 20,997 19,320 19,320		-		
Due to affiliates14,50014,500Deferred revenue5,0884,479Accrued interest expense1,068453Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies52Shareholders' equity:718,05318,053Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,72225,9688,722Cumulative dividends declared(21,984)(6,984)(1,222)1,347Total shareholders' equity20,99719,32019,320				
Deferred revenue5,0884,479Accrued interest expense1,068453Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies	Borrowings from IWLG	\$		
Accrued interest expense1,068453Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies52Shareholders' equity:Freferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,053Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Due to affiliates			14,500
Other liabilities42,55026,914Total liabilities554,589220,482Commitments and contingencies554,589220,482Shareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,722182Cumulative dividends declared Accumulated other comprehensive income (loss)(21,984)(8,984)Total shareholders' equity20,99719,320	Deferred revenue			
Total liabilities554,589220,482Commitments and contingenciesShareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,722Cumulative dividends declared Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Accrued interest expense		1,068	453
Commitments and contingenciesShareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,722Cumulative dividends declared Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Other liabilities		42,550	26,914
Shareholders' equity: Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 18,053 18,053 Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001 182 182 Retained earnings 25,968 8,722 Cumulative dividends declared (21,984) (8,984) Accumulated other comprehensive income (loss) (1,222) 1,347 Total shareholders' equity 20,997 19,320	Total liabilities		554,589	220,482
Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Commitments and contingencies			
Preferred stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 200118,05318,053Common stock, no par value; 10,000 shares authorized; 10,000 shares issued and outstanding at December 31, 2002 and 2001182182Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320	Shareholders' equity:			
2001182182Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320			18,053	18,053
Retained earnings25,9688,722Cumulative dividends declared(21,984)(8,984)Accumulated other comprehensive income (loss)(1,222)1,347Total shareholders' equity20,99719,320			182	182
Accumulated other comprehensive income (loss) (1,222) 1,347 Total shareholders' equity 20,997 19,320	Retained earnings		25,968	8,722
Accumulated other comprehensive income (loss) (1,222) 1,347 Total shareholders' equity 20,997 19,320	Cumulative dividends declared		(21,984)	(8,984)
	Accumulated other comprehensive income (loss)			
Total liabilities and shareholders' equity\$ 575,586\$ 239,802	Total shareholders' equity		20,997	19,320
	Total liabilities and shareholders' equity	\$	575,586 \$	239,802

See accompanying notes to consolidated financial statements.

IMPAC FUNDING CORPORATION AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE EARNINGS (LOSS)

(in thousands)

	 For the year ended December 31,		
	 2002	2001	2000
INTEREST INCOME:			
Mortgages held-for-sale	\$ 27,486	\$ 23,044 \$	5 28,009
Other interest income	2,907	1,131	640
Total interest income	 30,393	24,175	28,649
INTEREST EXPENSE:			
Borrowings from IWLG	18,044	18,362	28,063
Other affiliated borrowings	3,722	2,010	1,604
Other non-affiliated borrowings	359	493	389
Total interest expense	22,125	20,865	30,056
Net interest income (expense)	8,268	3,310	(1,407
NON-INTEREST INCOME:	 	·	
Gain on sale of loans	71,064	46,949	19,727
Loan servicing income (expense)	(1,863)	2,140	6,286
Gain on sale of investment securities	1,708	3,710	0,280
Other income	432	1,295	1,054
Total non-interest income	 71,341	54,094	27,118
MAN INTERPORT EVARMEN.	 		
NON-INTEREST EXPENSE:	25,746	16 550	0.766
Personnel expense General and administrative and other expense		16,559	9,766 3,636
Amortization of mortgage servicing rights	7,486 4,059	3,757 4,519	5,179
Professional services	4,0 <i>3</i> 9 3,865	2,286	2,535
Provision for repurchases	2,805	3,498	2,333
Equipment expense	2,000	2,583	1,414
Occupancy expense	2,406	1,983	1,626
Data processing expense	1,770	1,743	657
Impairment of mortgage servicing rights	855	825	
Mark-to-market (gain) loss – SFAS 133	(1,700)	346	
Write-down of investment securities available-for-sale	_		1,537
Total non-interest expense	 49,768	38,099	26,721
Earnings (loss) before income taxes and cumulative effect of change in accounting principle	 29,841	19,305	(1,010
Income taxes	 12,595	8,300	770
Earnings (loss) before cumulative effect of change in accounting principle	17,246	11,005	(1,780
Cumulative effect of change in accounting principle	 _	17	_
Net earnings (loss)	17,246	11,022	(1,780
Less: Cash dividends on preferred stock	 (12,870)	(8,894)	—
Net earnings (loss) available to common shareholders	 4,376	2,128	(1,780
Other comprehensive earnings (loss):	 		
Unrealized gains (losses) on securities:			
Unrealized holding gains (losses) arising during period	(1,092)	1,361	(15
Less: Reclassification of gains (losses) included in income	 (1,477)		168
Net unrealized gains (losses) arising during period	(2,569)	1,361	153
Comprehensive earnings (loss)	\$ 14,677	\$ 12,383	6 (1,627

See accompanying notes to consolidated financial statements.

IMPAC FUNDING CORPORATION AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CHANGES IN SHAREHOLDERS' EQUITY

(dollar amounts in thousands)

	Number of Preferred Shares	Prefe Sto		Number of Common Shares	 Common Stock	Retained Earnings (Accumulated Deficit)	Cumulative Dividends Declared	Accumulated Other Comprehensive Gain(Loss)	Total Shareholders' Equity
Balance, December 31, 1999	10,000	\$	18,053	10,000	\$ 182	\$ (520) \$	5 —	\$ (167)	\$ 17,548
Net loss, 2001						(1,780)			(1,780)
Other comprehensive income	_		_	_	_	_	—	153	153
Balance, December 31, 2000	10,000		18,053	10,000	182	(2,300)	_	(14)	15,921
Net earnings, 2001					—	11,022	—	_	11,022
Securities valuation allowance	—		_		—	—	_	1,452	1,452
Other comprehensive loss					—	_	—	(91)	(91)
Dividends declared	_					_	(8,984)	—	(8,984)
Balance, December 31, 2001	10,000		18,053	10,000	182	8,722	(8,984)	1,347	19,320
Net earnings, 2002					—	17,246	—	_	17,246
Other comprehensive loss	_			_	_		_	(2,569)	(2,569)
Dividends declared	—		—		—	—	(13,000)	—	(13,000)
Balance, December 31, 2002	10,000	\$	18,053	10,000	\$ 182	\$ 25,968 \$	5 (21,984)	\$ (1,222)	\$ 20,997

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

	For the year ended December 31,			
		2002	2001	2000
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net earnings (loss)	\$	17,246 \$	11,022 \$	(1,780)
Adjustments to reconcile net earnings (loss) to net cash (used in)provided by operating activities:				
Provision for repurchases		2,806	3,498	371
Gain on sale of loans		71,064	46,949	19,727
Depreciation and amortization		6,429	7,011	6,280
Amortization of deferred revenue		(2,654)	(1,914)	(2,313)
Impairment of mortgage servicing rights		855	825	—
Net change in accrued interest receivable		(300)	910	(992)
Net change in other assets and liabilities		(10,802)	3,355	6,058
Net change in deferred taxes		2,787	5,547	(150)
Net change in deferred revenue		3,263	1,367	(296)
Gain on sale of investment securities available-for-sale		(1,708)	(3,710)	(51)
Write-down of investment securities available-for-sale		_	_	1,537
Purchase of mortgages held-for-sale		(6,044,162)	(3,203,559)	(2,112,724)
Sale of and principal reductions on mortgages held-for-sale		5,648,016	3,256,542	1,884,847
Net change in accrued interest expense		615	(1,723)	1,333
Net cash (used in) provided by operating activities		(306,545)	126,120	(198,153)
CASH FLOWS FROM INVESTING ACTIVITIES:				
Additions to master servicing rights and mortgage servicing rights		(4,720)	(2,874)	(496)
Purchase of securities available-for-sale		—	(10,250)	—
Sale of securities available-for-sale		3,164	11,965	
Purchase of premises and equipment		(1,985)	(2,788)	(2,563)
Net cash used in investing activities		(3,541)	(3,947)	(3,059)
CASH FLOWS FROM FINANCING ACTIVITIES:				
Net change in borrowings from Impac Warehouse Lending Group		317,247	(92,858)	200,869
Dividends paid		(13,000)	(8,984)	_
Net change in other borrowings				(181)
Net cash provided by (used in) financing activities		304,247	(101,842)	200,688
Net change in cash and cash equivalents		(5,839)	20,331	(524)
Cash and cash equivalents at beginning of year		28,612	8,281	8,805
Cash and cash equivalents at end of year	\$	22,773 \$	28,612 \$	8,281
SUPPLEMENTARY INFORMATION:				
Interest paid	\$	21,510 \$	22,588 \$	28,723
Taxes paid		14,983	7,010	24
Conversion of IMH preferred stock to common stock		—	10,250	—

See accompanying notes to consolidated financial statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note A—Summary of Business and Significant Accounting Policies

1. Business and Financial Statement Presentation

The mortgage operations are conducted by Impac Funding Corporation (IFC), Impac Lending Group (ILG), a division of IFC, and Novelle Financial Services, Inc. (Novelle) and Impac Secured Assets Corporation (ISAC), wholly-owned subsidiaries of IFC, collectively (the mortgage operations). The mortgage operations acquire, originate, sell and securitize primarily adjustable and fixed rate non-conforming Alt-A mortgages (Alt-A mortgages) and, to a lesser extent, sub-prime mortgages (B/C mortgages) from correspondents, mortgage brokers and retail customers. Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but have loan characteristics that make them non-conforming under those guidelines.

IFC acquires primarily adjustable and fixed rate Alt-A mortgages from its network of third party correspondents. Correspondents originate and close mortgages under IFC's mortgage programs on a flow (loan-by-loan) basis or through bulk sale commitments. Correspondents include savings and loan associations, commercial banks and mortgage bankers and brokers. IFC acts as an intermediary between the originators of mortgages that do not meet the guidelines for purchase by Fannie Mae and Freddie Mac and permanent investors in mortgage-backed securities secured by or representing an ownership interest in mortgages. ILG markets, underwrites, processes and funds mortgages for both wholesale and retail customers. ILG works directly with mortgage brokers to originate, underwrite and fund their mortgages. Many of ILG's wholesale customers cannot conduct business with the mortgage operations as correspondents because they do not meet the higher net worth requirements. Through its retail channel, ILG markets mortgages directly to the public. ILG offers the same mortgage programs to its mortgage brokers and retail customers that are offered by IFC. Novelle is a wholly-owned subsidiary of IFC that originates B/C mortgages through a network of wholesale brokers and subsequently sells its loans to third party investors for cash gains. Novelle began operations during September of 2001 in San Diego, California, where it is currently headquartered, with capital provided by IFC. ISAC is a wholly-owned specialty purpose entity through which the mortgage operations issues its Real Estate Mortgage Investment Conduit (REMICs).

The consolidated financial statements include the operations of IFC and its subsidiaries and have been prepared in conformity with accounting principles generally accepted in the United States of America and prevailing practices within the mortgage banking industry. All significant inter-company balances and transactions with IFC's consolidated subsidiaries have been eliminated in consolidation. Interest income and interest expense on affiliated short-term advances or borrowings have been earned or paid at the rate of 8% per annum. Certain amounts in the prior year's consolidated financial statements have been reclassified to conform to the current presentation. Management has made a number of estimates and assumptions relating to the reporting of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period to prepare these financial statements in conformity with accounting principles generally accepted in the United States of America. Actual results could differ from those estimates.

2. Cash and Cash Equivalents

For purposes of the consolidated statements of cash flows, cash and cash equivalents consists of cash and money market mutual funds. Investments with maturities of three months or less at date of acquisition are considered to be cash equivalents.

3. Gain on Sale of Loans

The mortgage operations recognizes gains or losses on the sale of mortgages when the sales transaction settles or upon the securitization of the mortgages when the risks of ownership have passed to the purchasing party. Gains

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

and losses may be increased or decreased by the amount of any servicing released premiums received and costs associated with the acquisition or origination of mortgages. Gain on sale of loans or securities to IMH are deferred and accreted over the estimated life of the mortgages or securities using the interest method. A transfer of financial assets in which control is surrendered is accounted for as a sale to the extent that consideration other than a beneficial interest in the transferred assets is received in the exchange. Liabilities and derivatives incurred or obtained by the transfer of financial assets are required to be measured at fair value, if practicable. Also, servicing assets and other retained interests in the transferred assets must be measured by allocating the previous carrying value between the asset sold and the interest retained, if any, based on their relative fair values at the date of transfer. To determine the value of the securities, management estimates future rates of prepayments, prepayment penalties to be received, delinquencies, defaults and default loss severity and their impact on estimated cash flows.

4. Securities Available-for-Sale and Securities Held-for-Trading

Investment and mortgage-backed securities are classified as held-to-maturity, available-for-sale, and/or trading securities. Held-to-maturity investment and mortgage-backed securities are reported at amortized cost, available-for-sale securities are reported at fair value with unrealized gains and losses, net of related income taxes, as a separate component of shareholders' equity, and trading securities are reported at fair value with unrealized gains and losses reported in operations. Investment securities are held as available-for-sale reported at fair value with unrealized gains and losses reported as a separate component of shareholders' equity. Premiums or discounts obtained on investment securities are accreted or amortized to interest income over the estimated life of the investment securities using the interest method.

Residual interests in securitization of mortgages are recorded as a result of the sale of mortgages through securitizations. The mortgage operations sell a portfolio of mortgages to ISAC that has been established for the limited purpose of buying and reselling mortgages. ISAC then transfers the same mortgages to a REMIC or owners trust (the Trust). The Trust issues interest-bearing asset-backed securities in an amount equal to the aggregate principal balance of the mortgage operations typically sell these certificates at face value and without recourse except for representations and warranties customary to the mortgage banking industry. Credit enhancement may be provided for the benefit of the investors in the form of additional collateral (over-collateralization) held by the Trust. The over-collateralization account is required to be maintained at specified levels.

At the closing of each securitization, mortgages held-for-sale are removed from the consolidated balance sheets and cash received and the estimated fair value of the portion of the mortgages retained from the securitizations (Residuals) is added to the consolidated balance sheet. The Residuals consist of the over-collateralization account and the net interest receivables, which represent the estimated cash flows to be received by the Trust in the future. The excess of the cash received and the assets retained over the carrying value of the mortgages sold, less transaction costs, equals the net gain on sale of loans recorded by the mortgage operations.

The basis in the mortgages is allocated between the portion of the mortgages sold through the certificates and the portion retained based on the relative fair values of those portions on the date of the sale. Gains or losses attributable to the changes in the fair value of the residuals, which are recorded at estimated fair value and accounted for as held-for-trading securities, may be recognized. The market for the purchase or sale of residuals is not considered liquid. Management determines the estimated fair value of the residuals by discounting the expected cash flows using a discount rate, which management believes is commensurate with the risks involved. Some of the residual interests generated by the mortgage operations are sold to IMH and accounted for as available-for-sale securities at IMH, although, the mortgage operations has not sold any mortgage-backed securities to IMH since 1999.

The mortgage operations receive periodic servicing fees for the servicing and collection of the mortgages as master servicer of the securitized mortgages. The mortgage operations is also entitled to the cash flows from the residual that represent collections on the mortgages in excess of the amounts required to pay the certificate

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

principal and interest, the servicing fees and certain other fees such as trustee and custodial fees. At the end of each collection period, cash collected from the mortgages are allocated to the base servicing and other fees for the period, then to the certificate holders for interest at the pass-through rate on the certificates plus principal as defined in the servicing agreements. If the amount of cash required for the above allocations exceeds the amount collected during the collection period, the shortfall is drawn from the over-collateralization account. If the cash collected during the period exceeds the amount necessary for the above allocations, and there is no shortfall in the related over-collateralization account, the excess is released to the mortgage operations. If the over-collateralization account balance is not at the required credit enhancement level, the excess cash collected is retained in the over-collateralization account is restricted from use by the mortgage operations. Pursuant to certain servicing agreements, cash held in the over-collateralization accounts may be used to make accelerated principal pay downs on the certificates to create additional excess collateral in the over-collateralization account.

5. Mortgages Held-for-Sale

Mortgages held-for-sale are stated at the lower of cost or market in the aggregate as determined by outstanding commitments from investors or current investor yield requirements. Interest is recognized as revenue when earned according to the terms of the mortgages and when, in the opinion of management, it is collectible. Nonrefundable fees and direct costs associated with the origination or purchase of mortgages are deferred and recognized when the mortgages are sold as gain or loss on sale of loans.

6. Premises and Equipment

Premises and equipment are stated at cost, less accumulated depreciation or amortization. Depreciation on premises and equipment is recorded using the straight-line method over the estimated useful lives of individual assets, typically, three to seven years.

7. Master Servicing Rights and Mortgage Servicing Rights

Master Servicing Rights. Master servicing rights on mortgages sold are generally retained. Master servicing fees receivable represent the present value of the difference between the interest rate on mortgages purchased or acquired and the interest rate received by investors who purchase the securities backed by such loans, in excess of the normal loan servicing fees charged by either the mortgage operations on loans acquired "servicing released" or correspondents who sold mortgages to the mortgage operations with "servicing retained." Master servicing fees receivable have characteristics similar to interest-only securities. Accordingly, master servicing fees receivable have many of the same risks inherent in interest-only securities, including the risk that they will lose a substantial portion of their value as a result of rapid prepayments occasioned by declining interest rates. Master servicing fees receivable are retained for investment. Accordingly, if the mortgage operations had to sell these receivables, the value received may or may not be at or above the values at which the mortgage operations carried them on the balance sheet. In determining present value of future cash flows, management uses a market discount rate. Prepayment assumptions are based on recent evaluations of the actual prepayments of the mortgage operations servicing portfolio or on market prepayment rates on new portfolios and the interest rate environment at the time the master servicing fees receivable are created. As of December 31, 2002 management used estimates for mortgage prepayment rates ranging from 20% to 37%.

Mortgage Servicing Rights. A portion of the cost of acquiring a mortgage is allocated to mortgage servicing rights (MSRs) based on its fair value relative to the components of the mortgage. Gain or loss on the sale of MSRs is recognized when the sales contract has been executed and ownership is determined to have passed to the purchasing party. Gains and losses are computed by deducting the basis in the MSRs and any other costs associated with the sale from the purchase price. To determine the fair value of the servicing rights created, management uses a valuation model that calculates the present value of future net servicing revenues to determine the fair value of the servicing rights. In using this valuation method, management incorporates assumptions that it

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

believes market participants would use in estimating future net servicing, an inflation rate, ancillary income per loan, a prepayment rate, a default rate and a discount rate commensurate with the risk involved. MSRs are amortized in proportion to, and over the period of expected net servicing income. The MSRs are considered impaired when the fair value using a discounted cash flow analysis is less than the carrying value. In that event, an impairment loss is recognized in the respective period. Management uses certain assumptions and estimates in determining the fair value allocated to the retained interest at the time of initial sale and each subsequent sale. As of December 31, 2002 management used estimates for mortgage prepayment rates ranging from 20% to 37%.

8. Income Taxes

Deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax base. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that includes the enactment date.

9. Derivative Instruments and Hedging Activities

In June 1998 the Financial Accounting Standards Board issued Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities," as amended by SFAS No. 137 and SFAS No. 138, collectively, (SFAS 133). SFAS 133 establishes accounting and reporting standards for derivative instruments, including a number of derivative instruments embedded in other contracts, collectively referred to as derivatives, and for hedging activities. It requires that an entity recognize all derivatives as either assets or liabilities in the statement of financial position and measure those instruments at fair value. If specific conditions are met, a derivative may be specifically designated as (1) a hedge of the exposure to changes in the fair value of a recognized asset or liability or an unrecognized firm commitment; (2) a hedge of the exposure to variable cash flows of a forecasted transaction; or (3) a hedge of the foreign currency exposure of a net investment in a foreign operation, an unrecognized firm commitment, an available for sale security or a foreign-currencydenominated forecasted transaction. Under SFAS 133, an entity that elects to apply hedge accounting is required to establish at the inception of the hedge the method it will use for assessing the effectiveness of the hedging derivative and the measurement and approach for determining the ineffective aspect of the hedge. Those methods must be consistent with the entity's approach to managing risk. This statement is effective for all fiscal quarters of fiscal years beginning after June 15, 2000.

On January 1, 2001 the mortgage operations adopted SFAS 133, and at that time, designated the derivatives in accordance with the requirements of the new standard. These cash flow derivatives hedge the variability of forecasted cash flows attributable to interest rate risk. Cash flow hedges are accounted for by recording the value of the derivative in the financial statements as either an asset or liability with a corresponding offset recorded in accumulated other comprehensive income within stockholders' equity, net of tax. Amounts are reclassified from accumulated other comprehensive income to the income statement in the period the hedged cash flow occurs. Derivative gains and losses not considered effective in hedging the change in expected cash flows of the hedged item are recognized immediately in the income statement. With the implementation of SFAS 133 IFC recorded pre-tax transition amounts associated with establishing the fair values of the derivatives and hedged items outstanding as of December 31, 2000. The effect of pre-tax transition adjustments on the balance sheet and statement of operations as of December 31, 2000 in compliance with SFAS 133 are as follows (in thousands):

Balance Sheet Adjustments:	
Other assets	\$ 260
Other liabilities	1,191
Accumulated other comprehensive loss	(914)
Statement of Operations Adjustment:	
Net earnings	\$ (17)

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

On August 10, 2001 the Derivatives Implementation Group (DIG) of the FASB published DIG G20 which further interpreted SFAS 133. On October 1, 2001 the mortgage operations adopted the provisions of DIG G20 and net income and accumulated other comprehensive income were adjusted by the amount needed to reflect the cumulative impact of adopting the provisions of DIG G20.

The mortgage operations hedges interest rate risk and price volatility on its mortgage loan investment portfolio during the time it commits to acquire or originate mortgages at a pre-determined rate and the time it sells or securitizes mortgages. To mitigate interest rate and price volatility risks, the mortgage operations enters into forward commitments and futures transactions. The nature and quantity of hedging transactions are determined based on various factors, including market conditions and the expected volume of mortgage acquisitions and originations.

10. Servicing Income

Servicing income is reported as earned principally on a cash basis when the majority of the service process is completed.

11. Recent Accounting Pronouncements

In June 2001 FASB issued SFAS No. 143, "Accounting for Asset Retirement Obligations" (SFAS No. 143) requires the Company to record the fair value of an asset retirement obligation as a liability in the period in which it incurs a legal obligation associated with the retirement of tangible long-lived assets that result from the acquisition, construction, development, and/or normal use of the assets. The Company also records a corresponding asset that is depreciated over the life of the asset. Subsequent to the initial measurement of the asset retirement obligation, the obligation will be adjusted at the end of each period to reflect the passage of time and changes in the estimated future cash flows underlying the obligation. The Company is required to adopt SFAS No. 143 on January 1, 2003. The adoption of SFAS No. 143 did not have a material effect on the Company's financial statements.

In April 2002 the FASB issued SFAS No. 145, "Rescission of FASB Statements No. 4, 44 and 64, Amendment of FASB Statement No. 13, and Technical Corrections" (SFAS No. 145). SFAS No. 145 amends existing guidance on reporting gains and losses on the extinguishment of debt to prohibit the classification of the gain or loss as extraordinary, as the use of such extinguishments have become part of the risk management strategy of many companies. SFAS No. 145 also amends SFAS No. 13 to require sale-leaseback accounting for certain lease modifications that have economic effects similar to sale-leaseback transactions. The provisions of the Statement related to the rescission of Statement No. 4 is applied in fiscal years beginning after May 15, 2002. Earlier application of these provisions is encouraged. The provisions of the Statement related to Statement No. 13 were effective for transactions occurring after May 15, 2002, with early application encouraged. The adoption of SFAS No. 145 is not expected to have a material effect on the Company's financial statements.

In June 2002 the FASB issued SFAS No. 146, "Accounting for Costs Associated with Exit or Disposal Activities" (SFAS No 146). SFAS No. 146 addresses financial accounting and reporting for costs associated with exit or disposal activities and nullifies Emerging Issues Task Force (EITF) Issue 94-3, Liability Recognition for Certain Employee Termination Benefits and Other Costs to Exit an Activity. The provisions of this Statement are effective for exit or disposal activities that are initiated after December 31, 2002, with early application encouraged. The adoption of SFAS No. 146 is not expected to have a material effect on the Company's financial statements.

In November 2002 the FASB issued Interpretation No. 45, Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness to Others, an interpretation of FASB Statements No. 5, 57 and 107 and a rescission of FASB Interpretation No. 34. This Interpretation elaborates on the

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

disclosures to be made by a guarantor in its interim and annual financial statements about its obligations under guarantees issued. The Interpretation also clarifies that a guarantor is required to recognize, at inception of a guarantee, a liability for the fair value of the obligation undertaken. The initial recognition and measurement provisions of the Interpretation are applicable to guarantees issued or modified after December 31, 2002 and are not expected to have a material effect on the Company's financial statements. The disclosure requirements are effective for financial statements of interim or annual periods ending after December 15, 2002.

In January 2003 the FASB issued Interpretation No. 46, Consolidation of Variable Interest Entities, an interpretation of ARB No. 51. This Interpretation addresses the consolidation by business enterprises of variable interest entities as defined in the Interpretation. The Interpretation applies immediately to variable interests in variable interest entities obtained after January 31, 2003. For nonpublic enterprises, such as IFC, with a variable interest in a variable interest entity created before February 1, 2003, the Interpretation is applied to the enterprise no later than the end of the first annual reporting period beginning after June 15, 2003. IMH is subject to a majority of the risk of loss from IFC and will therefore need to be included as a consolidated company no later than the first reporting period beginning after June 15, 2003. The result of this consolidation would be the elimination of certain inter company balances and certain gains or losses. The adoption of FIN 46 is not expected to have a material effect on the IFC financial statements.

Note B—Securities Available-for-Sale

During 2000 IFC wrote-off substantially all of its investment securities available-for-sale, which were secured by franchise loan receivables. As of December 31, 2002 and 2001 the carrying value of franchise loan receivables was \$129,000 and \$3.4 million, respectively.

Note C—Mortgages Held-for-Sale

Mortgages acquired and originated by the mortgage operations are primarily adjustable- and fixed rate Alt-A mortgages and, to a lesser extent, B/C mortgages secured by first and second liens on single-family residential properties. During the years ended December 31, 2002 and 2001 the mortgage operations acquired \$6.0 billion and \$3.2 billion, respectively, of mortgages, including net premiums paid, and sold \$5.6 billion and \$3.2 billion, respectively, of mortgages, including net premiums paid, and sold \$5.6 billion and \$3.2 billion, respectively, of mortgages as REMICs, whole loan sales to third party investors and the long-term investment operations. During the years ended December 31, 2002 and 2001 the mortgage operations sold \$3.9 billion and \$1.5 billion, respectively, in principal balance of mortgages to the long-term investment operations at net premiums of \$71.7 million and \$24.4 million, respectively. As of December 31, 2002 and 2001 approximately 57% and 50%, respectively, of mortgages held-for-sale were collateralized by properties located in California. Mortgages held-for-sale consisted of the following for the indicated periods:

	 At December 31,		
	 2002	2001	
	(in thousands)		
Mortgages held-for-sale	\$ 488,700 \$	172,204	
Premiums on mortgages held-for-sale	8,980	2,126	
Mark-to-market loss – SFAS 133	(1,803)	(158)	
Total mortgages held-for-sale	\$ 495,877 \$	174,172	

Mark-to-market loss-SFAS 133 reflects the fair value of \$229.4 million and \$36.0 million in notional balance of derivative instruments that are hedging interest rates on mortgages held-for-sale as of December 31, 2002 and 2001, respectively. As of December 31, 2002 and 2001 \$1.0 million and \$91,000, respectively, of unrealized mark-to-market loss-SFAS 133 is reflected in accumulated other comprehensive income (loss), after applying a 42% tax rate, and \$757,000 and \$67,000, respectively, is reflected in deferred taxes.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Included in other liabilities as of December 31, 2002 and 2001 was an allowance for repurchases of \$1.8 million and \$2.3 million, respectively. Allowance for repurchases is a provision that is maintained for the purpose of purchasing previously sold mortgages from third party investors if there has been a breach of representations or warranties and then which may be subsequently sold at a loss. In the opinion of management, the potential exposure related to these representations and warranties will not have a material adverse effect on IFC's financial condition and results of operations.

Note D—Premises and Equipment

Premises and equipment consisted of the following for the periods indicated:

	 At December 31,			
	 2002	2001		
	(in thou	sands)		
Premises and equipment	\$ 13,891	\$	11,733	
Less: Accumulated depreciation	 (8,943)		(6,400)	
Total premises and equipment	\$ 4,948	\$	5,333	

Note E—Master Servicing Rights and Mortgage Servicing Rights

The mortgage operations subcontract all servicing obligations to independent third parties pursuant to sub-servicing agreements. However, master servicing rights on substantially all mortgage acquisitions and originations are retained. Master servicing fees are generally 0.03% per annum on the declining principal balances of the mortgages serviced. The value of master servicing fees is subject to prepayment and interest rate risks on the transferred financial assets. The retained interest in master servicing fees is included in mortgage servicing rights on the balance sheet. For the years ended December 31, 2002 and 2001 the mortgage operations master serviced 48,520 and 37,972, respectively, of mortgages with a principal balance of \$8.7 billion and \$5.6 billion, respectively. As of December 31, 2002 and 2001 the mortgage operations master serviced \$3.0 billion and \$3.1 billion, respectively, of mortgages collateralizing REMICs and \$4.9 billion and \$2.1 billion, respectively, of mortgages collateralizing CMOs. Activity for master servicing rights for the indicated periods was as follows:

		For the year ended December 31,				
		2002	2001			
		(in thousands)				
Beginning balance	\$	4,024 \$	2,350			
Add: Net additions		4,583	2,861			
Subtract: Impairment of mortgage servicing rights		(287)	—			
Amortization of mortgage servicing rights		(2,257)	(1,187)			
	¢	C 0 C 0	4.024			
Ending balance of master servicing rights	\$	6,063 \$	4,024			

The mortgage operations generally acquire and originate all mortgages on a servicing released basis and subsequently sell servicing rights upon the sale or securitization of the mortgage. As of December 31, 2002 and 2001 the mortgage operations owned servicing rights on \$2.7 billion and \$1.8 billion, respectively, in principal balance of mortgages. Activity for mortgage servicing rights for the indicated periods was as follows:

		For the year ended December 31,					
	2	002	2001				
		(in thousands)				
Beginning balance	\$	4,444 \$	8,588				
Add: Net additions		137	13				
Subtract: Impairment of mortgage servicing rights		(568)	(825)				
Amortization of mortgage servicing rights		(1,802)	(3,332)				
Ending balance of mortgage servicing rights	\$	2,211 \$	4,444				

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note F—Borrowings from IWLG

The mortgage operations had uncommitted warehouse line agreements to obtain financing of up to \$600.0 million from IWLG to provide interim mortgage financing during the period that the mortgage operations accumulate mortgages until the mortgages are securitized or sold. The mortgage operations had \$15.2 million in pledge accounts with IWLG as of December 31, 2002, which allows them to finance 100% of the fair market value of their mortgages at prime minus 0.50%. As of December 31, 2002 and 2001 the mortgage operations had \$491.4 million and \$174.1 million, respectively, in outstanding warehouse advances with IWLG.

Note G—Income Taxes

Income taxes were as follows for the periods indicated:

	For the year ended December 31,							
	 2001		2000		2002			
		(i	n thousands)					
Current income taxes:								
Federal	\$ 11,763	\$	7,260	\$	42			
State	3,619		2,167		1,832			
Total current income taxes	15,382		9,427		1,874			
Deferred income taxes:								
Federal	(2,229)		(924)		(960)			
State	(558)		(203)		(144)			
Total deferred income taxes	(2,787)		(1,127)		(1,104)			
		-						
Total income taxes	\$ 12,595	\$	8,300	\$	770			
		_		_				

Effective income taxes differ from the amount determined by applying the statutory Federal rate of 35%, 35% and 34% for the periods indicated as follows:

		For the year ended December 31,							
		2002		2001		2000			
				(in thousands)					
Income taxes (benefit) at Federal tax rate	\$	10,444	\$	6,763	\$	(343)			
State tax, net of Federal income tax (benefit)		1,990		1,277		(89)			
California franchise tax accrual, net of Federal income tax		_		_		1,203			
Other		161		260		(1)			
Total income taxes	\$	12 505	\$	0.200	¢	770			
Total income taxes	φ	12,595	Φ	8,300	\$	//0			

The tax affected cumulative temporary differences that give rise to deferred tax assets and liabilities for the periods indicated are as follows:

	At December 31,			
	2002	2001		
	(in thou	sands)		
Deferred tax assets:				
Deferred revenue	\$ 2,118	\$ 1,867		
Forward commitments	—			
Depreciation and amortization	690	505		
Salary accruals	1,026	1,086		
Other accruals	1,067	1,035		
Loan mark-to-market	2,331	252		
Non-accrual loans	77	55		
Accrued interest	281	281		
State franchise tax	1,267	759		
Provision for repurchases	733	972		

Total gross deferred tax assets	9,590	6,812
Deferred tax liabilities:		
Mortgage servicing rights	(3,444)	(3,530)
Deferred gain	(139)	(223)
REMIC interest	(201)	(40)
Total gross deferred tax liabilities	(3,784)	(3,793)
Net deferred tax (asset) liability	\$ 5,806	\$ 3,019

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

As of December 31, 2002 the mortgage operations had no net operating loss carry-forwards for federal and state income tax purposes and no minimum tax carry-forward for federal purposes.

Management believes that the deferred tax asset will more likely than not be realized due to the reversal of the deferred tax liability and expected future taxable income. In determining the possible future realization of deferred tax assets, future taxable income from the following sources are taken into account: (a) the reversal of taxable temporary differences, (b) future operations exclusive of reversing temporary differences.

Note H—Disclosures About Fair Value of Financial Instruments

The estimated fair value amounts have been determined by management using available market information and appropriate valuation methodologies, however, considerable judgment is necessarily required to interpret market data to develop the estimates of fair value. Accordingly, the estimates presented are not necessarily indicative of the amounts the mortgage operations could realize in a current market exchange. The use of different market assumptions and/or estimation methodologies may have a material effect on the estimated fair value amounts.

	 December	002	_	December 31, 2001				
	 Carrying Estimated Amount Fair Value				Carrying Amount		Estimated Fair Value	
			(in tho	usands)				
Assets								
Cash and cash equivalents	\$ 22,773	\$	22,773	\$	28,612	\$	28,612	
Securities available-for-sale	129		129		3,394		3,394	
Mortgages held-for-sale, including derivative assets	495,877		507,014		174,172		174,187	
Liabilities								
Borrowings from IWLG	\$ 491,383	\$	491,383	\$	174,136	\$	174,136	
Due to affiliates	14,500		14,500		14,500		14,500	

The fair value estimates as of December 31, 2002 and 2001 are based on pertinent information available to management as of that date. Although management is not aware of any factors that would significantly affect the estimated fair value amounts, such amounts have not been comprehensively revalued for purposes of these financial statements since those dates and, therefore, current estimates of fair value may differ significantly from the amounts presented. The following describes the methods and assumptions used by management in estimating fair values:

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Cash and Cash Equivalents

Fair value approximates carrying amounts as these instruments are demand deposits and do not present unanticipated interest rate or credit concerns.

Securities Available-for-Sale

To determine the value of securities, management estimates future rates of prepayments, prepayment penalties to be received, delinquencies, defaults and default loss severity and their impact on estimated cash flows.

Mortgages Held-for-Sale

Fair value of mortgages held-for-sale is estimated based on quoted market prices from dealers and brokers for similar types of mortgages.

Borrowings from IWLG

Fair value approximates carrying amounts because of the short-term maturity of the liabilities.

Other Borrowings

Fair value approximates carrying amounts because of the short-term maturity of the liabilities.

Due From/To Affiliates

Fair value approximates carrying amounts because of the short-term maturity of the liabilities and does not present unanticipated interest rate or credit concerns.

Loan Commitments

Fair value of loan commitments is determined in the aggregate based on current investor yield requirements.

Note I—401(k) Plan

After meeting certain employment requirements, employees of the mortgage operations can participate in IFC's 401(k) plan. Under the 401(k) plan, employees may contribute up to 25% of their salaries, pursuant to certain restrictions. IFC and its divisions and subsidiaries will match 50% of the first 4% of employee contributions. Additional contributions may be made at the discretion of management. During the years ended December 31, 2002, 2001 and 2000 the mortgage operations recorded approximately \$714,000, \$604,000 and \$384,000, respectively, for matching and discretionary contributions.

Note J—Related Party Transactions

Related Party Cost Allocations

The mortgage operations charge IMH and IWLG for management and operating services based upon usage which management believes is reasonable. Total cost allocations charged by the mortgage operations to IMH and IWLG for the years ended December 31, 2002, 2001 and 2000 were \$2.5 million, \$1.7 million and \$1.5 million, respectively.

IFC and IMH entered into a premises operating sublease agreement to rent approximately 74,000 square feet of office space in Newport Beach, California, for a ten-year term, which expires in May 2008. IFC pays monthly rental expenses and allocates the cost to subsidiaries and affiliated companies on the basis of square footage occupied. Total rental expense for the years ended December 31, 2002, 2001, and 2000 was \$2.3 million, \$2.0

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

million and \$1.6 million, respectively, of which \$2.2 million, \$1.9 million and \$1.5 million, respectively, was allocated to the mortgage operations as most of the Company's employees are employed by the mortgage operations.

Credit Arrangements

The mortgage operations maintain a warehouse financing facility with IWLG. Advances under the warehouse facility bears interest at Bank of America's prime rate minus 0.50%. As of December 31, 2002 and 2001 amounts outstanding on the mortgage operations warehouse line with IWLG were \$491.4 million and \$174.1 million, respectively. Interest expense recorded by the mortgage operations related to its warehouse line with IWLG for the years ended December 31, 2002, 2001, and 2000 was \$18.0 million, \$18.4 million are \$28.1 million, respectively.

During the normal course of business the mortgage operations may advance or borrow funds on a short-term basis with affiliated companies. Advances to affiliates are reflected as "Due from affiliates" while borrowings are reflected as "Due to affiliates" on the consolidated balance sheet. These short-term advances and borrowings bear interest at a fixed rate of 8% per annum. Interest income recorded by the mortgage operations related to short-term advances with affiliates for the years ended December 31, 2002, 2001 and 2000 was \$2.7 million, \$758,000 and \$161,000, respectively. Interest expense recorded by the mortgage operations related to short-term advances with affiliates for the years ended December 31, 2002, 2001 and 2000 was \$2.6,000, respectively.

During 1999 IFC was advanced \$14.5 million in cash by IMH at an interest rate of 9.50% per annum due June 30, 2004 in exchange for an interest only note to fund the operations of IFC and other strategic opportunities deemed appropriate by IFC. At December 31, 2002 and 2001 the amount outstanding on the note was \$14.5 million. Interest expense recorded by IFC related to the note for the years ended December 31, 2002, 2001 and 2000 was \$1.4 million.

Transactions with IMH

During the years ended December 31, 2002 and 2001 the mortgage operations sold to IMH mortgages having a principal balance of \$3.9 billion and \$1.5 billion, respectively, at premiums of \$71.7 million and \$22.4 million, respectively. Master servicing rights on all mortgages purchased by IMH were retained by the mortgage operations.

On February 20, 2001 IFC purchased \$5.0 million of IMH's Preferred Stock from LBP, Inc. (LBPI) at cost plus accumulated dividends. On March 27, 2001 IFC purchased an additional \$5.0 million of IMH's Preferred Stock from LBPI for \$5.25 million plus accumulated dividends. In August 2001 the Preferred Stock was converted into 2,118,644 shares of IMH common stock. Subsequently, IFC sold 1.7 million shares of IMH common stock for a gain of \$3.5 million. As of December 31, 2002 IFC did not own any shares of IMH common stock.

Indebtedness of Management

On December 10, 2001 IFC provided William B. Ashmore, President of IFC, with a \$600,000 adjustable rate mortgage to provide financing with an initial rate of 4.13%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. As of December 31, 2002 and 2001 the outstanding principal balance on the mortgage was \$587,000 and \$600,000 respectively.

On November 15, 2001 IFC provided Richard J. Johnson, Executive Vice President and Chief Financial Officer of IFC, with a \$140,000 fixed rate mortgage to provide financing with an interest rate of 6.50%. In the

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. Subsequently, this loan has been sold to a non-affiliate.

On February 27, 2003 IFC provided William B. Ashmore, President of IFC, with a \$295,600 fixed rate mortgage to provide financing with an initial rate of 5.50%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons.

Note K—Derivative Instruments and Hedging Activities

The mortgage operations enter into forward commitments and derivative transactions to mitigate changes in the value of its "locked pipeline." The locked pipeline are mortgages that have not yet been acquired, however, the mortgage operations have committed to acquire the mortgages in the future at predetermined rates through rate-lock commitments. The mortgage operations records the value of its locked pipeline as it qualifies as a derivative instrument under the provisions of SFAS 133, however, it does not qualify for hedging treatment. Therefore, the locked pipeline and the related value of derivative instruments is marked to market each reporting period along with a corresponding entry to non-interest expense. The mortgage operations generally sells call or buys put options on U.S. Treasury bonds and mortgage-backed securities to hedge against adverse movements of interest rates affecting the value of mortgage increases and the option is exercised. The mortgage operations also has the additional risk of not being able to enter into a closing transaction if a liquid secondary market does not exist. The risk of buying a put option is limited to the premium paid for the put option. These hedges are treated as cash flow hedges under the provisions of SFAS 133 to hedge forecasted transactions with the entire change in market value of the hedges recorded through accumulated other comprehensive income.

As of December 31, 2002 and 2001 IFC had \$286.0 million and \$48.0 million, respectively, in outstanding commitments to sell mortgages through mortgage-backed securities. These commitments allow the mortgage operations to enter into mandatory commitments when the mortgage operations notify the investor of its intent to exercise a portion of the forward delivery contracts. IFC was not obligated under mandatory commitments to deliver loans to such investors as of December 31, 2002 and 2001. The credit risk of forward contracts relates to the counter-parties ability to perform under the contract. Management evaluates counter-parties based on their ability to perform prior to entering into any agreements.

As of December 31, 2002 and 2001 IFC had written option contracts with an outstanding principal balance of \$10.0 million and \$5.0 million, respectively. IFC may sell call or buy put options on U.S. Treasury bonds and mortgage-backed securities. The risk in writing a call option is that IFC gives up the opportunity for profit if the market price of the mortgages increases and the option is exercised. IFC also has the additional risk of not being able to enter into a closing transaction if a liquid secondary market does not exist. The risk of buying a put option is limited to the premium paid for the put option.

The following table presents certain information related to derivatives and the related component in the financial statements as of December 31, 2002 (dollars in thousands):

	Fair Value of Derivatives		Related Amount In OCI		Amount		Amount in Earnings		Amount in Deferred Taxes		Related Amount in Loans Held-for-Sale		Related Amount in Other Assets/ Liabilities	
Fannie Mae forward commitments hedging														
mortgages held-for-sale	\$ (1,803)	\$	(1,046)	\$	_	\$	757	\$	(1,803)	\$	_			
Fannie Mae forward commitments hedging														
pipeline	(2,863)				(2,935)				—		(2,863)			
Value of locked pipeline	4,234		_		4,635		_		_		4,234			
Totals	\$ (432)	\$	(1,046)	\$	1,700	\$	757	\$	(1,803)	\$	1,371			
		_		_		_		_						

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The following table presents certain information related to derivative instruments and hedging activities as of December 31, 2001 (dollars in thousands):

	 Fair Value of Derivatives		Related Amount in OCI	<u> </u>	Amount in Earnings		Amount in Deferred Taxes	 Related Amount in Loans Held-for-Sale		Related Amount in Other Assets/ Liabilities
Fannie Mae forward commitments hedging mortgages held-for-sale	\$ (158)	\$	(91)	\$	_	\$	(67)	\$ (158)	\$	
Fannie Mae forward commitments hedging pipeline	_	·	_		(72)	·	_	_	·	72
Value of locked pipeline	_		_		401		_	_		(401)
Option on Fannie Mae forward	 2				(2)			 		2
Totals	\$ (156)	\$	(91)	\$	327	\$	(67)	\$ (158)	\$	(327)

Note L—Commitments and Contingencies

Master Servicing

Properties securing mortgages in the mortgage operations master servicing portfolio are primarily located in California. As of December 31, 2002 and 2001 approximately 53% and 46%, respectively, of mortgages in the master servicing portfolio were located in California. As of December 31, 2002 and 2001 the mortgage operations master serviced mortgages totaling approximately \$8.7 billion and \$5.6 billion, respectively, in principal balance, of which \$6.0 billion and \$3.8 billion, respectively, were serviced for others. As of December 31, 2002 and 2001 the mortgage operations master serviced \$3.0 billion and \$3.1 billion, respectively, of mortgages collateralizing REMICs and \$4.9 billion and \$2.1 billion, respectively, of mortgages collateralizing CMOs. Related fiduciary funds are held in trust for investors in non-interest bearing accounts. These funds are segregated in special bank accounts.

Master Commitments

The mortgage operations establishes mortgage purchase commitments (Master Commitments) with sellers that, subject to certain conditions, entitle the seller to sell and obligate the mortgage operations to purchase a specified dollar amount of mortgages over a period generally ranging from six months to one year. The terms of each Master Commitment specify whether a seller may sell mortgages to the mortgage operations on a mandatory, best efforts or optional basis. Master commitments generally do not obligate the mortgage operations to purchase mortgages at a specific price, but rather provide the seller with a future outlet for the sale of its originated mortgages based on quoted prices at the time of purchase. As of December 31, 2002 and 2001 the mortgage operations had outstanding short-term Master Commitments with 136 and 156 sellers, respectively, to purchase mortgages in the aggregate principal amount of \$4.1 billion and \$2.5 billion, respectively, over periods ranging from one month to one year, of which \$2.3 billion and \$1.3 billion, respectively, had been purchased or committed to be purchased pursuant to rate locks. These rate-locks were made pursuant to Master Commitments, bulk rate-locks and other negotiated rate-locks. There is no exposure to credit loss in this type of commitment until the loans are funded, and interest rate risk associated with the short-term commitments is mitigated by the use of forward contracts to sell loans to investors.

Sellers who have entered into Master Commitments may sell mortgages to the mortgage operations by executing individual, bulk or other rate-locks, or "rate-lock." Each rate-lock, in conjunction with the related Master Commitment, specifies the terms of the related sale, including the quantity and price of the mortgages or the formula by which the price will be determined, the rate-lock type and the delivery requirements. Historically, the up-front fee paid by a seller to obtain a Master Commitment on a mandatory delivery basis is often refunded

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

pro rata as the seller delivers loans pursuant to rate-locks. We retain any remaining fee after the Master Commitment expires.

Following the issuance of a specific rate-lock the mortgage operations is subject to the risk of interest rate fluctuations and enters into hedging transactions to diminish such risk. Hedging transactions may include mandatory or optional forward sales of mortgages or mortgage-backed securities, interest rate caps, floors and swaps, mandatory forward sales, mandatory or optional sales of futures, and other financial futures transactions. The nature and quantity of hedging transactions are determined by the management based on various factors, including market conditions and the expected volume of mortgage purchases.

Bulk and Other Rate-Locks. The mortgage operations also acquire mortgages from sellers that are not purchased pursuant to Master Commitments. These purchases may be made on an individual rate-lock basis. Bulk rate-locks obligate the seller to sell and the mortgage operations to purchase a specific group of mortgages, generally ranging from \$500,000 to \$125 million in aggregate committed principal amount, at set prices on specific dates. Bulk rate-locks enable the mortgage operations to acquire substantial quantities of mortgages on a more immediate basis. The specific pricing, delivery and program requirements of these purchases are determined by negotiation between the parties but are generally in accordance with the provisions of our seller/servicer guide. Due to the active presence of investment banks and other substantial investors in this area, bulk pricing is extremely competitive. Mortgages are also purchased from individual sellers, typically smaller originators of mortgages, who do not wish to sell pursuant to either a Master Commitment or bulk rate-lock. The terms of these individual purchases are based primarily on our seller/servicer guide and standard pricing provisions.

Mandatory, Best-Efforts and Optional Rate-Locks. Mandatory rate-locks require the seller to deliver a specified quantity of mortgages over a specified period of time regardless of whether the mortgages are actually originated by the seller or whether circumstances beyond the seller's control prevent delivery. The mortgage operations are required to purchase all mortgages covered by the rate-lock at prices established at the time of rate-lock. If the seller is unable to deliver the specified mortgages, it may instead deliver comparable mortgages approved by the mortgage operations within the specified delivery time. Failure to deliver the specified mortgages or acceptable substitute mortgages under a mandatory rate-lock obligates the seller to pay a penalty. In contrast, mortgages sold on a best efforts basis must be delivered to the mortgage operations only if they are actually originated by the seller. The best-efforts rate-lock provides sellers with an effective way to sell mortgages during the origination process without any penalty for failure to deliver. Optional rate-locks give the seller the option to deliver mortgages to us at a fixed price on a future date and require the payment of up-front fees. The mortgage operations retain any up-front fees paid in connection with optional rate-locks if the mortgages are not delivered.

Sale of Mortgages and Servicing Rights

In the ordinary course of business, the mortgage operations are exposed to liability under representations and warranties made to purchasers and insurers of mortgages and the purchasers of servicing rights. Under certain circumstances, the mortgage operations are required to repurchase mortgages if there has been a breach of representations or warranties. In the opinion of management, the potential exposure related to these representations and warranties will not have a material adverse effect on financial condition and results of operations. As of December 31, 2002 and 2001 included in other liabilities are \$1.8 million and \$2.3 million, respectively, in allowances for repurchases related to possible off-balance sheet recourse and repurchase agreement provisions.

Lease Committments

IFC has entered in premise operating leases with minimum rental commitments under non-cancelable leases of \$727,000 through 2005.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Legal Proceedings

On September 1, 2000, a complaint captioned Michael P. and Shellie Gilmor v. Preferred Credit Corporation and Impac Funding Corporation, et al. was filed in the Circuit Court for Clay County, Missouri, Case No. CV100-4263-CC, as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loans Act and Merchandising Practices Act. In July 2001, the Missouri complaint was amended to include IMH and other Impac-related entities. A plaintiffs class was certified on January 2, 2003. Plaintiffs have also filed a motion to certify a defendant class, which motion will be held in abeyance pending resolution by the Missouri Supreme Court of certain issues relating to the applicable statute of limitations. On August 2, 2001, a complaint captioned Frazier, et al. v. Preferred Credit, et al., was filed in the Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004762-01. This case was subsequently removed to federal court, Case No. 01-2714 (W.D. Tenn). This was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration of the ruling. Plaintiffs' motion was denied in its entirety by order dated March 7, 2003. This case is currently dismissed as to all defendants. On August 8, 2001, a complaint captioned Mattie L. Street v. PSB Lending Corp., et al., was filed in the Circuit Court of Tennessee for the Thirtieth Judicial District at Memphis, Case No. CT004888-01. This case was subsequently removed to federal court. Case No. 01-2751 (W.D. Tenn.). The Street action was also stated as a purported class action lawsuit alleging violations of Tennessee's usury statute and Consumer Protection Act. On July 31, 2002, the matter was dismissed and plaintiffs then filed a motion for reconsideration. Plaintiffs' motion was denied in its entirety by order dated March 6, 2003. This case is currently dismissed as to all defendants. On October 2, 2001, a complaint captioned Deborah Searcy, Shirley Walker, et al. vs. Impac Funding Corporation, Impac Mortgage Holdings, Inc. et al. was filed in the Wayne County Circuit Court, State of Michigan, as a purported class action lawsuit alleging that the defendants violated Michigan's Secondary Mortgage Loan Act, Credit Reform Act and Consumer Protection Act. In February 2003, the Court granted a motion to accept for filing an amended complaint adding certain defendants, including an Impac-related entity, and dropping others, including certain Impac-related entities. Service of the amended complaint has not yet occurred. On October 10, 2001, a complaint captioned Haves v Impac Funding Corporation, et al. was filed in the Circuit Court of Vanderburgh County, Indiana as Case No. 82C01-0110-CP580. This was stated as a purported class action lawsuit alleging a violation of the Indiana Uniform Consumer Credit Code when the loans were originated. This matter was dismissed and the plaintiffs have noticed an appeal to the Indiana Court of Appeals. This appeal remains pending. On November 30, 2001, a complaint captioned Garry Lee Skinner and Judy Cooper Skinner, et al. v. Preferred Credit, et al. was filed in the Superior Court of Durham County, North Carolina as Case No. 1CV-05596. This is stated as a purported class action alleging a violation of the North Carolina Interest Statutes and Unfair and Deceptive Trade Practices Act when the secondary mortgage loans were originated by the defendants. A motion to dismiss the complaint has been filed. This motion remains pending.

All of the purported class action lawsuits are similar in nature in that they allege that the mortgage loan originators violated the respective state's statutes by charging excessive fees and costs when making second mortgage loans on residential real estate. The complaints allege that IFC was a purchaser, and is a holder, along with other Impac-related entities, of second mortgage loans originated by other lenders. The plaintiffs in the lawsuits are seeking damages that include disgorgement, restitution, rescission, actual damages, statutory damages, exemplary damages, pre-judgement interest and punitive damages. No material amount of damages are specified in the complaints.

The mortgage operations believes that it has meritorious defenses to the above claims and intends to defend these claims vigorously. Nevertheless, litigation is uncertain and the mortgage operations may not prevail in the lawsuits and can express no opinion as to its ultimate outcome.

The mortgage operations are a party to litigation and claims, which are normal in the course of its operations. While the results of such other litigation and claims cannot be predicted with certainty, management believes the final outcome of such other matters will not have a material adverse effect on the financial condition or operating results.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note M—Retained Interests in Securitizations

During the years ended December 31, 2002 and 2001 the mortgage operations sold \$600.0 million and \$1.7 billion of mortgages through the issuance of REMIC securitizations. The mortgage operations sold to third party investors all mortgage-backed securities and MSRs created from REMICs at the time of securitization. However, the mortgage operations retained master servicing fees on all securitizations by virtue of its responsibilities as master servicer. The mortgage operations generally earns 0.03% per annum on the declining principal balance of loans master serviced. The value of master servicing fees is subject to prepayment and interest rate risks on the transferred financial assets. The retained interest in master servicing fees is included in mortgage servicing rights on the balance sheet. During the years ended December 31, 2002 and 2001 the fair value assigned to the retained interest in master servicing fees at the date of securitization was \$4.6 million and \$2.9 million, respectively. Management uses certain assumptions and estimates in clude projections for mortgage prepayment rates and discount rates commensurate with the risks involved and are reviewed periodically by management. If these assumptions change, the related asset and income would be affected. As of December 31, 2002 and 2001 the carrying amount and estimated fair value of MSRs was \$8.3 million and \$8.5 million, respectively. As of December 31, 2002 and 2001 the mortgage operations had \$495.9 million and \$174.2 million, respectively, of mortgages held-for-sale, which were available for securitization.

Note N—Quarterly Financial Data (unaudited)

Selected quarterly financial data for 2002 for the periods indicated follows (in thousands):

		For the Three Months Ended,						
	Dec	cember 31,		September 30,		June 30,		March 31,
Net interest income								
	\$	2,506	\$	2,081	\$	2,010	\$	1,671
Non-interest income		21,084		13,192		19,529		17,536
Non-interest expense, including income taxes		19,290		12,490		16,031		14,552
Net earnings		4,300		2,783		5,508		4,655

Selected quarterly financial data for 2001 for the indicated periods follows (in thousands):

	 For the Three Months Ended,						
	 December 31,		September 30,		June 30,		March 31,
Net interest income							
	\$ 1,597	\$	940	\$	479	\$	294
Non-interest income	18,518		13,140		13,709		8,727
Non-interest expense, including income taxes and cumulative effect							
of change in accounting principle	17,029		10,991		10,639		7,723
Net earnings	3,086		3,089		3,549		1,298

Note O—Allowance for Repurchases

Activity for allowance for repurchases was as follows:

		Fo	r the y	year ended December	31,	
	_	2002		2001		2000
				(in thousands)		
balance	\$	2,332	\$	300	\$	592
r loan losses		2,806		3,498		371
hases		(3,377)		(1,466)		(663)
	-					
ice	\$	1,761	\$	2,332	\$	300
	_		_		_	

RESOLUTION TO CHANGE PRINCIPAL OFFICE OR RESIDENT AGENT

The directors/stockholders/general partner/authorized person of IMPAC MORTGAGE HOLDINGS, INC., organized under the laws of Maryland, passed the following resolution:

CHECK APPLICABLE BOX(ES)

x The principal office is changed from: (old address)

c/o Ballard Spahr Andrews & Ingersoll 300 East Lombard Street, Suite 1900 Baltimore, Maryland 21202

to: (new address)

c/o CSC-Lawyers Incorporating Service Company 11 East Chase Street Baltimore, Maryland 21202

x The name and address of the resident agent is changed from:

James J. Hanks, Jr. c/o Ballard Spahr Andrews & Ingersoll 300 East Lombard Street, Suite 1900 Baltimore, Maryland 21202

to:

CSC-Lawyers Incorporating Service Company 11 East Chase Street Baltimore, Maryland 21202

I certify under penalties of perjury the foregoing is true.

/s/ RONALD MORRISON

Ronald M. Morrison Secretary and General Counsel

I hereby consent to my designation in this document as resident agent for this entity.

CSC LAWYERS INCORPORATING SERVICE COMPANY

CUST ID: 0000958355 WORK ORDER: 0000638751 DATE: 09-11-2002 08:51 AM AMT. PAID: \$60.00 By:

/s/ ANNE MARTIN

Resident Agent Anne M. Martin, Asst. Vice President

301 W. Preston Street, Room 801, Baltimore, MD 2120

THE IMPAC COMPANIES DEFERRED COMPENSATION PLAN

ARTICLE I

INTRODUCTION

1.1 <u>Purpose</u>. The purpose of this Plan is to enable the Sponsor and Affiliated Companies to provide a competitive tax-deferred capital accumulation vehicle in order to attract and retain key management.

1.2 <u>Effective Date and Term</u>. This Plan is effective as of February 1, 2000 (the "Effective Date"), unless otherwise expressly provided herein.

1.3 <u>Participation</u>. Participation in this Plan is open only to employees of the Sponsor or any Affiliated Company who meet the requirements specified in Section 4.1 of the Plan and who are selected for participation in the Plan by the Committee. Participation in this Plan by any such employees, and the payment of any benefits under this Plan to any such employee, shall be governed by the terms of this Plan and by the terms of the Participation Agreement entered into with respect to this Plan by the Company and such employee pursuant to Section 1.4 hereof.

1.4 <u>Participation Agreement</u>. As a condition to the commencement of participation in this Plan, each employee approved for participation must enter into an agreement covering such employee's participation (a "Participation Agreement") which agreement shall be executed by the Sponsor and such employee and, if such employee is employed by an Affiliated Company, such Affiliated Company. Each Participation Agreement shall include such terms and conditions related to the employee's participation in the Plan as the Committee may deem appropriate.

1.5 <u>Applicability of ERISA</u>. This Plan is intended to be a "top-hat" plan -- that is, an unfunded plan maintained primarily for the purpose of providing deferred compensation to a select group of management or highly compensated employees within the meaning of ERISA. As such, the Plan is exempt from the participation and vesting requirements of Part 2 of Title I of ERISA, the funding requirements of Part 3 of Title I of ERISA, and the fiduciary responsibility provisions of Part 4 of Title I of ERISA.

ARTICLE II

DEFINITIONS

2.1 Account. "Account" shall mean one of the separate Accounts maintained by the Committee pursuant to the provisions of Section 5.1.

2.2 <u>Active Participant</u>. "Active Participant" shall mean a Participant whose participation in this Plan has been approved pursuant to the provisions of Section 1.3, who has entered into a Participation Agreement pursuant to the provisions of Section 1.4, and whose active participation in the Plan has not been terminated pursuant to the provisions of Section 3.5.

2.3 <u>Affiliated Company</u>. "Affiliated Company" shall mean any of the following affiliates of the Sponsor which the Board expressly designates as having the status of an Affiliated Company for purpose of this Plan: (a) any corporation which is included in a controlled group of corporations, within the meaning of Section 414(b) of the Code, of which controlled group the Sponsor is also a member; (b) any trade or business which is under common control with the Sponsor within the meaning of Section 414(c) of the Code; (c) any service organization that is included in an affiliated service group, within the meaning of Section 414(m) of the Code, of which affiliated service group the Sponsor is also a member; (d) any other entity required to be aggregated with the Sponsor pursuant to regulations promulgated under Section 414(o) of the Code and (e) any other entity or entities which are designated by the Board as Affiliates as listed on Exhibit A hereto and which adopt this Plan.

2.4 <u>Beneficiary</u>. "Beneficiary" shall mean, with respect to any Participant, the person or persons and/or entity or entities designated or deemed designated by the Participant pursuant to Section 12.1 hereof to receive such Participant's Distributable Benefit payable under the Plan following such Participant's death.

2.5 <u>Board of Directors</u>. "Board" and "Board of Directors" shall mean the board of directors of the Sponsor.

2.6 <u>Bonuses</u>. "Bonuses" shall mean bonus compensation which a Participant is entitled to receive from the Company, including but not limited to bonus compensation based on the performance of Sponsor and certain affiliates of Sponsor, such as bonuses to which a Participant may be entitled under an incentive plan. Bonuses shall not include any amounts not currently includible in such Participant's gross income by reason of Code Section 402(e)(3) and/or Code Section 125.

2.7 <u>Bonus Deferrals</u>. "Bonus Deferrals" shall mean Bonuses deferred at the election of the Participant pursuant to a Compensation Deferral Agreement under the provisions of Article IV.

2.8 <u>Code</u>. "Code" shall mean the Internal Revenue Code of 1986, as amended.

2.9 <u>Commissions</u>. "Commissions" shall mean fees paid to salespersons as compensation for their services as specified by the Company on its payroll records. Commissions shall not include any amounts not currently includible in such Participant's gross income by reason of Code Section 402(e)(3) and/or Code Section 125.

2.10 <u>Commission Deferrals</u>. "Commission Deferrals" shall mean Commissions deferred at the election of the Participant pursuant to a Compensation Deferral Agreement under the provisions of Article IV.

2.11 <u>Committee</u>. "Committee" shall mean, as of any date of reference, the committee, if any, then authorized and appointed to administer this Plan as set forth in Section 3.1 hereof.

2.12 <u>Company</u>. "Company" shall mean the Sponsor or an Affiliated Company. A reference to the term "Company" shall be deemed to be (a) a reference to the Sponsor with respect to Participants who are employees of the Sponsor and (b) a reference to an Affiliated Company with respect to Participants who are employees of such Affiliated Company.

2.13 <u>Compensation</u>. "Compensation" shall mean the aggregate of Salary, and Bonuses and Commissions, including any amounts not currently includible in such Participant's taxable income by reason of Code Section 402 or Code Section 125.

2.14 <u>Compensation Deferrals</u>. "Compensation Deferrals" shall mean the Compensation amounts deferred at the election of the Participant pursuant to a Compensation Deferral Agreement under the provisions of Article IV (including Bonus Deferrals, Salary Deferrals and Commission Deferrals).

2.15 <u>Compensation Deferral Agreement</u>. "Compensation Deferral Agreement" shall mean the agreement entered into by a Participant to defer Compensation pursuant to the provisions of Sections 4.2.

2.16 <u>Disability</u>. "Disability" is deemed to exist, with respect to a Participant, if and when, as a result of injury or sickness, the Participant is permanently impaired to such an extent that he or she cannot perform the material duties of his or her position of employment with the Company. The determination hereunder as to whether and when a Participant has a Disability shall be made by the Committee, and for purposes of assisting the Committee in making any such determination, the Committee may require the

Participant to submit to an examination by a competent physician or medical clinic selected by the Committee.

2.17 <u>Distributable Benefit</u>. "Distributable Benefit" shall mean the vested interest of a Participant in his or her Account, which is determined and distributable to the Participant in accordance with the provisions of Articles V, VI, VIII and XI.

2.18 <u>Early Retirement</u>. "Early Retirement" shall mean retirement from service for the Sponsor and all Affiliates on or after the date that the sum of the Participant's age plus Years of Service equals or exceeds 65.

2.19 Effective Date. "Effective Date" shall mean January 1, 2000.

2.20 ERISA. "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

2.21 <u>Investment</u>. "Investment" shall mean the one or more funds under one or more variable life insurance contracts or one or more mutual funds (or any other investment) deemed suitable by the Committee for purposes of determining the investment experience adjustments that shall be made to a Participant's Account pursuant to the provisions of Section 5.4.

2.22 <u>Normal Retirement</u>. "Normal Retirement" shall mean retirement from service for the Sponsor and all Affiliates on or after the date that the Participant attains age 65.

2.23 <u>Participant</u>. "Participant" shall mean any employee of the Company who is selected and approved for participation in this Plan as provided in Section 1.3 hereof and who has executed a Participation Agreement as required under Section 1.4 hereof.

2.24 <u>Plan Year</u>. "Plan Year" shall mean January 1 through December 31.

2.25 <u>Salary</u>. "Salary" shall mean, with respect to any Participant, the base salary which such Participant is entitled to receive from the Company, excluding any amounts not currently includible in such Participant's gross income by reason of Code Section 402(e)(3) and/or Code Section 125. Salary shall not include any Bonuses or Commissions which a Participant is entitled to receive from the Company.

2.26 <u>Salary Deferrals</u>. "Salary Deferrals" shall mean Salary amounts deferred at the election of the Participant pursuant to a Compensation Deferral Agreement under the provisions of Article IV.

2.27 Sponsor. "Sponsor" shall Impac Funding Corporation, a California corporation.

2.28 <u>Termination</u>. "Termination" shall mean the voluntary or involuntary termination of a Participant's employment with the Sponsor and all Affiliated Companies for any reason (including Disability or death).

2.29 <u>Valuation Date</u>. "Valuation Date" shall mean the last of each calendar quarter which is also a business day, as of which the fair market value of Participants' Accounts shall be determined.

2.30 <u>Year of Service</u>. "Year of Service" shall mean any completed 12 month Period of Service including periods prior to the Effective Date. A Period of Service begins on the date an employee first performs an hour of service and ends on the date his or her employment ends. Employment ends on the date that the employee quits, retires, is discharged, dies or (if earlier) the first anniversary of his or her absence for any other reason. Any Period of Service less than 12 months shall be disregarded.

ARTICLE III

ADMINISTRATION OF THE PLAN

3.1 <u>Administration By Committee</u>. This Plan shall be administered by a committee composed of at least four (4) individuals appointed from time to time by the Board (the "Committee"); provided, however, that if the Board does not appoint a committee, the Board shall constitute the Committee. Any member of the Board or the Committee may be a Participant in this Plan, provided, however, that any action to be taken by the Board or Committee solely with respect to the particular interest in this Plan of a Board or Committee member who is also a Participant in this Plan shall be taken by the remaining members of the Board or Committee. Notwithstanding the foregoing, in the event of a Change in Control of the Sponsor, the members of the Value of their Accounts. An election to appoint new Committee members following a Change in Control of the Sponsor shall be held by the Committee within 60 days of the date of such Change in Control pursuant to procedures established by the Committee. At a minimum, such procedures shall provide that Participants shall have the right to submit written nominations of one or more persons to serve on the new Committee and all Participants shall be provided with written notice of any such election at their last known residence at least 30 days before the date of the election.

3.2 <u>Board and Committee Authority; Rules and Regulations</u>. The Committee shall have discretionary authority to (a) make, amend, interpret and enforce all appropriate rules and regulations for the administration of the Plan, and (b) decide or resolve, in its discretion, any and all questions, including interpretations of the Plan, as may arise in connection with the Plan. The Committee shall have authority to take or approve, in its discretion, all such actions in relation to the Plan (including, without limitation, actions described in the preceding sentence) as may be taken or approved by the Board; provided, however, that, except as provided in Section 8.2, the Committee shall have no authority to (a) amend, suspend or terminate the Plan or (b) to designate or remove a Company from the list of Affiliates authorized to participate in the Plan. Notwithstanding the foregoing, but subject to the provisions of Section 8.2, the Board may, by written notice to the Committee, withdraw all or any part of the Committee's authority at any time, in which case such withdrawn authority shall immediately revest in the Board. The decision or action of the Committee in respect of any question arising out of or in connection with the administration, interpretation and application of the Plan and the rules and regulations promulgated hereunder shall be final, conclusive and binding upon all persons having any interest in the Plan.

3.3 <u>Appointment of Agents</u>. In the administration of the Plan, the Board and/or the Committee may from time to time employ agents (which may include officers and/or employees of the Sponsor or any Affiliated Company) and delegate to them such

administrative duties as it sees fit and may from time to time consult with counsel who may be counsel to the Sponsor or any Affiliated Company.

3.4 <u>Reports</u>. Within a reasonable amount of time following each calendar quarter, the Committee shall cause to be given to each Participant a statement showing the dollar amount of his or her Account under the Plan as of the end of such calendar quarter.

3.5 <u>Termination of Active Participation</u>. In the event that the Committee determines that a Participant's employment performance is no longer at a level which merits continued active participation in the Plan, the Committee may terminate such Participant's active participation in the Plan (without necessarily terminating such Participant's employment) as of the date specified by the Committee. After the effective date of a Participant's termination of active participation in the Plan, such Participant shall not be eligible to make Compensation Deferrals under the Plan nor shall such Participant be entitled to have credited to his or her Account Compensation Deferrals pursuant to the provisions of Section 5.2; nevertheless, such a Participant shall be entitled to have his or her Account adjusted as provided in Section 5.3 hereof.

3.6 <u>Leave of Absence</u>. In the event the Participant takes a leave of absence from active employment with the Sponsor and all Affiliated Companies, the Committee shall determine, in its discretion, whether such leave of absence shall be deemed to constitute, a Termination for purposes of the Plan.

ARTICLE IV

COMPENSATION DEFERRALS AND COMPANY CONTRIBUTIONS

4.1 <u>Eligibility to Make Compensation Deferrals</u>. Subject to the limitations of this Article IV, each employee who, as of the last day of the Plan Year immediately preceding the Plan Year for which an eligibility determination is being made (the "Determination Date"), meets at least one of the following requirements, shall be eligible to enter into a Compensation Deferral Agreement to make Compensation Deferrals and become a Participant, effective as of first day of the Plan Year:

(a) The employee is designated as, and performs the functions of an officer of the Company having the rank of at least a Vice President and the employee's annual base salary as of the Determination Date was at least the greater of: (i) \$80,000 or (ii) the amount specified in Section 414(q)(1)(B)(i) of the Internal Revenue Code; or

(b) The employee is designated as, and performs the functions of an officer of the Company having the rank of at least a Vice President and the employee's Compensation actually received in the calendar year ending on the Determination Date was at least the greater of: (i) \$80,000 or (ii) the amount specified in Section 414(q)(1)(B)(i) of the Internal Revenue Code.

4.2 Election to Make Compensation Deferrals.

(a) <u>Salary Deferrals</u>. Each Active Participant who desires to make Salary Deferrals for a Plan Year shall enter into a Compensation Deferral Agreement for such year specifying the amount of Salary for such year to be deferred (stated in terms of a whole percentage of Salary not to exceed 50%) and such other information as the Committee shall determine. The amount of Salary Deferrals elected for a Plan Year shall not be less than 5% of Salary.

(b) <u>Monthly Bonus/Commission Deferrals</u>. Each Active Participant who desires to make Deferrals of monthly Bonuses or Commissions for a Plan Year shall enter into a Compensation Deferral Agreement for such year specifying the amount of monthly Bonuses or Commissions for such year to be deferred (stated after in terms of a whole percentage of Bonus or Commission Deferrals not to exceed 100%) and such other information as the Committee shall determine. The amount of monthly Bonus or Commission Deferrals elected for a Plan Year shall not be less than 5% of monthly Bonuses or Commissions.

(c) <u>Quarterly Bonus/Commission Deferrals</u>. Each Active Participant who desires to make Deferrals of quarterly Bonuses or Commissions for a Plan Year shall enter into a Compensation Deferral Agreement for such year specifying the amount of quarterly Bonuses or Commissions for such year to be deferred (stated in terms of a whole percentage not to exceed 100%) and such other information as the Committee shall determine. The amount of quarterly Bonus or Commission Deferrals elected for a Plan Year shall not be less than 5% of quarterly Bonuses or Commissions.

(d) <u>Annual Bonus/Commission Deferrals</u>. Each Active Participant who desires to make Deferrals of annual Bonuses or Commissions for a Plan Year shall enter into a Compensation Deferral Agreement for such year specifying the amount of annual Bonuses or Commissions for such year to be deferred (stated in terms of a whole percentage not to exceed 100%) and such other information as the Committee shall determine. The amount of annual Bonus or Commission Deferrals elected for a Plan Year shall not be less than 5% of annual Bonuses or Commissions.

(e) <u>Procedures</u>. A Compensation Deferral Agreement under this Section 4.2 shall not be effective unless it is executed by the Participant and returned to the Committee before the first day of the first Plan Year with respect to which the Agreement relates. Once executed, a Compensation Deferral Agreement shall remain in effect for all subsequent Plan Years unless revoked or modified in writing by the Participant before the first day of the Plan Year for which the revocation or modification is to take effect. An election to make Compensation Deferrals for a Plan Year (including a failure to revoke or modify a prior election in a timely manner), shall be irrevocable after the beginning of the Plan Year with respect to which the election relates. A Participant's failure to revoke or modify his original deferral election in writing before the first day of the following Plan Year shall constitute a waiver by the Participant of the Participant's right to elect a different deferral amount and shall constitute an election to make deferrals as previously elected and shall apply as of the new Plan Year.

4.3 <u>Continuing Participation</u>. Once an employee becomes eligible to participate in the Plan, the employee shall continue to be eligible to participate in the Plan until eligibility for participation is terminated as provided in Section 3.5.

4.4 <u>No Deferrals After a Termination</u>. No Compensation Deferrals shall be made after a Participant terminates employment with the Company.

4.5 <u>Spousal Consent</u>. Compensation Deferrals under this Article IV shall not be subject to the consent of the Participant's spouse, if any.

4.6 <u>Withholding</u>. No deferral election shall reduce a Participant's compensation below the amount necessary to pay any applicable employment taxes on amounts deferred,

any income tax applicable to compensation that is not deferred or any amount required to be withheld to fund any other benefit or plan of the Company.

4.7 <u>Company Contributions</u>. Each participating Company may, in its sole discretion, make a Company Contribution to the Plan in such amount as the Company, in its sole discretion determines. Company Contributions shall be allocated among the Accounts of Participants as specified by each Company in its sole discretion.

4.8 <u>Special Provisions for 2000 Salary Deferrals</u>. Notwithstanding the requirement of Section 4.2 that elections to defer Compensation must be made before the first day of the Plan Year, Participants shall be permitted to make Compensation Deferrals for the period beginning on February 1, 2000 and continuing through December 31, 2000 (the "2000 Deferral Period") pursuant to this Section 4.8. A Compensation Deferral Agreement under this Section 4.8 shall not be effective unless it is executed by the Participant and returned to the Committee before January 25, 2000. A Participant's election to make Compensation Deferrals for the 2000 Deferral Period shall be irrevocable after January 25, 2000.

ARTICLE V

ACCOUNTS; VALUATION OF ACCOUNTS; VESTING

5.1 <u>Separate Accounts for Participants</u>. The Committee shall have the responsibility to keep separate Accounts for each Participant in accordance with the provisions of this Article V reflecting amounts credited to the Participant under the Plan (subject to the restrictions set forth in Section 9.1 below). Each Participant's Account shall represent a book reserve in the name of the Participant.

5.2 <u>Crediting of Compensation Deferrals to Participants' Accounts</u>. All Compensation Deferrals and all Company Contributions made for a particular year shall be credited to separate Accounts (the "Compensation Deferral Account "and the "Company Contribution Account") for each Participant by whom such deferrals are made.

5.3 <u>Adjustments for Investment Experience</u>. Each Participant's Account(s) shall be adjusted as of each Valuation Date. The adjustments made shall be those adjustments reflecting income, gains and losses that would have been made to the Participant's Account(s) had the dollar value of the Account been invested in the Investments selected by the Participant pursuant to the provisions of Section 5.4.

5.4 <u>Designation of Investments</u>. For purposes of determining the adjustments for investment experience set forth in Section 5.3 above, the dollar value of each Participant's Account(s) shall be deemed to be invested in accordance with the Participant's investment designation in one or more Investments established by the Committee pursuant to rules established by the Committee. The Committee may, at its discretion, establish alternative Investments or eliminate any previously established Investments at any time and from time to time. In accordance with rules established by the Committee, each Participant may elect the Investment in which his or her Account is deemed invested.

5.5 <u>Valuation of Accounts</u>. A Participant's Account(s) under the Plan shall be valued by the Committee as of each Valuation Date. A Participant's Account(s) shall be calculated by starting with the balance of the Account as of the prior Valuation Date (assuming a \$0 balance for the initial Valuation Date) and by (a) adding to the Participant's Account those amounts, if any, credited to the Account of such Participant under Section 5.2 above since the last Valuation date, (b) making the adjustments to such Participant's Account set forth in Section 5.3 above since the last Valuation Date, and (c) subtracting the aggregate amount of distributions or withdrawals made since the last Valuation Date to or with respect to such Participant.

5.6 <u>Vesting of Accounts</u>. Participant Accounts shall vest in accordance with the following provisions:

(a) <u>Compensation Deferrals Account</u>. A Participant shall be fully vested at all times in the amount of their Compensation Deferrals Account.

(b) <u>Company Contributions Account</u>. A Participant's Company Contributions Account shall become fully vested upon the Participant's Early Retirement, Normal Retirement, death or Disability while employed by the Company or upon a Change in Control as defined in Article XI. In addition, a Participant's Company Contributions Account shall become vested in accordance with the following schedule:

Years of Participation	Vested Percentage					
Less than 3	0	%				
3 or more	100	%				

(c) <u>Year of Participation</u>. For purposes of this Section 5.6(b), a "Year of Participation" shall be defined as a twelve consecutive month period beginning on the date that the Company Contribution is approved by the Board of Directors of the Company electing to make a Company Contribution and ending on the anniversary of such date. No credit shall be given for participation.

5.7 <u>Forfeitures</u>. Any amounts not vested at the time of a Participant's Termination or upon termination of the Plan as provided in Section 8.1 shall be forfeited.

ARTICLE VI

BENEFITS

6.1 <u>Eligibility</u>. Any Participant who incurs any Termination shall receive his or her Distributable Benefit in an amount equal to the vested balance of the Participant's Accounts as of the Valuation Date immediately following such Termination. Except as otherwise provided in Article XI hereof, a Participant's Distributable Benefit shall be paid at the times and in the amounts specified in this Article VI. A Participant's Distributable Benefit shall be paid (or commenced to be paid) within 30 days after the first Valuation Date following such Termination or as soon as practicable thereafter.

6.2 <u>Form of Plan Benefit</u>. Any Participant or Beneficiary who is entitled to receive his or her Distributable Benefit shall receive such Benefit in the form of a lump sum payment unless (a) the value of the Participant's Accounts exceeds \$100,000 in the aggregate, and (b) the Participant elects in accordance with the provisions of Section 6.3 to have his or her Distributable Benefit paid in substantially equal annual payments payable over fifteen (15) years. In the event a Participant elects to have his or her Distributable Benefit paid in substantially equal annual payments, the investment experience adjustments set forth in Section 5.3 shall continue to be made to the Participant's Account until the final payment is made to the Participant, reducing his or her Account balance to zero.

6.3 <u>Election of Form of Benefit</u>. A Participant shall elect the form in which his or her Distributable Benefit will be paid in the event of a Termination on account of Early Retirement, Normal Retirement, death, or Disability in his or her Initial Participation Agreement. Except to the extent that the Participant qualified for a Hardship Withdrawal as provided in Section 6.4 or elects a Short Term Pay-Out as provided in Section 6.5, the Participant's election as to the timing and form of distribution that is made on the Participant's Initial Participation Agreement shall apply to all subsequent Compensation Deferrals and shall be irrevocable once made.

6.4 <u>Hardship Withdrawals</u>. A Participant may withdraw from his or her Compensation Deferral Account Participant Deferrals previously made, but only if the Committee determines that the Participant has an unforeseeable emergency and that a withdrawal is reasonably necessary in order to satisfy the emergency need.

(a) An "unforeseeable emergency," as used in this Section 6.4, shall mean severe financial hardship to the Participant resulting from a sudden and unexpected illness or accident of the Participant or a dependent (as defined in Code Section 152(a)) of the Participant, loss of the Participant's property due to casualty, or other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant.

(b) The determination as to whether a withdrawal is reasonably necessary to satisfy an unforeseeable emergency need is to be made on the basis of all relevant facts and circumstances. However, a withdrawal shall be necessary in order to satisfy an immediate and heavy financial need only if:

(i) The amount of the withdrawal is not in excess of the amount required to relieve the emergency need or in excess of the amount that such need could not be satisfied from other sources that are reasonably available to the Participant. The amount of any hardship withdrawal may include any amounts necessary to pay federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution.

(ii) The Participant submits a signed statement to the Committee, on which the Committee can reasonably rely, to the extent that the need cannot be relieved:

(I) Through reimbursement or compensation by insurance or otherwise;

(II) By reasonable liquidation of the Participant's assets, to the extent such liquidation would not itself cause an immediate and heavy financial need;

(III) By cessation of deferrals under the Sponsor's Retirement Savings Plan; or

(IV) By other withdrawals, distributions or nontaxable loans from any plan maintained by the Company or any other employer, or by borrowing from commercial sources on reasonable commercial terms.

(iii) A Participant's resources shall be deemed to include those assets of his or her spouse and minor children that are reasonably available to the Participant.

(iv) A Participant who makes a hardship withdrawal pursuant to this Section 6.4 shall not be eligible to make Compensation Deferrals for the taxable year of the Participant immediately following the taxable year of such hardship withdrawal.

6.5 <u>Short Term Pay-Outs</u>. A Participant may designate in advance on his or her Compensation Deferral Agreement the time and amount of one or more withdrawals of Compensation Deferrals (but not investment earnings thereon) from his or her Account, subject to the following rules: (a) A Participant may designate a time for withdrawal not earlier than two years after the year that Compensation was deferred under such

Compensation Deferral Agreement which otherwise would have been paid absent the deferral election (such amount to be designated as payable after 2, 3, 5 or 10 years); and (b) such election shall be irrevocable. Notwithstanding a Participant's election to receive a Short Term Pay-Out in a particular dollar amount, the maximum amount payable as a Short Term Pay-Out shall be limited to the lesser of (a) the account balance of the Participant's Compensation Deferrals Account or (b) the dollar value of the Compensation Deferrals with respect to the Plan Year for which the Deferral was elected, each determined as of the Valuation Date preceding the payment of such Short Term Pay-Out.

6.6 <u>Disputes</u>. In the event of a dispute between a Participant and the Company concerning the amount of the Participant's Distributable Benefit, the claims procedures and mandatory arbitration provisions of Article VII shall apply.

6.7 <u>Lump Sum Payment of Withdrawals</u>. All Short Term Pay-Outs shall be paid to Participants as soon as reasonably practicable following the approval by the Committee of a properly completed withdrawal request. All Short Term Pay-Outs shall be made in a single lump sum payment in cash.

6.8 <u>Lump Sum Payment Upon Termination of Plan</u>. Upon a termination of the Plan pursuant to Section 8.1 or Section 8.2, all Distributable Benefits shall be paid to Participants (including Participants receiving installment payments), in the form of a single lump sum payment in cash within 30 days after the first Valuation Date following termination of the Plan, or as soon as practicable thereafter.

ARTICLE VII

APPLICATION FOR BENEFITS

7.1 <u>Initial Application</u>. In order for a Participant to receive or commence receiving his or her Distributable Benefit, such Participant (or, in the case of a Participant who incurs a Termination by reason of death, the Participant's Beneficiary) must file a written application for such Plan benefit (an "Initial Application for Benefits") with the Committee. The Initial Application for Benefits shall contain at least the following information:

(a) The Participant's name;

(b) The Participant's date of Termination and the reason for such Termination (that is, whether such Termination is on account of the Participant's death, Disability or Retirement);

(c) A brief statement setting forth the basis (with references to pertinent Plan provisions where applicable) of the Participant's claim for his or her Distributable Benefit; and

(d) Such other documents and information as the Committee may reasonably require.

Notwithstanding the foregoing, if the Committee determines, based on such evidence as it deems appropriate, that a Participant has incurred Termination (or that the Plan has been terminated) and that such Participant (or other person claiming benefits with respect to such Participant) would be entitled, but for the failure to file an application for benefits, to commence receiving benefit payments under this Plan, then the Committee may, in its discretion, deem an application for such benefits to have been filed.

7.2 <u>Action on Applications</u>. At such time as the Committee receives any Initial Application for Benefits (hereinafter shall be referred to as a "Claim," and the person filing any Claim shall be referred to hereinafter as a "Claimant"), the Committee shall review the Claim, together with other pertinent information (including any additional information the Committee may reasonably request from the Claimant), and shall allow or deny the Claim. The Committee shall provide written notice to the Claimant of the Committee's decision to allow or deny the Claim (a "Notice of Action on Claim") within 60 days after the Committee receives the Claim. If the Claim is denied, the Notice of Action on Claim shall include the following information:

(a) The specific reasons for the denial;

(b) Specific reference to pertinent Plan provisions on which the denial is based;

(c) If applicable, a description of any additional information or material necessary to perfect the Claim, and an explanation of why such information or material is necessary; and

(d) An explanation of the claims review procedure.

7.3 <u>Review Procedure</u>.

(a) A Claimant is entitled to request the Committee to review any denial by the Committee of his or her Claim. The request for review must be submitted in writing to the Committee within 60 days after the date the Notice of Action on Claim is deemed given pursuant to Section 12.6 hereof. If the Claimant fails to make a timely request for review, the Claim will be deemed conclusively to be denied. The Claimant or his or her representative shall be entitled to review all pertinent documents and to submit relevant issues and comments to the Committee in writing.

(b) The Committee shall review the Claim and shall render the final decision to allow or deny the Claim. The Committee shall give written notice of its decision to the Claimant within 60 days after it receives the written request for review. If the Claim is denied, such written notice shall recite the facts on which the Committee based its decision as well as its reasons therefor, with specific reference to pertinent Plan provisions where applicable.

7.4 <u>Mandatory Arbitration</u>. Following exhaustion of all administrative remedies as described in Sections 7.1-7.4, above, in the event of any unresolved dispute under the provisions of this Plan, such dispute shall be submitted to arbitration in accordance with the provisions of Section 12.12 of the Plan.

7.5 <u>Claims By Representative</u>. In the event that any person who is entitled (or required) to file a Claim under this Article VII is deceased or is mentally or physically incapacitated in a manner that prevents such person from filing such Claim, then such Claim may be filed by another person on behalf of such Claimant, subject to such other person's providing to the Committee or Committee, as applicable, reasonable verification of such other person's authorization to act on the Claimant's behalf.

ARTICLE VIII

CHANGES TO THE PLAN

8.1 <u>Termination, Suspension or Amendment of Plan by Board</u>. Subject to the provisions of Section 8.2, the Board may, in its sole discretion, terminate or suspend this Plan at any time or from time to time, in whole or in part, and, except as otherwise provided by the Board, no further Compensation Deferrals or Company Contributions shall be credited to Participants' Accounts as of the date of such termination or suspension. Subject to the provisions of Section 8.3, the Board may amend this Plan at any time or from time to time, and any amendment may provide benefits different in kind and/or amount from those herein set forth. Notwithstanding the foregoing, in the case of any termination, suspension or amendment of this Plan, such termination, suspension or amendment shall not adversely affect (a) the Account balance of any such Participant as of the effective date of such termination, suspension or amendment, (b) the Participant's right to the investment experience adjustments after any suspension or amendment pursuant to the provisions of Section 5.3, (c) the Participant's right to investment experience adjustments after any suspension or amendment pursuant to the provisions of Section 5.3, or (d) the right or ability of any such Participant (and, if applicable, such Participant's Beneficiary) to receive his or her vested Distributable Benefit in accordance with the terms of this Plan as in effect immediately prior to such termination, suspension or amendment. Any termination, suspension or amendment of this Plan by the Board shall be binding on each Affiliated Company, without further action by any such Affiliated Company.

8.2 <u>Termination of Plan Upon a Change in Control</u>. Notwithstanding the provisions of Section 8.1 of the Plan, upon the occurrence of a Change in Control, the Committee may, in its sole discretion, elect to terminate all or a portion of this Plan, in which case no further Compensation Deferrals or Company Contributions shall be credited to Participants' Accounts as of the date of such termination and benefits shall be paid out as soon as practicable thereafter as provided in Section 6.8. In the case of any termination of the Plan by the Committee, such termination shall not adversely affect (a) the Account balance of any such Participant as of the effective date of such termination, (b) the Participant's right to investment experience adjustments after termination of the Plan, up to and including the effective date of such termination, pursuant to the provisions of Section 5.3, or (c) subject to the provisions of Section 6.8, the right or ability of any such Participant (and, if applicable, such Participant's Beneficiary) to receive his or her vested Distributable Benefit in accordance with the terms of this Plan as in effect immediately prior to such termination. Any termination of this Plan by the Committee pursuant to this Section 8.2 shall be binding on each Affiliated Company, without further action by any such Affiliated Company.

^{8.3 &}lt;u>Amendment of Plan Following a Change in Control</u>. Notwithstanding the provisions of Section 8.1 of the Plan, upon the occurrence of a Change in Control of the Sponsor, if the Committee does not terminate the Plan as provided in Section 8.2, the Plan shall continue in force as provided in Section 12.4: provided, however, that for a period of 24 months following the date of a Change in Control of the Sponsor, no amendment to the Plan shall be effective unless such amendment is consented to in writing by a majority of Plan Participants pursuant to procedures established by the Committee.

ARTICLE IX

FUNDING

9.1 In General. Benefits under this Plan shall be payable solely from the general assets of the Sponsor (and, with respect to any Participant who is an employee of an Affiliated Company, also from the general assets of such Affiliated Company), and no person shall be entitled to look to any other source for payment of such benefits. The Sponsor (and, if applicable, any Affiliated Company) shall have and possess all title to, and beneficial interest in, any and all funds or reserves maintained or held by the Sponsor (or such Affiliated Company) on account of any obligation to pay benefits as required this Plan, whether or not earmarked as a fund or reserve for such purpose; any such funds, other property or reserves shall be subject to the claims of the creditors of the Sponsor (or such Affiliated Company), and the provisions of this Plan are not intended to create, and shall not be interpreted as vesting, in any Participant, Beneficiary or other person, any right to or beneficial interest in any such funds, other property or reserves. Nothing in this Section 9.1 shall be construed or interpreted as prohibiting or restricting the establishment of a grantor trust (within the mean of Code Section 671) from which benefits under this Plan may be payable, so long as such trust or contract does not cause the plan to be funded for purposes of Sections 201(a), 301(a)(3) or 401(a)(1) of ERISA.

9.2 <u>Establishment of Rabbi Trust</u>. The Sponsor has established (or will establish) a trust ("Trust") substantially in the form attached hereto as Exhibit B for the purpose of funding benefits under this Plan as provided in this Article IX and for the purpose of paying the reasonable legal and administration fees and expenses incurred by Participants and their Beneficiaries in connection with lawsuits, actions or other proceedings brought by such persons to enforce their rights under the Plan.

9.3 <u>Funding of Rabbi Trust</u>. Subject to the amendment or termination of this Plan and the restrictions set forth in Section 9.1, the Company shall contribute to the Trust, on behalf of each Participant, the amount of Compensation Deferrals made by such Participant and the amount of any Company Contribution allocated to such Participant's Account. Amounts corresponding to Salary Deferrals for each payroll period shall be contributed to the Trust as soon as reasonably practicable following the end of such payroll period. Bonus Deferrals for a Plan Year shall be contributed to the Trust as soon as reasonably practicable following the date on which such Bonus amounts otherwise would have been paid to the Participant.

ARTICLE X

AFFILIATED COMPANIES

10.1 <u>Affiliated Company Participation</u>. The Board of Directors may revoke or modify an Affiliated Company's participation in this Plan at any time and for any or no reason, or terminate this Plan with respect to such Affiliated Company's employees and Participants, subject to the provisions of Section 8.1 concerning changes in the Plan.

10.2 <u>Adoption Agreement</u>. Any Affiliated Company participating in the Plan shall adopt the Plan and shall agree to be bound by the terms and conditions of the Plan and any future changes made by the Board to the Plan.

ARTICLE XI

CHANGE IN CONTROL

11.1 <u>Change in Control</u>. As used in this Plan, "Change in Control" shall mean the following and shall be deemed to occur if any of the following events occur:

(a) Any "person," as such term is used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of Impac Mortgage Holdings Company (the "Parent") representing 50% or more of the combined voting power of the Parent's then outstanding voting securities.

(b) Any "person," as such term is used in Sections 13 (d) and 14(d) of the Exchange Act, other than any person who were the beneficial owners of securities of Impac Funding Corporation, a California corporation, as January 1, 2000, or a direct or indirect subsidiary of Parent ("Parent Subsidiary"), is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the 50% or more of the combined voting power of Sponsor's then outstanding voting securities).

(c) During any period of two consecutive years, individuals who, as of the beginning of such period, constitute the Parent's Board of Directors (the "Parent's Incumbent Board") cease for any reason to constitute at least a majority of the Parent's Board of Directors, unless the election, or nomination for election by the Parent's stockholders, of each new director is approved by a vote of at least two-thirds of the directors then comprising the Parent's Incumbent Board.

(d) During any period of two consecutive years, individuals who, as of the beginning of such period, constitute the Sponsor's Board of Directors (the "Sponsor's Incumbent Board") cease for any reason to constitute at least a majority of the Sponsor's Board of Directors, unless the election, or nomination for election by the Sponsor's stockholders, of each new director is approved by a vote of at least two-thirds of the directors then comprising the Sponsor's Incumbent Board.

(e) The stockholders of the Parent approve a merger or consolidation of Parent with any other corporation, other than the following

(i) A merger or consolidation which would result in the voting securities of the Parent outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities or another entity) more

than 50% of the combined voting power of the voting securities of the Parent or such other entity outstanding immediately after such merger or consolidation; or

(ii) A merger or consolidation effected to implement a recapitalization of the Parent (or similar transaction) in which no person acquires 50% or more of the combined voting power of the Parent's then outstanding voting securities.

(f) The stockholders of the Sponsor approve a merger or consolidation of Sponsor with any other corporation, other than the following

(i) A merger or consolidation which would result in the voting securities of the Sponsor outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being 50% of the combined voting power of the voting securities of the Sponsor or such other entity outstanding immediately after such merger or consolidation); or

(ii) A merger or consolidation effected to implement a recapitalization of the Sponsor (or similar transaction) in which no person acquires 50% or more of the combined voting power of the Sponsor's then outstanding voting securities; or

(iii) A merger or consolidation with a Parent Subsidiary.

(g) The stockholders of Parent approve a plan of complete liquidation of the Parent or an agreement for the sale or other disposition by the Parent of all or substantially all of the Parent's assets.

(h) The stockholders of Sponsor approve a plan of complete liquidation of the Sponsor or an agreement for the sale or other disposition by the Sponsor of all or substantially all of the Sponsor's assets to a person that is not a Parent Subsidiary.

Notwithstanding the preceding provisions of this Section 11.1, a Change in Control shall not be deemed to have occurred (1) if the "person" described in the preceding provisions of this Section 11.1, is an underwriter or underwriting syndicate that has acquired the ownership of 50% or more of the combined voting power of the Parent's then outstanding voting securities solely in connection with a public offering of the Parent's securities, or (2) if the "person" described in the preceding provisions of this Section 11.1 is an underwriter or underwriting syndicate that has acquired the ownership of 50% or more of the combined voting power of the Sponsor's then outstanding voting securities solely in connection with a public offering of the Sponsor's securities. Further, a Change of Control of any one participating Company shall be deemed to be a Change in Control only

with respect to employees of that Company and shall not be deemed to be a Change in Control with respect to employees of any other participating Company.

ARTICLE XII

MISCELLANEOUS PROVISIONS

12.1 <u>Designation of Beneficiary</u>. A Participant shall be entitled to designate one or more individuals or entities (including a trust or trusts), in any combination, as his or her "Beneficiary" or "Beneficiaries" to receive any Plan benefit payments to which such Participant is entitled as of, or by reason of, his or her death; provided, however, that the designation of any individual or entity other than the Participant's spouse shall be effective only with the consent of the Participant's spouse. Any such designation may be made or changed at any time prior to the Participant's death by written notice filed with the Committee, with such written notice to be in such form and contain such information as the Committee may from time to time determine. In the event that (a) a Beneficiary designation is not on file or is not effective at the date of a Participant's death, (b) no Beneficiary survives the Participant, or (c) no Beneficiary is living at the time any payment becomes payable under this Plan, then, for purposes of making any further payment of any unpaid benefits under the Plan, such Participant's Beneficiary or Beneficiaries shall be deemed to be the Participant's estate. A Beneficiary entitled to receive any Plan Benefit payments pursuant to this Section 12.1 shall receive such payments at the time(s) and in the manner elected by the Participant in his or her Participation Agreement.

12.2 <u>Payments During Incapacity</u>. The Committee shall have no duty or obligation to determine the competence or capacity of any Participant or Beneficiary. However, in the event that the Committee is made aware that a Participant (or Beneficiary) is under mental or physical incapacity at the time of any payment to be made to such Participant (or Beneficiary) pursuant to this Plan, any such payment may be made to the conservator or other legally appointed personal representative having authority over and responsibility for the person or estate of such Participant (or Beneficiary), as the case may be, and for purposes of such payment references in this Plan to the Participant (or Beneficiary) shall mean and refer to such conservator or other personal representative of the person or estate of the Participant (or Beneficiary), any such payment may be made to any person or institution that has apparent responsibility for the person and/or estate of the Participant (or Beneficiary) as determined by the Committee. Any payment made in accordance with the provisions of this Section 12.2 to a person or institution other than the Participant (or Beneficiary) shall be deemed for all purposes of this Plan as the equivalent of a payment to such Participant (or Beneficiary), and the Company shall have no further obligation or responsibility with respect to such payment.

12.3 <u>Prohibition Against Assignment</u>. Except as otherwise expressly provided in Section 12.1 and Section 12.2 hereof, the rights, interests and benefits of a Participant under this Plan (a) may not be sold, assigned, transferred, pledged, hypothecated, gifted, bequeathed or otherwise disposed of to any other party by such Participant or any Beneficiary, executor, administrator, heir, distribute or other person claiming under such Participant, and (b) shall not be subject to execution, attachment or similar process. Any

attempted sale, assignment, transfer, pledge, hypothecation, gift, bequest or other disposition of such rights, interests or benefits contrary to the foregoing provisions of this Section 12.3 shall be null and void and without effect.

12.4 <u>Assumption of Plan</u>. The Company expressly agrees that it shall not merge or consolidate into or with another corporation, or sell substantially all of its assets to another corporation, firm or person until such corporation, firm or person expressly agrees in writing to assume and discharge the duties and obligations of the Company under this Plan. The powers, duties, rights and obligations of the Company under this Plan may be assigned (i) to any company into which the Company may be merged, or with which it may be consolidated, or (ii) to any company resulting from any merger, reorganization or consolidation to which the Company is a party, or (iii) to any company to which the business of the Company may be transferred, to the extent affected Participants transfer their employment to such company. If such an assignment occurs, the successor-in-interest to the Company shall assume the powers, duties, rights and obligations of the Company in writing.

12.5 <u>Binding Effect</u>. The provisions of this Plan shall be binding upon the Sponsor, the Participants, all Affiliated Companies employing any Participants, and any successor-in-interest, beneficiary, heir or personal representative to the Sponsor, any Participant or any such Affiliated Company.

12.6 <u>No Right to Employment</u>. This Plan is voluntary on the part of the Sponsor and its Affiliated Companies, and shall not be deemed to constitute an employment contract between any Participant and the Sponsor or any Affiliated Company, nor shall the adoption or existence of the Plan or any provision contained in the Plan be deemed to be a required condition of the employment of any Participant. Nothing contained in this Plan shall be deemed to give any Participant the right to continued employment with the Sponsor or any Affiliated Company, and the Sponsor and its Affiliated Companies may terminate any Participant at any time, in which case the Participant's rights arising under this Plan shall be only those expressly provided under the terms of this Plan.

12.7 <u>Notices</u>. All notices, requests, or other communications (hereinafter collectively referred to as "Notices") required or permitted to be given hereunder or which are given with respect to this Plan shall be in writing and may be personally delivered, or may be sent by United States registered mail, postage prepaid, return receipt requested and addressed as follows:

To the Sponsor, and Affiliated Company or the Committee at:

> Impac Funding Corporation 1401 Dove Street Newport Beach, CA 92660

To Participant at:

The Participant's residential mailing address as reflected in the Sponsor's or Affiliated Company's employment records

A Notice which is delivered personally shall be deemed given as of the date of personal delivery, and a Notice mailed as provided herein shall be deemed given on the date received. Any Participant may change his or her address for purposes of Notices hereunder pursuant to a Notice to the Committee, given as provided herein, advising the Committee of such change. The Sponsor, any Affiliated Company and/or the Committee may at any time change its address for purposes of Notices hereunder pursuant to a Notice to all affected Participants, given as provided herein, advising such Participants of such change.

12.8 <u>Governing Law</u>. This Plan shall be governed by, interpreted under and construed and enforced in accordance with the internal laws, and not the laws pertaining to conflicts or choice of laws, of the State of California applicable to agreements made and to be performed wholly within the State of California.

12.9 <u>Titles and Headings; Gender of Terms</u>. Article and Section headings herein are for reference purposes only and shall not be deemed to be part of the substance of this Plan or in any way to enlarge or limit the meaning or interpretation of any provision in this Plan. Use in this Plan of the masculine, feminine or neuter gender shall be deemed to include each of the omitted genders if the context so requires.

12.10 <u>Severability</u>. In the event that any provision of this Plan is found to be invalid or otherwise unenforceable by a court or other tribunal of competent jurisdiction, such invalidity or unenforceability shall not be construed as rendering any other provision contained herein invalid or unenforceable, and all such other provisions shall be given full force and effect to the same extent as though the invalid and unenforceable provision was not contained herein.

12.11 <u>Tax Effect of Plan</u>. Neither the Sponsor nor any Affiliated Company warrants any tax benefit or any financial benefit under this Plan. Without limiting the foregoing, the Sponsor and each Affiliated Company and their directors, officers, employees and agents shall be held harmless by the Participant from, and shall not be subject to any liability on account of, any Federal or State tax consequences or any consequences under ERISA of any determination as to the amount of plan benefits to be paid, the method by which plan benefits are paid, the persons to whom plan benefits are paid, or the commencement or termination of the payment of plan benefits.

12.12 <u>Mandatory Arbitration</u>. In the event of any unresolved dispute between a Participant or Beneficiary and the Company or the Trustee regarding this Plan, such dispute shall be submitted to arbitration in accordance with the rules and regulations of the American Arbitration Association for the arbitration of employee benefit plan claim disputes (including the rules contained therein for the selection of an arbitrator who is familiar with employee benefit plans). The written determination of the arbitrator shall be final, binding and conclusive on the Parties and judgment on the award rendered by the arbitrator may be entered by any court having jurisdiction thereof. The prevailing party or parties of any arbitration of legal or equitable proceeding shall be entitled to have and recover his or their attorneys fees, costs and expenses as determined by the arbitrator.

IN WITNESS WHEREOF, the Sponsor has caused this Plan to be executed by its duly authorized officer effective as of the Effective Date hereof.

IMPAC FUNDING CORPORATION

E.V.P

/s/ Ron Morrison

28

By:

Title:

Independent Auditors' Consent

The Board of Directors Impac Mortgage Holdings, Inc.:

We consent to the incorporation by reference in the registration statement (No. 333-83650) on Form S-8 and registration statement (No. 333-74432) on form S-3 of Impac Mortgage Holdings, Inc. and Subsidiaries, of our report dated January 29, 2003, except as to the last paragrah of Note M to the consolidated financial statements which is as of February 27, 2003, relating to the consolidated balance sheets of Impac Mortgage Holdings, Inc. and Subsidiaries as of December 31, 2002 and 2001, and the related consolidated statements of operations and comprehensive income (loss), changes in stockholders' equity and cash flows for each of the years in the three-year period ended December 31, 2002, which report appears in the December 31, 2002, annual report on Form 10-K of Impac Mortgage Holdings, Inc.

As discussed in Note A to the consolidated financial statements, the Company changed its method of accounting for derivative instruments and hedging activities in 2001.

Orange County, California March 27, 2003

Independent Auditors' Consent

The Board of Directors Impac Funding Corporation:

We consent to the incorporation by reference in the registration statement (No. 333-83650) on Form S-8 and registration statement (No. 333-74432) on form S-3 of Impac Mortgage Holdings, Inc., of our report dated January 29, 2003, except as to the last paragraph of note J to the consolidated financial statements, which is as of February 27, 2003, relating to the consolidated balance sheets of Impac Funding Corporation and subsidiaries as of December 31, 2002 and 2001, and the related consolidated statements of operations and comprehensive income (loss), changes in stockholders' equity and cash flows for each of the years in the three-year period ended December 31, 2002, which report appears in the December 31, 2002, annual report on Form 10-K of Impac Mortgage Holdings, Inc.

As discussed in Note A to the consolidated financial statements, the Company changed its method of accounting for derivative instruments and hedging activities in 2001.

Orange County, California March 27, 2003

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the annual report of Impac Mortgage Holdings, Inc. (the "Company") on Form 10-K for the period ending December 31, 2002 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each of the undersigned, in the capacities and on the dates indicated below, hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to his knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Joseph R. Tomkinson

Joseph R. Tomkinson Chief Executive Officer March 28, 2003

/s/ RICHARD J. JOHNSON

Richard J. Johnson Chief Financial Officer March 28, 2003

A signed original of this written statement required by Section 906 has been provided to Impac Mortgage Holdings, Inc. and will be retained by Impac Mortgage Holdings, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.